7/6/83

National Criminal Justice Reference Service



This microfiche was produced from documents received for inclusion in the NCJRS data base. Since NCJRS cannot exercise control over the physical condition of the documents submitted, the individual frame quality will vary. The resolution chart on this frame may be used to evaluate the document quality.



MICROCOPY RESOLUTION TEST CHART NATIONAL BUREAU OF STANDARDS-1963-A

Microfilming procedures used to create this fiche comply with the standards set forth in 41CFR 101-11.504.

Points of view or opinions si ited in this document are those of the author(s) and do not represent the official position or policies of the U.S. Department of Justice.

National Institute of Justice United States Department of Justice Washington, D.C. 20531

GUIDELINES FOR RESPONDING TO LAW ENFORCEMENT REQUESTS FOR ALCOHOL AND DRUG ABUSE PATIENT RECORDS

granted by

sion of the copyright owner.

Public Health Service



NATIONAL INSTITUTE ON DRUG ABUSE NATIONAL INSTITUTE ON ALCOHOL ABUSE AND ALCOHOLISM

U.S. Department of Justice National Institute of Justice

This document has been reproduced exactly as received from the person or organization originating it. Points of view or opinions stated in this document are those of the authors and do not necessarily represent the official position or policies of the National

Permission to reproduce this copyrighted material has been

Public Domain/US Dept. of Health & Human Services

to the National Criminal Justice Reference Service (NCJRS).

Further reproduction outside of the NCJRS system requires permis

U.S. Department of Health and Human Services

Alcohol, Drug Abuse, and Mental Health Administration

May 1980

National Criminal Justice Reference Service



· 0

While portions of this document are illegible, it was microfilmed from the best copy available. It is being distributed because of the valuable information it contains.

National Institute of Justice United States Department of Justice Washington, D.C. 20531



CONTENTS

Page

GUIDELINES LETTER TO ALCOHOL AND DRUG ABUSE 1-5 TREATMENT PROGRAMS

TAB A, REGULATIONS TITLED, "CONFIDENTIALITY 6-26 OF ALCOHOL AND DRUG ABUSE PATIENT RECORDS," 42 CFR PART 2

TAB B, COMPLAINT SEEKING INJUNCTION AND TO 27-51 BLOCK USE OF INFORMATION

TAB C, OPINIONS OF THE OFFICE OF THE 52-74 GENERAL COUNSEL

> NCJRS FEB 10 1983

> > ACQUISITIONS

DEPARTMENT OF HEALTH AND HUMAN SERVICES PUBLIC HEALTH SERVICE ALCOHOL, DRUG ABUSE, AND MENTAL HEALTH ADMINISTRATION 5600 FISHERS LANE ROCKVILLE, MARYLAND 20857

Directors Alcohol and Drug Abuse Treatment Programs

Dear Program Directors:

In response to the recent seizure of patient records from a drug treatment program in San Francisco, questions have arisen about how alcohol and drug abuse program personnel should handle requests from 'aw enforcement officials for information about patients. This letter, is intended to answer these questions and assist program personnel by setting forth guidelines for complying with the Federal confidentiality regulations (42 CFR Part 2) and the authorizing legislation (21 U.S.C. 1175, 42 U.S.C. 4582) when responding to law enforcement requests for copies of patient records or other patient identifying information. Enclosed at Tab A for your convenience is a copy of the regulations which quote the authorizing legislation at §§ 2.1 and 2.2.

Because the primary responsibility for compliance with the confidentiality statutes and regulations lies with the program and its staff, we recommend that these guidelines be thoroughly discussed with the program's legal counsel and that the program promptly undertake steps to ensure that its staff is familiar with and able to implement the recommended procedures.

General

1.

These guidelines apply to the personnel of all alcohol or drug abuse programs conducted, regulated, or directly or indirectly assisted by the Federal Government (See 42 CFR 2.12(a); 21 U.S.C. 1175(a); 42 U.S.C. 4582(a)). They provide information on how to handle law enforcement requests for alcohol or drug abuse patient records or other patient identifying information for the purpose of investigating or prosecuting any patient. They do not apply to other types of law enforcement requests for patient information, such as requests for information about a patient's treatment during probation, parole, or other pre or post-trial conditional release, which have been consented to by the patient in accordance with 42 CFR 2.39.

Any disclosure of patient records or other patient identifying information in response to law enforcement requests that are related to the investigation

May 5, 1980

legense en son de son de la service en son de l Terre en service en serv

Page 2 - Directors, Alcohol and Drug Abuse Treatment Programs

or prosecution of any patient must be authorized by a court order issued in accord with the requirements of 42 CFR Part 2, Subpart E. If a program employee is merely served with compulsory process from a Federal, State, or local court the individual is prohibited from disclosing the requested patient information under the confidentiality statutes and regulations (See 42 CFR 2.13(a), 2.61; 21 U.S.C. 1175(c); 42 U.S.C. 4582(c)).

Compulsory Process With a Court Order 2.

In those cases in which a program employee is served with both compulsory process and an authorizing court order issued under 42 CFR Part 2, Subpart E, the individual may comply with the compulsory process without violating the Federal confidentiality statutes and regulations (See 42 CFR 2.61). If the compulsory process requires a court appearance (such as a subpoena) or if the program employee has any questions regarding compliance with the request for information, he or she should immediately contact the program's legal counsel.

Compulsory Process Without a Court Order

3.

4.

If a program employee is served with compulsory process without a 42 CFR Part 2, Subpart E, authorizing court order, he or she must make a noncommittal response (See generally 42 CFR 2.13). The program employee should inform the law enforcement officials making the request that Federal law prohibits disclosure of the identity, the absence, presence, or whereabouts of any patient, or even the patient status of any person (See 42 CFR 2.13(b) and (c)). The officials should be referred to the confidentiality regulations, 42 CFR Part 2, and the authorizing statutes, 21 U.S.C. 1175 and 42 U.S.C. 4582, including specifically, the provisions under which a court order authorizing the disclosure may be sought (See 42 CFR 2.61-2.67). If the person about whom information is requested never has been a patient, the program may acknowledge this fact to the law enforcement officials.

If the law enforcement officials persist in trying to obtain patient information, they should be requested, but not forced, to leave the program premises and the program should immediately consult with its legal counsel. As indicated in item 6, programs should inform local law enforcement officials of the confidentiality restrictions before the officials attempt to obtain patient records. This will avoid crisis, confrontation situations which are likely to arise if the confidentiality restrictions are first communicated to law enforcement officials in the context of a particular investigation and are perceived as limiting their good faith efforts to perform their public responsibilities.

Seizure of Records or Arrest of Program Personnel

If law enforcement officials seize patient records in apparent violation of the Federal confidentiality statutes and regulations or arrest program

- 2 -

personnel because they have refused to disclose patient information which is subject to the Federal confidentiality statutes and regulations, the program's legal counsel should be contacted immediately. In the case of seized records, the program's counsel should consider immediately seeking a court injunction to recover the records and to block the use of any information that the law enforcement officials have obtained from the records. The enclosed (Tab B) complaint filed by the American Civil Liberties Union in San Francisco is illustrative of such an effort.

If a program staff member is arrested or must show cause why he or she should not be held in contempt of court, the program's counsel should immediately inform the court of the prohibition of Federal law which led to the staff member's refusal to provide the information sought and the preeminence of the Federal law over any conflicting State or local law, including the court's compulsory process (See 42 CFR 2.13(b), 2.61; 2.23) and take other appropriate legal action.

At the earliest practicable time following the seizure by law enforcement officials of patient records in violation of the Federal confidentiality statutes and regulations a full report of the incident, including the factual background and the response of program personnel, should be sent to:

Mr. Fleetwood Roberts, Special Projects Branch, NIAAA, Room 11A-02, Parklawn Building, 5600 Fishers Lane, Rockville, Maryland 20857, if an alcohol abuse program is involved; or

Ms. Sheila Gardner, Confidentiality Compliance Specialist, Division of Community Assistance, NIDA, Parklawn Building, Room 9-03, 5600 Fishers Lane, Rockville, Maryland 20857, if a drug abuse program is involved.

This report should describe what immediate steps have been taken to recover any seized records or to take other remedial action and what actions are planned to prevent reoccurrences of the incident. The information provided will be used to determine whether (1) program personnel took all necessary steps to comply with the confidentiality statutes and regulations, (2) an investigation of the incident should be conducted and whether the matter should be referred to the Department of Justice for possible prosecution under the confidentiality statutes and regulations, and (3) the procedures established for handling these incidents should be modified or supplemented to assist other program personnel across the country in avoiding, or better dealing with, similar occurrences.

The alleged violation may also be reported to the local office of the United States Attorney (See 42 CFR 2.7).

- 3 -

Page 3 - Directors, Alcohol and Drug Abuse Treatment Programs

. Page 4 - Directors, Alcohol and Drug Abuse Treatment Programs

5. Use of Legal Counsel: Obtaining Advice and Pursuing Remedies

We emphasize that program staff must rely upon the program's legal counsel and that counsel must become familiar with the requirements of the confidentiality statutes and regulations. Representation of program personnel in court or other legal proceedings must be undertaken by counsel to the program and cannot be performed by HEW, the Department of Justice, or any other agency of the Federal Government. However, oral advice can be obtained on the requirements of 42 CFR Part 2 directly from the HEW Office of General Counsel in those cases in which the program's legal counsel is unavailable and time is of the essence. In these situations, the program may make direct inquiries to Mr. Chris Pascal (301-443- 3096) or Mr. Robert Lanman (301-443-1212) of the HEW General Counsel's Office. Written requests for interpretation of the confidentiality regulations should be directed to Mr. Lanman or Mr. Pascal at the following address: Public Health Division, HEW Office of the General Counsel, Room 4A-53, Parklawn Building, 5600 Fishers Lane, Rockville, Maryland 20857. We suggest that these requests be prepared in consultation with the program's legal counsel. Copies of prior legal opinions interpreting the confidentiality regulations may be obtained from Mr. Roberts or Ms. Gardner at the addresses listed above.

Enclosed at Tab C are pertinent opinions of the HEW Office of the General Counsel interpreting the confidentiality regulations. These opinions should be shared and discussed with the program's legal counsel. They deal with permissible disclosures to law enforcement officials and related matters, including the obtaining of authorizing court orders under 42 CFR Part 2, Subpart E, and the making of cooperative agreements between treatment programs and local law enforcement agencies.

Programs which receive funds from the National Institute on Alcohol Abuse and Alcoholism or the National Institute on Drug Abuse may, under the HEW grants administration regulations, 45 CFR Part 74, use the grant funds to pay the cost of reasonable attorneys' fees incurred for legal advice and assistance in complying with the confidentiality regulations. (See 45 CFR Part 74, Subpart Q, Appendix C, section II.B.16, and Appendix F, sections B2 and G31.) Included in the authorized use of these funds would be the pursuit of legal remedies to recover patient records or to prohibit the use of information gained from patient records in the investigation or prosecution of any patient. It is up to the individual program to determine how much, if any, grant funds it wishes to use for legal services in complying with the confidentiality regulations. However, a determination by the program not to use grant funds in this manner will not be considered an acceptable basis for failure to comply with the confidentiality regulations.

Preventing the Occurrence of Incidents With Local Law Enforcement Agencies Which Lead to Prohibited Disclosures and Uses of Patient Records

6.

We encourage treatment programs and their legal counsel to explore methods for preventing disputes with law enforcement agencies over patient

confidentiality. Sometimes these disputes arise solely from a lack of prior information about the Federal confidentiality requirements and from a misunderstanding of these requirements. One way to prevent this problem is for programs and their counsel to meet with local law enforcement agencies and discuss the Federal confidentiality requirements before an incident occurs. The exchange of information and the potential for education will be enhanced in an environment free from hostility and crisis. Sharing HEW Office of General Counsel opinions with these agencies may also be helpful. especially those which explain the regulatory requirements for obtaining an authorizing court order (See the legal opinions numbered 77-12 and 77-19 at Tab C). Each program's counsel should be consulted for other suggestions for preventing law enforcement disputes over patient confidentiality, including the development of cooperative agreements in this area within the confines of Federal requirements (See the January 24, 1979, legal opinion at Tab C). Requests for technical assistance in developing a good working relationship with law enforcement agencies should be directed to Mr. Roberts or Ms. Gardner at the addresses above.

Director

National Institute on Alcohol Abuse and Alcoholism

Enclosures

Tab A: Regulations, 42 CFR Part 2. Tab B: Complaint filed by the American Civil Liberties Union. Tab C: Legal Opinions (Opinion dated January 24, 1979, and Opinion Nos. 77-29, 77-19, and 77-12).

- 5 -

Page 5 - Directors, Alcohol and Drug Abuse Treatment Programs

William Pollin

Director National Institute on Drug Abuse



TUESDAY, JULY 1, 1975 WASHINGTON. D.C. Volume 40 Number 127 PART IV



DEPARTMENT OF HEALTH, EDUCATION, AND WELFARE

Public Health Service

CONFIDENTIALITY OF ALCOHOL AND DRUG **ABUSE PATIENT RECORDS**

General Provisions

- 6 -

27802

Title 42—Public Health

CHAPTER I-PUBLIC HEALTH SERVICE;" DEPARTMENT OF HEALTH, EDUCATION, AND WELFARE

SUBCHAPTER A-GENERAL PROVISIONS PART 2-CONFIDENTIALITY OF ALCOHOL AND DRUG ABUSE PATIENT RECORDS

On May 9, 1975, the Department of Health, Education, and Welfare and the social behavior. Special Action Office for Drug Abuse Pre-In addition to the foregoing major vention published in the FEDERAL REGISchanges, the following minor policy TER (40 FR 20522) a notice of proposed changes were made. Provisions relating to destruction or 2.11-1 joint rulemaking setting forth a proposed other disposition of records were dropped new Part 2 of Title 42 of the Code of from § 2.21 (§ 2.22 in the May 9 proposal) Federal Regulations governing the confidentiality of alcohol and drug abuse as unnecessary except in the case of programs discontinuing operations. patient records.

Interested persons were invited to sub-The fixed limitation on the permissible duration of written consent for dismit written comments, views, or arguclosure was dropped from § 2.31 in favor ments with respect to the proposed regulations within 30 days of the date of pubof a limitation to such duration as may 2.14-1 Penalty for violations-basis and lication of that notice. All comments so be reasonably necessary to effectuate the submitted were carefully considered, and purpose for which the consent is given. The specification of crimes in § 2.65 at various stages in the rulemaking process, the Administrator of Veterans' Af- for which a court order may be granted fairs and the heads of other Federal de- authorizing use of program records in partments and agencies substantially af- the investigation or prosecution of a pafected by the proposed regulations were tient was broadened to cover any "exconsulted tremely serious" crime, with those listed 2.17-1 Security precautions-basis and pur-As finally adopted and set forth here- in the May 9 notice being retained as

inafter, the regulations contain two ma- examples. jor substantive changes from the May 9 Finally, a number of clarifying, techproposal. The separate treatment of nical, and conforming changes were 2.19 funding sources and third-party payers made in the May 9 proposal, but these (\$\$ 2.21 and 2.37 of the proposed regula- are without significant substantive effect. 2.19-1 Undercover agents and informantstions) was abandoned as unworkable. Accordingly, pursuant to the authority primarily because the prohibitions which of section 408 of the Drug Abuse Office the proposed regulations would have and Treatment Act of 1972, as amended placed on funding sources would have by Pub. L. 92-282 (21 U.S.C. 1175), and directly conflicted with requirements section 333 of the Comprehensive Alcohol which have been proposed in implemen- Abuse and Alcoholism Prevention, Treattation of Title XX of the Social Security ment, and Rehabilitation Act of 1970. Act (see proposed 45 CFR 228.63, 40 FR as amended by Pub. L. 93-282 (42 U.S.C. 16802, 16809, April 14, 1975). In lieu of 4582), and under the authority delegated this approach, § 2.37 has been revised to the General Counsel of the Special to provide that funding sources and Action Office for Drug Abuse Prevention 2.23-1 third-party payers maintaining drug or (39 FR 17901, May 21, 1974), Subchapter alcohol abuse patient records are subject to restrictions upon disclosure to the A of Chapter I, Title 42, Code of Federal same extent and in the same manner as Regulations, is amended, by inserting any other entity maintaining records immediately after Part 1 thereof a new which are within the scope of the au- Part 2 to read as set forth below. thorizing legislation and this Part.

Effective date. These regulations shall The other major change is in the area be effective on August 1, 1975. of criminal justice system referrals, and the grounds for the rules finally adopted Dated: June 25, 1975. are set forth in the basis and purpose section ($\S 2.39-1$) pertaining thereto. In connection with that change, it must be frankly acknowledged that the arguments set forth in the corresponding basis and purpose section ($\S 2.40-1$) of the May 9 proposal have merit. The final rule may in certain instances result in a compromise of the treatment process, if judges or other authorities in the criminal justice system overreact to information whose communication is allowed under the final rules but would have been prohibited under the proposed rules.

D

Against such an adverse effect, however, there must be weighed the very real advantage which genuine cooperation between community social service systems and the criminal justice system can yield for those whose lives are crippled and scarred by the consequences of their own

and the second design of the second second

RULES AND REGULATIONS

criminal conduct. Governmental responses based on a pure medical model have not met with noticeably greater 2.1 success than those based on a purely pu-2.2 nitive approach, and it would be tragic if 2.3 these rules were so constructed as to become a barrier to the development of 2 5 better ways to deal with those who are 2.6 caught up in a pattern of seriously anti-27

R. Moun Acting Assistant Secretary Health, Department Health, Education, and fare.	for
pproved: June 26, 1975.	
CASPAR W. WEINBERGER, Secretary of Health, Educa and Welfare.	tion,
ated: June 27, 1975.	
GRASTY CREWS II, General Counsel, Special Ac Office for Drug Abuse Pre- tion.	etion ven-
ated: June 27, 1975.	
ROBERT L. DUPONT, Director, Special Action O for Drug Abuse Prevention	fice 1.

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

- 7 -

Subpart A----Introductory Statement

Statutory authority-drug abuse.

Statutory authority—alcohol abuse. Previous regulations as controlling authority.

General purposes.

Format.

- Administration and enforcement in general. Reports of violations.

Subpart B-General Provisions

- Definitions and usages-rules. Definitions and usages-basis and purpose.
- 2.12
- Applicability—rules. Applicability—basis and purpose. 2.12 - 1
- General rules regarding confiden-2.13
- tiality-rules. General rules regarding confidenti-2.13-1
- ality-basis and purpose. Penalty for violations-rules. 2.14
 - purpose. Minor patients—rules
- 2 15-1
- Minor patients-basis and purpose. 2.16 Incompetent and deceased patients-rules.
- 2.16-1 Incompetent and deceased patients—basis and purpose. Security precautions—rules. 2.17
- pose. 2.18 Extent of disclosure-rules.
- 2.18-1 Extent of disclosure-basis and pur-
- pose. Undercover agents and informants-
- rules. basis and purpose.
- Identification cards-rules.
- Identification cards-basis and pur-2.20 - 1pose. Disposition of discontinued program 2.21
- records—rules. 2.21-1 Disposition of discontinued program
- records—basis and purpose. Former employees and others—rules. 2.22
- 2.22 1Former employees and others-basis and purpose.
- Relationship to State laws-rules.
- Relationship to State laws—basis and 2.24
 - purpose. Relationship to section 303(a) of Public Health Service Act and section 502(c) of Controlled Sub-
- stances Act—rules. Relationship to section 303(a) of Public Health Service Act and 2.24 - 1section 502(c) of Controlled Sub-stances Act—basis and purpose.

Subpart C-Disclosures With Patient's Consent

- Written consent required-rules. 2.31 2.31-1 Written consent required-basis and
- purpose. Prohibition on redisclosure—rules. 2.82 2.32 - 1Prohibition on redisclosure-basis
- and purpose. 2.33 Diagnosis, treatment, and rehabili-
- tation-rules. 2.33-1 Diagnosis, treatment, and rehabilita-
- tion—basis and purpose. Prevention of certain multiple 2.34
- enrollments-rules. 2.34-1 Prevention of certain multiple
- enrollments-basis and purpose. 2.35 Legal counsel for patient—rules. 2.35-1 Legal counsel for patient—basis and
- purpose. Patient's family and others—rules. 2.36
- 2.36-1 Patient's family and others-basis
- and purpose. 2.37 Third party payers and funding sources-rules.
- 2.37-1 Third party payers and funding sources—basis and purpose.

RULES AND REGULATIONS

§ 408. Confidentiality of patient records. Employers and employment agen-Employers and employment agencies-basis and purpose.

Criminal justice system referrals and functions-rules. Criminal justice system referrals and

functions-basis and purpose. Situations not otherwise provided 2.40 for-rules.

cies-rules.

Sec.

2.38

2.38 - 1

2.39

2.39 - 1

- Situations not otherwise provided 2.40--1 for-basis and purpose.
- Subpart D-Disclosures Without Patient Consent
- 2 51 Medical emergencies-rules.
- Medical emergencies-basis and pur-2.51 - 1pose
- 2.52 Research, audit, and evaluationrules.
- 2.52-1 Research, audit, and evaluationbasis and purpose. 2.53
- Government agencies-rules. Governmental agencies-basis and 2.53-1 DUTDOSE.
- Patient identifying information in 2.54 connection with examinationsriles.
- 2.54-1 Patient identifying information in connection with examinationsbasis and purpose.
- 2.55 Supervision and regulation of narcotic maintenance and detoxification programs-rules.
- Supervision and regulation of nar-2.55-1 cotic maintenance and detoxifica-tion programs-basis and purpose.
- Prohibition on disclosure of patient 2.56 identities from research, audit, or evaluation records-rules.
- Prohibition on disclosure of patient 2.56-1 identites from research, audit, or evaluation records-basis and purpose.

Subpart E-Court Orders

- Legal effect of order-rules. 2.61 2.61-1 Legal effect of order-basis and pur-
- Inapplicability to secondary rec-2.62
- ords-rules. Inapplicability to secondary rec-2.62 - 1ords-basis and purpose,
- Limitation to objective data-rules. 2 63_1 Limitation to objective data-basis
- and purpose. 2,64 Procedures and criteria in general-
- rules. Procedures and criteria in general-2.64 - 1
- basis and purpose. Investigation and prosecution of pa-2.65
- tients-rules. 2.65 - 1Investigation and prosecution of pa-
- tients-basis and purpose. Investigation and prosecution of 2.66
- programs-rules. Investigation and prosecution of 2.66 - 1
- programs-basis and purpose. 2.67 Undercover agents and informants-
- rules. 2.67-1 Undercover agents and informantsbasis and purpose.

Subpart A-Introductory Statement

§ 2.1 Statutory authority-drug abuse.

(a) Statutory provisions effective May 14, 1974. Insofar as the provisions of this part pertain to any program or activity relating to drug abuse education, training, treatment, rehabilitation, or research, such provisions are authorized under section 408 of Pub. L. 92-255, the Drug Abuse Office and Treatment Act of 1972 (21 U.S.C. 1175) as amended by section 303 of Pub. L. 93-282 (88 Stat. 137), That section reads as follows:

(a) Records of the identity; diagnosis, prognosis, or treatment of any patient which are maintained in connection with the performance of any drug abuse prevention function conducted, regulated, or directly or indirectly assisted by any department or agency of the United States shall, except as provided in subsection (c), be confidential and be disclosed only for the purposes and under the circumstances expressly authorized under subsection (h) of this section (b) (1) The content of any record referred

cordance with the prior written consent of the patient with respect to whom such record is maintained, but only to such extent, under such circumstances, and for such nurposes as may be allowed under regulations prescribed pursuant to subsection (g). (2) Whether or not the patient, with re-

spect to whom any given record referred to in subsection (a) of this section is maintained. gives his written consent, the content of such record may be disclosed as follows:

(A) To medical personnel to the extent necessary to meet a bona fide medical emergency

(B) To qualified personnel for the purpose of conducting scientific research, management audits, financial audits, or program evaluation, but such personnel may not identify, directly or indirectly, any individual patient in any report of such research, audit, or evaluation, or otherwise disclose patient identities in any manner.

(C) If authorized by an appropriate order of a court of competent jurisdiction granted after application showing good cause therefor. In assessing good cause the court shall weigh the public interest and the need for disclosure against the injury to the patient, to the physician-patient relationship, and to the treatment services. Upon the granting of such order, the court, in determining the extent to which any disclosure of all or any part of any record is necessary, shall impose appropriate safeguards against unauthorized disclosure.

(c) Except as authorized by a court order granted under subsection (b) (2) (C) of this section, no record referred to in subsection (a) may be used to initiate or substantiate any criminal charges against a patient or to

conduct any investigation of a patient. (d) The prohibitions of this section continue to apply to records concerning any. individual who has been a patient, irrespective of whether or when he ceases to be a patient. (e) The prohibitions of this section do not

apply to any interchange of records-(1) within the Armed Forces or within those components of the Veterans' Administration furnishing health care to veterans.

(2) between such components and the Armed Forces.

(f) Any person who violates any provision of this section or any regulation issued pur-suant to this section shall be fined not more than \$500 in the case of a first offense, and not more than \$5,000 in the case of each subsequent offense.

(g) The Director of the Special Action Office for Drug Abuse Prevention, after consultation with the Administrator of Veterans' Affairs and the heads of other Federal departments and agencies substantially affected thereby, shall prescribe regulations to carry out the purposes of this section. These regulations may contain such definitions, and may provide for such safeguards and procedures, including procedures and criteria for the issuance and scope of orders under subsection (b) (2) (C), as in the judgment of the Director are necessary or proper to effectuate the purposes of this section, to prevent circumvention or evasion thereof, or to facilitate compliance therewith.

(b) Amendments effective June 30, 1975. Effective on the date specified in section 104 of the Drug Abuse Office and Treatment Act of 1972 (June 30, 1975), the first sentence of section 408(g) above, will be amended by striking "Director of the Special Action Office for Drug Abuse Prevention" and inserting in lieu thereof Secretary of Health. Education. and Welfare", and the second sentence of such section will be amended by striking "Director" and inserting "Secretary" in lieu thereof. Also effective on that date. section 408. above, will be further amended by (1) striking out "The" and inserting in lieu thereof "Except as provided in subsection (h) of this section, the" in the first sentence of subsection (g) of such section; and (2) adding at the end of such section the following

new subsection: (h) The Administrator of Veterans' Affairs. through the Chief Medical Director, shall, to the maximum feasible extent consistent with their responsibilities under title 38. United States Code, prescribe regulations making applicable the regulations established by the Secretary under subsection (g) of this section to records maintained in connection with the provision of hospital care, nursing home care, domiciliary care, and medical services under such title 38 to veterans suffering from drug abuse. In prescribing and implementing regulations pursuant to this subsection, the Administrator shall, from time to time, consult with the Secretary in order to achieve the maximum possible coordination of the regulations, and the implementation thereof, which they each prescribe.

§ 2.2 Statutory authority-alcohol abuse.

Insofar as the provisions of this part pertain to any program or activity relating to alcoholism or alcohol abuse education, training, treatment, rehabilitation, or research, such provisions are authorized under section 333 of Pub. L. 91-616, the Comprehensive Alcohol Abuse and Alcoholism Prevention, Treatment and Rehabilitation Act of 1970 (42 U.S.C. 4582), as amended by section 122(a) of Pub. L. 93-282, the Comprehensive Alcohol Abuse and Alcoholism Prevention. Treatment, and Rehabilitation Act Amendments of 1974 (88 Stat. 131). As so amended, that section reads as follows:

CONFIDENTIALITY OF RECORDS

SEC. 333. (a) Records of the identity, diagnosis, prognosis, or treatment of any patient which are maintained in connection with the performance of any program or activity relating to alcoholism or alcohol abuse education, training, treatment, rehabilitation, or research, which is conducted, regulated, or directly or indirectly assisted by any department or agency of the United States shall, except as provided in subsection (e), be confidential and be disclosed only for the purposes and under the circumstances expressly authorized under subsection (b) of this section.

(b) (1) The content of any record referred to in subsection (a) may be disclosed in accordance with the prior written consent of the patient with respect to whom such record is maintained, but only to such extent, under such circumstances, and for such nurposes as may be allowed under regulations prescribed pursuant to subsection (g).

#

- 8 -

27803

(2) Whether or not the patient, with respect to whom any given record referred to in subsection (a) of this section is main-

27804

gency

Attention is called to the interpretatained, gives his written consent, the content (b) Statutory rules fully incorporated. tive regulations, issued by the Special of such record may be disclosed as follows: Although, for convenience of reference, Action Office for Drug Abuse Prevention (A) To medical personnel to the extent the statutory basis for this part is set out necessary to meet a bona fide medical emer-(37 FR 24636, November 17, 1972, as rein full in §§ 2.1 and 2.2; the regulations vised 38 FR 33744, December 6, 1973. (B) To qualified personnel for the purin Subparts B through E of this part are referred to hereinafter in this part as intended to include all of the operative pose of conducting scientific research, manthe "previous regulations"). Those reguagement audits, financial audits, or program statutory provisions. lations have been given a special status evaluation but such personnel may not identify. directly or indirectly, any individual as controlling authority by the provi-§ 2.6 Administration and enforcement sions of section 203(d) of Pub. L. 93-282, patient in any report of such research, audit, in general. or evaluation, or otherwise disclose patient as well as the references in the legisla-It is not contemplated that any partive history of that act to the precedents identities in any manner. ticular agency will be set up specifically (C) If authorized by an appropriate order established under section 408 of Pub. L. to enforce compliance with this part. of a court of competent jurisdiction granted 92-255. Such references appear at page Programs which receive Federal grants after application showing good cause there-11 of House Committee Report No. 93for. In assessing good cause the court shall weigh the public interest and the need for may be monitored for compliance with 759 and at page H3563 of the Congresthis and other applicable Federal law as sional Record for May 6, 1974. The latter disclosure against the injury to the patient, an incident to the grant administration citation is to a detailed analysis of the to the physician-patient relationship, and process. Similarly, FDA inspections of to the treatment services. Upon the granting bill in its final form which was submitted methadone programs will include inspecof such order, the court, in determining the for the Record by its floor manager. tion for compliance with this part, which extent to which any disclosure of all or any Chairman Staggers of the Interstate and is incorporated by reference in the methpart of any record is necessary, shall impose Foreign Commerce Committee, when the adone regulation (21 CFR 310.505). appropriate safeguards against unauthorized bill was up for final action by the House disclosure. § 2.7 Reports of violations. of Representatives.

(c) Except as authorized by a court order

Any violation may be reported to the granted under subsection (b) (2) (C) of this § 2.4 General purposes. ection, no record referred to in subsection United States Attorney for the judicial (a) Policy objectives. The purpose of (a) may be used to initiate or substantiate district in which the violation occurs. any criminal charges against a patient or to the regulations set forth in this part is Violations on the part of methadone proconduct any investigation of a patient. (d) The prohibitions of this section conto implement the authorizing legislation grams may be reported to the regional in a manner that, to the extent practicaoffices of the Food and Drug Administinue to apply to records concerning any inble, takes into account two streams of tration. Violations on the part of a Feddividual who has been a patient, irrespective legal thought and social policy. One has eral grantee or contractor may be reof whether or when he ceases to be a patient. to do with mhancing the quality and ported to the Federal agency monitoring (e) The prohibitions of this section do attractiveness of treatment systems. The not apply to any interchange of recordsthe grant or contract. other is concerned with the interests of (1) within the Armed Forces or within Subpart B----General Provisions patients as citizens, most particularly those components of the Veterans' Administration furnishing health care to veterans, in regard to protecting their rights of § 2.11 Definitions and usages .--- Rules. privacy. Within each stream there are (a) Authorizing legislation. The term (2) between such components and the cross-currents, and it should come as no "authorizing legislation" means section Armed Forces. surprise that areas of turbulence are to 408 of the Drug Abuse Office and Treat-(f) Any person who violates any provibe found at their confluence:

sion of this section or any regulation issued pursuant to this section shall be fined not more than \$500 in the case of a first offense subsequent offense.

(b) Limited purpose. The regulations section 333 of the Comprehensive Alcohol contained in this part are not intended Abuse and Alcoholism Prevention, Trea to direct the manner in which substanand not more than \$5,000 in the case of each ment, and Rehabilitation Act of 1970 (42 tive functions, such as research, treat-U.S.C. 4582), as such sections may be ment, and evaluation, should be carried (g) Except as provided in subsection (h) amended and in effect from time to time. out, but rather to define the minimum of this section, the Secretary shall prescribe (b) Construction of terms. The definirequirements for the protection of conregulations to carry out the purposes of this section. These regulations may contain such tions and rules of construction set forth fidentiality of patient records which must in this section are applicable for the purdefinitions, and may provide for such safebe satisfied in connection with the conposes of this part. To the extent that they guards and procedures, including procedures duct of those functions in order to carry refer to terms used in the authorizing and criteria for the issuance and scope of out the purposes of the authorizing orders under subsection (b) (2) (C), as in legislation, they are also applicable for legislation. This does not mean that obthe judgment of the Secretary are necessary the purposes of such legislation. servance of only the minimum legal reor proper to effectuate the purposes of this (c) Alcohol abuse. The term "alcohol section, to prevent circumvention or evasion quirements is always the wisest course. abuse" includes alcoholism. but in framing these regulations, allowthereof, or to facilitate compliance there-(d) Drug abuse. 'The term "drug abuse' ance has necessarily been made for a with includes drug addiction. (h) The Administrator of Veterans' Affairs. diversity of emphasis and approach in (e) Diagnosis and treatment. The through the Chief Medical Director, shall, to the many different jurisdictions and by terms "diagnosis" and "treatment" inthe maximum feasible extent consistent with the great variety of public and private clude interviewing, counselling, and any their responsibilities under title 38. United agencies which must find a way to func-States Code, prescribe regulations making other services or activities carried on for tion within the limits here prescribed. applicable the regulations prescribed by the the purpose of or as an incident to diag-Secretary under subsection (g) of this secnosis, treatment, or rehabilitation with § 2.5 Format. tion to records maintained in connection respect to drug abuse or alcohol abuse, (a) Basis and purpose sections. Each with the provision of hospital care, nursing home care, domiciliary care, and medical whether or not conducted by a member section setting forth rules on any given of the medical profession.

services under such title 38 to veterans suftopic in Subparts B through E of this (f) Program. fering from alcohol abuse or alcoholism. In part is followed by a section setting forth prescribing and implementing regulations (1) The term "program", when referpursuant to this subsection, the Administratheir basis and purpose. In many cases, ring to an individual or organization, tor shall, from time to time, consult with the basis and purpose section is itself means either an individual or an orgathe Secretary in order to achieve the maxian interpretative rule regarding the legal nization furnishing diagnosis, treatment, mum possible coordination of the regulations, and the implementation thereof, authority of the rulemakers. In other or referral for alcohol abuse or drug which they each prescribe, instances, it summarizes historical or abuse.

RULES AND REGULATIONS

authority.

§ 2.3 Previous regulations as controlling evidentiary material relevant to the validity and interpretation of the section which precedes it.

ment Act of 1972 (21 U.S.C. 1175) and

- 9 -

used in the sense defined in paragraph cal, accounting, or other professional coverage or similar arrangements) evi-(f)(1), means a plan or procedure, services. whether functional or organizational. and whether or not governmental, for dealing with alcohol abuse or drug abuse problems from either an individual or a social standpoint.

(g) Program evaluation

The term "program evaluation" means an evaluation of-

(1) The effectiveness, efficiency, compliance with applicable therapeutic. legal, or other standards, or other aspects of the performance, of a program as defined in paragraph (f)(1) of this section, or

(2) The validity, effectiveness, efficiency, practicability, or other aspects of the utility or success of a program in the sense defined in paragraph (f)(2)of this section.

(h) Program director. The term "program director" in the case of a program which is an individual means that individual, and in the case of a program which is an organization, the individual, if any, who is the principal, or, in the case of organizations consisting of partners or under the control of a board of directors, board of trustees or other governing body, the individual designated as program director, managing director, or otherwise vested with executive authority with respect to the organization.

(i) Patient. The term "patient" means any individual (whether referred to as a patient, client, or otherwise) who has applied for or been given diagnosis or treatment for drug abuse or alcohol abuse and includes any individual who, after arrest on a criminal charge, is interviewed and/or tested in connection with drug or alcohol abuse preliminary to a determination as to eligibility to participate in a treatment or rehabilitation program.

(i) Patient identifying information. The term "patient identifying information" means the name, address, social security number, or similar information by which the identity of a patient can be determined with reasonable accuracy and speed either directly or by reference to other publicly available information. The term does not include a patient identifying number assigned by a program.

(k) Alcohol abuse or drug abuse prevention function. The term "alcohol abuse or drug abuse prevention function" means any program or activity relating to alcohol abuse or drug abuse education, training, treatment, rehabilitation, or research, and includes any such function even when performed by an organization whose primary mission is in the field of law enforcement or is unrelated to alcohol or drugs.

(1) The term "person" means an individual, a partnership, a corporation, a trust, a Federal or State governmental agency, or any other legally cognizable entity.

(m) Service organization. The term "service organization" means a person is on the basis of an individual relation-

(n) Qualified service organization. The

term "qualified service organization" means a service organization which has entered into a written agreement with a organization-

(1) acknowledges that in receiving, about patients in the program, it is fully bound by the provisions of this part:

(2) undertakes to institute appropriate procedures for safeguarding such in- categories of eligible persons. formation, with particular reference to patient identifying information; and

(3) undertakes to resist in judicial proceedings any efforts to obtain access to information pertaining to patients otherwise than as expressly provided for in this part.

(o) Records. The term "records" includes any information, whether recorded or not, relating to a patient, received or acquired in connection with the performance of any alcohol abuse or drug abuse prevention function, whether such receipt or acquisition is by a program, a qualified service organization, or any other person.

(p) Communications not constituting disclosure. The following types of communications do not constitute disclosures of records:

(1) Communications of information within a program between or among personnel having a need for such information in connection with their duties.

(2) Communications between a program and a qualified service organization of information needed by the organization to perform its services to the program.

(3) Communications of information which includes neither patient identify-ing information nor identifying numbers assigned by the program to patients.

(q) Previous regulations. The term 'previous regulations" refers to the interpretative regulations issued by the Special Action Office for Drug Abuse Prevention, originally published November 17, 1972, 37 FR 24636, as revised December 6, 1973, 38 FR 33744.

(r) State law. The term "State law" refers to the law of a State or other jurisdiction, such as the District of Columbia, as distinguished from Federal law in general. As applied to transactions which do not take place in any State or other similar jurisdiction, the term refers to Federal common law as modified by any applicable Federal statutes and regulations.

(s) Third party payer. The term "third party payer" means any organization (or person acting as agent or trustee for an organization or fund) which pays or agrees to pay for diagnosis or treatment furnished or to be furnished to a particular individual, where such payment or agreement to pay

(2) The term "program", when not tion, laboratory analyses, or legal, medi- the case of self-and-family insurance denced by a contract, an insurance policy, a certificate of membership or participation, or similar documentation.

(t) Funding source. The term "funding source" means any individual or any program pursuant to which the service public or private organization, including any Federal, State, or local governmental agency, which makes payments in supstoring, processing, or otherwise dealing port of a program. A funding source is with any information from the program not, as such, a third party payer, even where its payment sare based directly or indirectly on the program's patient load with or without respect to specified

(u) August 22, 1974 draft. References to the "August 22, 1974 draft" are to the draft regulations set out in the Advance Notice of Proposed Joint Rulemaking published in the FEDERAL REGISTER on August 22, 1974, 39 FR 30426, by the Department of Health, Education, and Welfare and the Special Action Office for Drug Abuse Prevention.

§ 2.11-] Definitions and usages.-Basis and purpose.

(a) In general. The definitions are based upon the legislative history of and experience with the authorizing legislation, and are intended as aids to construing the provisions of this part to carry out the purposes of those statutes. (b) Coverage of applicants for treat-

ment. Section 2.11(i) is intended to make it clear that records of the identity and other information about a person whose application is rejected or withdrawn are fully as much covered by this part as records pertaining to a patient actually accepted for treatment.

(c) Program terminology for patients not controlling. While many programs prefer to use "client" or some other term instead of "patient" to describe the recipients of their services, it is believed preferable to use terminology in this part which is consistent with that used in the authorizing legislation. It should be clearly understood, however, that the records of any individual who fits the definition set forth in § 2.11(i) are covered, no matter what terminology the program may use to designate his status.

(d) Origin of "prevention function" terminology. The definition of alcohol abuse or drug abuse prevention function in § 2,11(k) is adapted from the definition of drug abuse prevention function in section 103(b) of the Drug Abuse Office and Treatment Act of 1972 (21 U.S.C. 1103(b)). Although there was no corresponding defined term available to the draftsman of the 1974 amendment to section 333 of the Comprehensive Alcohol Abuse and Alcoholism Prevention. Treatment, and Rehabilitation Act of 1970 (42 U.S.C. 4582), it is clear from the legislative history that the coverage of alcohol abuse patient records was intended to be fully as wide as the coverage of drug abuse patient records, and the definition in § 2.11(k) reflects that intention.

(e) "Ambiguity of the term "program". which provides services to a program ship between the payer and the patient It is recognized that it is ordinarily poor such as data processing, dosage prepara- (or a member of the patient's family in drafting technique to use the same term

27806

related. as those in §§ 2.11(f)(1) and ponents of the Veterans' Administration the United States". In the light of the 2.11(f) (2). This part, however, has to be .. furnishing health care to veterans, or be- multiplicity and extent of Federal proread both in conjunction with the Food tween such components and the Armed and Drug Administration's Methadone Forces, of records pertaining to a per-Regulation and the Drug Abuse Office son relating to a period when such per- grams, this wording strongly suggests an and Treatment Act of 1972. The Methadone Regulation (21 CFR 310.505) clearly uses the term "program" in the § 2.11(f)(1) sense. In section 103(b) of the Act (21 U.S.C. 1103(b)), it is clearly used in the $\S 2.11(f)(2)$ sense, and the usage in section 408(b) (2) (B) of the Act person within the Armed Forces. has from its original enactment been administratively interpreted to include both senses. As used in this part, the context should indicate the intended meanings with sufficient clarity to make this preferable to creating and defining new terminology which would be different from that used in related regulations and the authorizing legislation.

(f) Construction of disclosures, Section 2.11(p) is intended to clarify the status of communications which are carried on within a program or between a program and persons or organizations which are assisting it in providing patient care. The authorizing legislation was not intended to prohibit programs from carrying on accepted practices in terms of obtaining specialized services from outside organizations. In conjunction with the definition of qualified service organizations, set forth in $\S 2.11(n)$, the provisions of § 2.11(p) should prevent the development of abuses in this area.

§ 2.12 Applicability.-Rules.

(a) In general. Except as provided in paragraph (b) of this section, this part applies to records of the identity, diagnosis, prognosis, or treatment of any patient which are maintained in connection with the performance of any alcohol abuse or drug abuse prevention function-

(1) Which is conducted in whole or in part, whether directly or by grant, contract, or otherwise, by any department or agency of the United States.

(2) For the lawful conduct of which in whole or part any license, registration. application, or other authorization is required to be granted or approved by any department or agency of the United States.

(3) Which is assisted by funds supplied by any department or agency of the United States, whether directly through a grant, contract, or otherwise, or indirectly by funds supplied to a State or local government unit through the medium of contracts, grants of any descrip- § 2.12-1 Applicability.-Basis and purtion, general or special revenue sharing, or otherwise, or

(4) Which is assisted by the Internal Revenue Service of the Department of the Treasury through the allowance of income tax deductions for contributions to the program conducting such function, or by a way of a tax-exempt status for such program.

(b) Armed Forces and Veterans' Administration.

27805

RULES AND REGULATIONS

son is or was subject to the Uniform Code of Military Justice.

(2) Except as provided in paragraph (b) (1) of this section, this part applies to any communication between any nerson outside the Armed Forces and any

(3) Except as provided in paragraph (b) (1) of this section, this part applies, insofar as it pertains to any drug abuse prevention function, to any communication between any person outside those components of the Veterans' Administration furnishing health care to veterans and any person within such components. until such date as the Secretary of his authority (conferred by an amendment effective June 30, 1975) to prescribe regulations under section 408 of Pub. L. 92-255 (21 U.S.C. 1175). After such date. this part applies thereto to such extent. as the Administrator of Veterans' Aftor, by regulation makes the provisions of this part applicable thereto.

(4) Except as provided in paragraph (b) (1) of this section, this part applies, insofar as it pertains to any alcohol abuse prevention function, to any communication between any person outside those components of the Veterans' Administration furnishing health care to components, to such extent as the Administrator of Veterans' Affairs, through the Chief Medical Director, by regulation makes the provisions of this part applicable thereto.

(c) Period covered as affecting applicability. The provisions of this part apply to records of identity, diagnosis, prognosis, or treatment pertaining to any given individual maintained over any period of time which, irrespective of when it begins, does not end before March 21, 1972, in the case of diagnosis or treatment for drug abuse or before May 14, 1974, in the case of diagnosis or treatment for alcohol abuse.

(d) Applicability determined by nature and purpose of records. The applicability of the provisions of this part is determined by the nature and purpose of the records in question, and not by the status or primary functional capacity of the recordkeeper.

pose.

(a) The broad coverage provided by § 2.12(a) is appropriate in the light of the remedial purposes of the statutes as well as the practical desirability of certainty and uniformity. Sections 2.12(a) (1) and 2.12(a) (2) simply follow the terms of subsection (a) of the statutes. with some explanatory material for the sake of clarity and explicitness.

(b) Sections 2.12(a) (3) and 2.12(a) (1) The provisions of this part do not (4) are based upon the use by Congress apply to any interchange, entirely with- of the phrase "directly or indirectly as-

in senses which are as different, yet in the Armed Forces, within those com- sisted by any department or agency of grams and policies which can be of assistance to drug and alcoholism prointention to provide the broadest coverage consistent with the literal terms of the statutes. Many programs commence with direct Federal assistance, financial, technical, or both, and later continue with State aid and private, tax-deductible contributions. It would be manifestly contrary to the general policy sought to be effectuated by the legislation if the confidential status of a program's records were to terminate, or even be called into question, by the cessation of direct Federal assistance.

(c) With regard to $\S 212(a)(3)$, it seems clear that whenever a State or local government is assisted by the Fed-Health Education and Welfare exercises eral government by way of revenue sharing or other unrestricted grants, all of the programs and activities of the State or local government are thereby indirectly assisted, and thus meet that aspect of the statutory criteria for coverage.

(d) Section 2.12(a)'(4) follows the fairs, through the Chief Medical Direc- doctrine established in McGlotten v. Connally, 338 F. Supp. 448 (D.C. D.C., 1972), in which it was held that the deductible status of contributions to an organization constitutes "Federal financial assistance" within the meaning of section 601 of the 1964 Civil Rights Act (42 U.S.C. 2000d). The inclusion of the adjective "indirect" as a modifier of the term "assistance" as used in the proviveterans and any person within such sions of law authorizing this part suggests an intention to provide coverage at least as broad, if not broader than, section 601 of the Civil Rights Act in respect of financial assistance. See, also, Green v. Connally, 330 F. Supp. 1150 (D.C. D.C., 1971) aff'd sub. nom. Coit v Green, 404 U.S. 997, 92 S. Ct. 564, 30 L. Ed. 2d 550 (1971)

> (e) Section 2:12(b) essentially repeats the interpretation given in § 1401.02(b) of the previous regulation except that it . takes account of the special provisions. inserted in the new law with reference to the Veterans Administration, and makes clear that the exemption for communications within the military-VA system does not generally apply to records pertaining to civilians.

(f) Section 2.12(c), which deals with the question of how the period covered by any given set of records affects the applicability of these regulations to them, restates the principle set forth in § 1401.-02(a) of the previous regulations, and applies it to records in the field of alcohol abuse as well as drug abuse. The authorizing legislation contains no effective date provisions. A construction which would apply the statutes to records of completely closed treatment episodes, records necessarily made and maintained prior to the enactment of the legislation, would create serious administrative problems. It seems doubtful, in any case, whether such records have been "maintained," within the meaning of the statutes, during any period of time after their enactment. On the other hand, ff

5

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

- 11 -

enactment of the applicable statute, then all the records should be covered irrespective of when treatment was begun, because such records clearly are being "maintained" after the enactment of the legislation.

(g) Section 2.12(d) has been included to make explicit one of the legal impli- sary under the circumstances. cations of the authorizing legislation, which is cast in terms descriptive of the records which are to be confidential rather than of the recordkeepers on whom a duty is thus imposed. The result is that, for example, where a State attending a program, whether over a agency maintains an individual client period of time or on a particular occarecord which contains identifying information about a client (i.e., patient) re- hibitions and conditions of this part as ceiving treatment or rehabilitation services for drug abuse, such a record is clearly a record maintained in connection with a drug abuse prevention function, and is subject to the provisions of this part. The fact that the record may also be required by statute or regulations pertaining to eligibility for Federal Financial Participation would in no way exempt the record from the prohibitions and requirements of this part. Thus, it would be unlawful and a violation of these regulations for such a record to be made available to a law enforcement agency, or to determine (without the prior written consent of the client) eligibility for other welfare benefits, or for any other administrative or investigative uses or purposes which would involve or result in an identification of the client to a third § 2.13-1 General rules regarding confiparty.

§ 2.13 General rules regarding confi-dentiality.—Rules.

(a) In general. Records to which this part applies shall be confidential and may be disclosed only as authorized by this part, and may not-otherwise be divulged in any civil. criminal, administrative, or legislative proceeding conducted by any Federal, State, or local authority, whether such proceeding is commenced before or after the effective date of this part.

(b) Unconditional compliance required. The prohibition upon unauthorized disclosure applies irrespective of whether the person seeking disclosure already has the information sought, has other means of obtaining it, enjoys official status, has obtained a subpoena, or asserts any other justification or basis for disclosure not expressly authorized under this part.

(c) Information covered by prohibition. The prohibition on unauthorized disclosure covers all information about patients, including their attendance or absence, physical whereabouts, or status as patients, whether or not recorded, in the possession of program personnel, except as provided in paragraph (d) of this section.

(d) Crimes on program premises or against program personnel. Where a patient commits or threatens to commit a crime on the premises of the program or against personnel of the program, nothing in this part shall be construed as prohibiting personnel of the program the first the sea

agency, but such report shall not identify the suspect as a patient. In citation, if given by an institution or any such situation, immediate consideration should be given to seeking an order under Subpart E of this part to permit the disclosure of such limited information about the patient as may be neces-

(e) Implicit and negative disclosures prohibited. The disclosure that a person (whether actual or fictitious) answering to a particular description, name, or other identification is not or has not been sion, is fully as much subject to the proa disclosure that such a person is or has been attending such a program. Any improper or unauthorized request for any subsequent offense. disclosure of records or information subject to this part must be met by a noncommittal response.

(f) In-patients and residents. The presence of any in-patient in a medical facility or resident in a residential facility for the treatment of drug or alcohol abuse may be acknowledged to callers and visitors with his written consent. Without such consent, the presence of any in-patient or resident in a facility for the treatment of a variety of conditions may be acknowledged if done in such a way as not to indicate that the patient is being treated for drug or alcohol abuse.

dentiality .- Basis and purpose.

(a) Section 2.13(a) enunciates the general principle of the statutory provisions, and is unchanged from § 1401.03 of the previous regulations.

(b) Sections 2.13(b) and 2.13(c) have been added on the basis of written comments on the draft regulations published August 22, 1974, in which there was a documented report that counsel for a program had advised the program that it could furnish information to the FBI about patients without their written consent and without completing a full judicial proceeding in accordance with Subpart E of this part. Sections 2.13(b) and 2.13(c) should clarify the original intent of the statutes and regulations to the extent of precluding such errors in the future.

(c) In the situation described in § 2.13(d), the desirability of the general prophylactic rule prohibiting disclosures by program personnel about patients regardless of whether such disclosures are from a written record must yield to the practical necessity to permit protection from, and prompt reporting of, criminal acts. In the preface to the first set of regulations issued under 21 U.S.C. 1175. it was emphasized that the operation of that section "in no way creates a sanctuary for criminals." (37 FR 24636, November 17, 1972). Section 2.13(d) is consistent with that contemporaneous administrative construction.

(d) Section 2.13(e) is adapted from § 1401.11 of the August 22, 1974 draft. The suggestion that this part be cited from seeking the assistance of, or re- when declining to give information has

- 12 -

treatment is actually carried on after the porting such crime to, a law enforcement been deleted on the basis of comments that correctly pointed out that such a program maintaining some records covered by this part and some not, would serve to identify the records inquired about as pertaining to treatment covered by this part.

27807

Section 2.13(f) merely clarifies the effect of the preceding paragraphs in the special situations to which paragraph (f) relates

§ 2.14 Penalty for violations.-Rules.

(a) Penalty provided by law. Any person who violates any provision of the authorizing legislation or any provision of this part shall be fined not more than \$500 in the case of a first offense, and not more than \$5,000 in the case of each

(b) Application to subsequent offenses. Where a defendant has committed one offense under either section authorizing this part or any provision of this part authorized by that section, any offense thereafter committed under the same section or any provision of this part authorized under that section shall be treated as a subsequent offense.

§ 2.14-1 Penalty for violations.-Basis and purpose.

(a) Section 2.14 states the criminal penalty provided for in subsection (f) of the sections authorizing this part. It is included in this part for convenience and completeness. Some of the comments received on this section when originally proposed suggested that criminal penalties for violation should include imprisonment, but such a change would have to be made by legislation rather than rulemaking.

(b) Section 2.14(b) clarifies the intention that the "subsequent offense" need not be identical to the first offense, as long as it is committed with respect to the same statutory section. For example, a person whose first offense had consisted of improperly releasing the name of a patient in an alcoholism treatment program would be punishable for a "subsequent offense" if he later gives out information from the diagnostic work-up of an alcoholism patient.

§ 2.15 Minor patients .--- Rules.

(a) Definition of minor. The term "minor" means a person who has not attained the age of 18 years or, in a State where a different age is expressly provided by State law as the age at which a person ceases to be a minor, the age prescribed by the law of such State.

(b) Consent to disclosure in general. Except as provided in paragraph (c), where consent is required for any disclosure under this part, such consent in the case of a minor must be given by both the minor and his parent, guardian, or other person authorized under State law to act in his behalf, but any disclosure made after the patient has ceased to be a minor may be consented to only by the patient.

(c) Rule when State law authorizes treatment without parental consent. Whenever a patient, acting alone, has the

÷.,

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

legal capacity under the applicable State by subsection (g) of the authorizing law to apply for and obtain such diagnosis, counselling, administration of medication, or other services as actually are or were provided to him by the program with respect to which he is or was a patient, any consent required for disclosure under this part may be given only by the patient, notwithstanding the fact that the patient may be a minor.

27808

(d) Initial contacts. When a minor applies for services under circumstances policy. other than those described in paragraph (c) of this section the fact of such anplication may not be disclosed, except as an incident to a communication authorized under paragraph (f) of this section, without consent of the applicant, to the applicant's parent, guardian, or other person authorized under State law to act on behalf of the applicant. When without it.

Section 2.16 essentially repeats the such an applicant refuses consent, it over, should not be frustrated by atsubstance of § 1401.04 of the previous must be explained to the applicant that tempts to enforce parental financial reregulations, broadened to reflect the fact while he or she has the right (subject sponsibility in a situation where the to the provisions of paragraph (f) of this that the statutes now allow any con-State itself has determined that the section) to withhold such consent, the sensual disclosures permitted by the regminor should have a right to obtain ulations, and to cover the situation of services applied for cannot be provided services without involving the parent. deceased patients for whom no formal (d) A much more difficult problem is appointment of an executor, administra-(e) Collection or attempted collection presented in the case of a minor who apof payment for services. Where State tor, or other personal representative has plies for services in a jurisdiction which law authorizes the furnishing of services been made. Written comments were rehas not determined that a minor should ceived to the effect that the power to to a minor without the consent of the have the right to obtain them without consent to disclosure in the case of a minor's parent or guardian, no inquiry parental knowledge or consent. The may be made of the parent's or guarddeceased patient should be limited to a question may arise as to whether the ian's financial responsibility, and no bill, personal representative. The expense of clinician has an ethical or legal duty to probate or administration in some jurisstatement, request for payment, or any notify the parent which conflicts with a dictions could cause financial hardship other communication in respect of such duty of nondisclosure. The rules in § 2.15 to survivors, and on balance it is believed services may be transmitted directly or are based upon the theory that Federal indirectly to such parent or guardian. that where the assets of an estate are law should not invalidate a State policy without the express written consent of insufficient to justify the appointment which prohibits treatment without of a personal representative, the public the patient. Such consent may not be parental consent, but that keeping conmade a condition of the furnishing of interest is served by permitting others to fidential a mere application for treatservices except in the case of a program consent to disclosure. ment is not ordinarily a sufficient transwhich is not required by law, and does gression of such a State policy as to re-§ 2.17 Security precautions.-Rules. not in fact hold itself out as willing, to quire an exception to the general Federal (a) Precautions required. Approprifurnish services irrespective of ability policy prohibiting disclosure of an appliate precautions must be taken for the to pay. cation for services without the consent security of records to which this part (f) Applicant lacking capacity for of the applicant.

applies. Records containing any inforrational choice. When, in the judgment (e) Section 2.15(f) deals with the case mation pertaining to patients shall be of a program director a minor applicant of the minor applicant who lacks the cakept in a secure room, or in a locked file for services, because of extreme youth or pacity to make a rational choice about cabinet. safe, or other similar container. mental or physical condition, lacks the consenting to disclosure. It is based upon when not in use. capacity to make a rational decision on the theory that where a person is acwhether to consent to the notification (b) Policies and procedures. Dependtually as well as legally incapable of acting in his own interest, disclosures to ing upon the type and size of the proof a parent or guardian, and the situagram, appropriate policies and procetion of the applicant poses a substantial a person who is legally responsible for dures should be instituted for the further threat to the life or physical well being him may be made to the extent that the security of records. For example, except of the applicant or any other individual. best interests of the patient clearly so where this function is personally perand such threat might be reduced by require. Any other rule could subject formed by the program director, a single communicating the relevant facts to a clinicians to an intolerable choice bemember of the program staff should be parent or guardian of the applicant. tween violating the provisions of this designated to process inquiries and resuch facts may be so communicated by part on the one hand, or failing to take quests for patient information, and a the program director or by program peraction to avoid a preventable tragedy written procedure should be in effect sonnel authorized by the director to do involving a minor, on the other. The regulating and controlling access by statutes authorizing this part should not those members of the staff whose rebe read as requiring such a choice. § 2.15-1 Minor patients .- Basis and sponsibilities require such access, and purpose. § 2.16 Incompetent and deceased paproviding for accountability.

(a) The statutes authorizing this part § 2.17-1 Security precautions.-Basia are totally silent on the issue of the (a) Incompetent patients other than and purpose. capacity of a minor to give consent for minors. Where consent is required for The enormous variations in both the disclosures, and there is nothing in the any disclosure under this part, such consize and the type of programs to which sent in the case of a patient who has this part is applicable preclude the legislative history to suggest that the been adjudicated as lacking the capac- formulation of specific requirements question was ever considered by Congress. The question is, however, one ity, for any reason other than insuffiwith respect to the physical security of cient age, to manage his or her own afwhich arises repeatedly, and it is thererecords. Almost any requirement which fairs may be given by the guardian or fore appropriately addressed under the other person authorized under State law could be laid down would, under some general rulemaking authority conferred to act in the patient's behalf. circumstances, either be impracticable or

- 13 -

RULES AND REGULATIONS

legislation.

(b) Perhaps no legal issues are more highly charged than those affecting the relationship of parent and child. Since this part, such consent in the case of Congress has not evidenced an intention records of a deceased patient may be to affect this relationship, it is clear that local law should govern and the task of rulemaking is limited to that of insuring. is no appointment of a personal repreas far as possible, that the results under Federal law are consistent with local

(c) Where a State has authorized the family. furnishing of treatment or other services of a given type to a minor without notice to or consent by the parent or guardian, it seems clear that a consistent Federal policy with respect to disclosure requires that consent for any disclosure of the treatment record be given by the minor. This policy, more-

tients .--- Rules.

(b) Deceased natients

(1) In general. Except as provided in paragraph (b) (2) of this section, where consent is required for any disclosure of given by an executor, administrator, or other personal representative. If there sentative, such consent may be given by the natient's spouse or if none, by any responsible member of the patient's

(2) Vital statistics. It the case of a deceased patient, disclosures required under Federal or State laws involving the collection of death and other vital statistics may be made without consent.

§ 2.16-1 Incompetent and deceased patients.-Basis and purpose.

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

perverse in its effects. For example, in ized by a court order granted under a facility handling a variety of medical records, all of which are confidential and so marked, a requirement that those pertaining to drug or alcohol treatment be marked in any distinctive way would merely serve to identify such records as pertaining to drug or alcohol treatment-precisely the opposite of the intended result. The purpose of § 2.17, which is based upon § 1401.25 of the previous regulations, is to alert programs to the necessity of exercising due care with respect to the security of patient records.

§ 2.18 Extent of disclosure.--Rule.

Any disclosure made under this part, whether with or without the patient's solely for the purpose of enabling the consent, shall be limited to information necessary in the light of the need or purpose for the disclosure.

§ 2.18-1 Extent of disclosure .- Basis and purpose.

(a) Section '2.18 expresses the general principle, which has application in many different contexts, that any disclosure from records covered by this part should the previous regulations.

(b) This section should not be misunderstood as imposing a limitation on the scope of records which may or should be made available to health agencies conducting inspections as described in § 2.55. All of the records maintained by a program may be relevant to such inspection. The Congress has determined that disclosure under such circumstances is not a violation of the statutes authorizing this part; where such disclosure is required by Federal or State law, and the inspecting agency is a qualified State health agency as defined in § 2.55(e) (1), it becomes the responsibility of that agency to protect the confidentiality of information it acquires in the course of its lawful activities.

§ 2.19 Undercover agents and informants.----Rules.

(a) Definitions. As used in this section, § 2.19-1, and §§ 2.67 and 2.67-1,-(1) The term "undercover agent" means a member of any Federal, State, or local law enforcement or investigative agency whose identity as such is concealed from either the patients or personnel of a program in which he enrolls or attempts to enroll.

(2) The term "informant" means a person who, at the request of a Wederal. State, or local law enforcement or investigative agency or officer, carries on observation of one or more persons enrolled in or employed by a program in which he is enrolled or employed, for the purpose of reporting to such agency or officer information concerning such persons which he obtains as a result of such observation subsequent to such request

(b) General prohibition. Except as

\$ 2.67,-(1) No undercover agent or informant may be employed by or enrolled in any alcohol or drug abuse treatment program:

(2) No supervisor or other person having authority over an undercover agent enforcement and investigative agencies may knowingly permit such agent to be which are affected by § 2.19, there is as or remain employed by or enrolled in any such program; and

(3) No law enforcement or investiga- rectness. tive officer may recruit or retain an informant with respect to such a program. (c) Exceptions. The enrollment of a

law enforcement officer in a treatment program shall not be deemed a violation of this section if (1) such enrollment is officer to obtain treatment for his own abuse of alcohol or drugs, and (2) his status as a law enforcement officer is known to the program director.

§ 2.19-1 Undercover agents and informants.-Basis and purpose.

(a) In many instances, persons who making their first tentative efforts to-Moreover, it would appear that the purpose of such agents or informants, may be to obtain precisely the type of personal information which might be revealed by inspection of counselor notes and other patient records maintained by program, either as a patient or as an employee, would appear to be contrary to the purposes for which the provisions of law authorizing this part were enacted, and properly subject to prohibition under regulations expressly authorized to carry out those purposes.

(b) From a policy standpoint, § 2.19 is circumstances where their use may be justified. Accordingly, where a showing is made in an application for an order under § 2.67 that the criteria set forth in that section are satisfied, the court may grant such an order.

(c) When this section of the regulations was proposed, numerous written comments were received urging that there be an absolute prohibition on the use of undercover agents and informants, and most of the witnesses at the hearings who addressed the issue at all testified to the same effect. A number of comments were received to the effect that § 2.19 should be dropped altogether, but this request was always clearly and often explicitly predicated on the assumption that failure to say anything about undercover agents and informants statutes, standing alone, do not prohibit the practice, and thus that in the absence of a specific prohibition in these regulations, the use of undercover agents anu informants in treatment programs would not be unlawful. Since this is a view which we believe to be shared by the law a practical matter no alternative to predicating these regulations upon its cor-

(d) However desirable it may be to limit the use of undercover agents and informants in treatment programs, we think a strong argument can be made against our power to impose an absolute prohibition. To the extent that the practice is susceptible to regulation through the rulemaking process at all, it is on the theory that it opens the way to disclosure of information which is or should be in program records, and thus is contrary to the purposes of the statutes. Since subsection (g) of the statutes confers express rulemaking authority to carry out these purposes, regulation of are patients in treatment programs are the use of undercover agents and inbe limited to information necessary in ward re-integration into productive so- ercise of that authority. But even the the light of the need or purpose for the ciety. They may be both vulnerable and express statutory prohibition against disuspicious, and the presence in a treat- rect disclosure of the content of patient ment program of undercover law enforce- records is subject to the power of the ment agents or informants can have a courts to authorize such disclosure under devastating effect on the program's subsection (b) (2) (C) of the statutes. It morale and therapeutic effectiveness, seems difficult to argue that Congress intended to confer on rulemaking agencies the authority to impose an absolute prohibition even though its own restrictions (other than those on disclosures of patient identities from secondary records) are subject to being set aside by court the program. Thus, the placing of an order in particular cases. Since we have undercover agent or informant in a not attempted to exercise such an authority, it is not necessary to decide at this time whether it was conferred. (e) A careful reading of the definitions

set forth in § 2.19(a) is crucial to an understanding of the prohibitions which are imposed by § 2.19. Objections to the section were made informally but vigorously on behalf of the Drug Enforcement based on the reasoning that while the Administration, on the ground that the use of undercover agents and informants testimony of informants or undercover in treatment programs is ordinarily to agents is frequently if not normally be avoided, there may occasionally arise essential to the successful prosecution of cases arising under the Controlled Substances Act. It was said that in the form originally proposed, the section would cut off from treatment those who might agree to cooperate with law enforcement authorities, a result both inhumane and counterproductive. As the definition of an informant is intended to make clear. however, it is his function vis-a-vis personnel and fellow patients in the program in which he is enrolled which is controlling, and not his relationship, per se, with an investigative agency.

(f) Finally, the definition of informant is intended to clarify the distinction between an informant and an ordinary witness. It is the element of prearrangement which is crucial. In one of the comments received on \$2.19 as proposed, it was urged that treatment programs should be otherwise provided in paragraph (c) of would make their use illegal. Our view considered as sanctuaries, but such a this section, or as specifically and lor- is to the contrary: we think that the result was explicitly disclaimed in the

initial publication of the previous regula- § 2.21 Disposition of discontinued protions (37 FR 24636). In so saying, we are by no means insensitive to the anxieties repeatedly expressed in both testimony. and comments on this section, but we believe that the prohibition contained in § 2.19 and the procedures and criteria set forth in § 2.67 provide a measure of relief which is consistent with the structure and intent of the underlying statutes.

27810

§ 2.20 Identification cards .-- Rules.

(a) Required use prohibited. No program may require or request any patient to carry in his or her possession, while away from the program premises, an identification card or other form of identification which is issued by the program or which would tend to identify the bearer as a participant in it or any similar program.

(b) Conditions of voluntary use. Nothing in this section prohibits a program from issuing an identification card to a patient if the patient's counsellor or other authorized member of the program staff has explained to the patient that acceptance and use of the card is entirely voluntary and that neither an initial rejection nor a subsequent discontinuation of its use will in any way prejudice his or her record or standing in the program. In the case of any patient to whom an identification card or similar device was issued prior to the effective date of this section, or subsequent thereto in violation of this section, a counsellor or other authorized member of the program staff shall explain to the patient his right to turn it in without prejudice at any time.

(c) On-premises exemption. Nothing in this section prohibits a program from maintaining and using on its premises cards, photographs, tickets, or other devices, or using passwords or other information, to assure positive identification of patients, correct recording of attendance or medication, or for other proper purposes, as long as no pressure is brought on any patient to carry any such device when away from the program premises.

§ 2.20-1 Identification cards .--- Basis and purpose.

Section 2.20 is in furtherance of one of the basic purposes of the statutes authorizing this part, namely, protection of patients from improper disclosure of their status as such. Regrettably, there appear to be areas where possession of a treatment program identification card can be prejudicial to a person under arrest or subjected to a search. In any part of the country, the accidental display or circulation of such a card by reason of its loss or theft could have adverse consequences for a variety of reasons. Since programs have other means of achieving the ends which identification cards are meant to serve, patients who do not wish to assume whatever risks may be involved in carrying such cards should not be compelled to do so.

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

RULES AND REGULATIONS

gram records.—Rules.

(a) General rule. When a program discontinues operations or is taken over or acquired by another program, its records to which this part applies with respect to any patient may, with the written consent of that patient, be turned over to the acquiring program or, if none, to any other program specified in the patient's consent. Except as otherwise provided in this section, any records to which this part applies, but for the transfer of which patient consent is not obtained, shall be either completely purged of patient identifying information, or destroyed. If any effort to obtain consent for transfer is made, it shall be by means which minimize the likelihood of accidental or incidental disclosure to any third party of the patient's identity as such.

(b) Retention period. Where records are required by law to be kept for a specified period, and such period does not expire until after the discontinuation or acquisition of the program, and patient consent for their transfer is not obtained. such records shall be sealed in envelopes or other containers marked or labelled as follows: "Records of [insert name of program] required to be maintained pursuant to [insert citation to law or regulation requiring that records be kept] until a date not later than December 31, [insert appropriate year]." The same procedure may be followed when it is determined to retain records for the period of any applicable statute of limitations. (c) Custodial retention. Records marked and sealed in accordance with paragraph (b) of this section may be held by any lawful custodian, but may be disclosed by such custodian only under such circumstances and to such extent as would be permissible for the program in which they originated. As soon as practicable after the date specified on the label or legend required to be affixed pursuant to paragraph (b) of this section, the custodian shall destroy the records. In the case of any program terminated by reason of bankruptcy, the expense of compliance with this paragraph shall be an expense of administration of the bankrupt estate.

§ 2.21-1 Disposition of discontinued program records.-Basis and purpose-

While arguments can be made for requiring the destruction of records at the conclusion of their useful clinical life, there is wide disagreement on its span, and there are in addition research considerations which argue for an even longer period of retention. Except in the case of discontinued programs, it therefore seems best to leave this issue for determination by the programs concerned.

§ 2.22 Former employces and others .----Rules.

The prohibitions of this part on dis-

who are personnel of treatment programs, researchers, auditors, evaluators, service organizations, or others having access to such records or information, and continue to apply to such individuals with respect to such records or information after the termination of their employment or other relationship or activity giving rise to such access.

§ 2.22-1 Former employees and others.-Basis and purpose.

The probition contained in § 2.22 is arguably an interpretation of the authorizing legislation which would be necessary as a matter of law even in the absence of this part; its validity as an exercise of the rulemaking power conferred by subsection (g) of the authorizing legislation seems beyond dispute.

§ 2.23 Relationship to State laws.— Rules.

The enactment of the provisions of law authorizing this part was not intended to preempt the field of law covered thereby to the exclusion of State laws not in conflict therewith. If a disclosure permitted under the provisions of this part. or under a court order issued pursuant thereto, is prohibited under State law. nothing in this part or in the provisions of law authorizing this part may be construed to authorize any violation of such State law. No State law, however. may either authorize or compel any disclosure prohibited by this part.

2.23–1 Relationship to State laws.— **Basis and purpose.**

Section 2.23 sets forth publicly an interpretation which, in informal communications, has consistently been given to 21 U.S.C. 1175 since its original enactment, and clearly has equal applicability to 42 U.S.C. 4582.

§ 2.24 Relationship to section 303(a) of Public Health Service Act and section 502(c) of Controlled Substances Act.---Rules.

(a) Research privilege description. In some instances, there may be concurrent coverage of a program or activity by the provisions of this part and by a regulation or other administrative action under section 303(a) of the Public Health Service Act (42 U.S.C. 242a(a)) or section 502(c) of the Controlled Substances Act (21 U.S.C. 872(c)). The latter two provisions of law, referred to hereinafter in this section as the research privilege sections, confer on the Secretary of Health. Education, and Welfare, and on the Attorney General, respectively, the power to authorize researchers to withhold from all persons not connected with the research the names and other identifying information concerning individuals who are the subject of such research. The Secretary of Health. Education, and Welfare may grant this privilege with respect to any "research on mental health, including research on the use and effect of alcohol and other psychoactive drugs." closure of patient records or information The Attorney General's power is concontained therein apply to all individuals ferred as part of a section authorizing

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

- 15 - **15** -

research related to enforcement of laws under his jurisdiction concerning sub-stances which are or may be subject to stances which are or may be subject to control under the Controlled Substances Act, but is not expressly limited to such research. Regardless of whether a grant of research privilege is made by the Secretary or by the Attorney General, it is expressly provided that persons who obtain it "may not be compelled in any Federal, State, or local civil, criminal, administrative, legislative, or other proceeding to identify" the subjects of research for which the privilege was obtained.

(b) Comparison with authority for this part. Although they deal, in a sense, with the same subject matter, and may on oc- Act of 1972 with the provisions of the casion concurrently cover the same transactions, it is important to note the differences between the research privilege sections (21 U.S.C. 872(c) and 42 U.S.C. 242a(a)) and the provisions of law (21 U.S.C. 1175 and 42 U.S.C. 4582) which authorize this part. Briefly, these differences are as follows:

(1) Although they contain broad grants of express rulemaking authority, the provisions of law by which this part is authorized are self-executing in the sense that they are operative irrespective of whether the rulemaking authority is exercised. The protection afforded by the research privilege sections, on the other hand, can only come into existence as a result of affirmative administrative action

(2) The provisions of law authorizing this part, as well as the provisions of this part itself, impose affirmative duties with respect to the records to which they apply, and the violation of such duties is subject to criminal penalties. To the extent that a privilege is thereby created, it grows out of the duties thus imposed. The research privilege sections, by contrast, impose no duties by their own terms, and if any duties are implied from their existence, they would have to be enforced on the basis of an implicit civil liability for damages or by equitable relief, as there are no criminal or administrative sanctions available.

(3) The exercise of the authority conferred by the research privilege sections is subject to administrative discretion, whereas in the case of the duties imposed under this part there is judicial discre- (§§ 1401.61 and 1401.62 of the previous tion, within the limits and subject to procedures and criteria prescribed by statute and regulation, to grant relief in particular cases.

(c) Grant of research privilege not affected by (b) (2) (C) order. The issuance L. 93-282. of an order under subsection (b) (2) (C) of either of the sections authorizing this part (21 U.S.C. 1175 and 42 U.S.C. 4582) in no way affects the continuing effectiveness of any exercise of the authority of the Secretary of Health, Education, and Welfare under 303(a) of the Public Health Service Act (42 U.S.C. 242a(a)) or the Attorney General under section 502(c) of the Controlled Substances Act (21 U.S.C. 872(c)).

Act .- Basis and purpose.

(a) In Pub. L. 93-282, the Congress expressly amended (by sections 122(a) and 303(a), 88 Stat. 131 and 137) the provisions of law which authorize this part, expressly amended (by section 122 (b), 88 Stat. 132) the research privilege section under the Secretary's jurisdiction, and made explicit reference (in section 303(d), 88 Stat. 139) to the regulations previously issued by the Special Action Office for Drug Abuse Prevention reconciling the provisions of section 408 of the Drug Abuse Office and Treatment research privilege sections. When the bill which became Pub. L. 93-282 was before the House of Representatives for its last Congressional consideration before transmission to the President, its floor manager. Chairman Staggers of the Committee on Interstate and Foreign Commerce, inserted in the Record a detailed analysis of the bill in its final form (Congressional Record, daily edition, May 6, 1974, page H3563). This analysis contained the following paragraph:

The relationship of section 303(a) of the Public Health Service Act, authorizing the administrative grant of absolute confidentiality for research, to section 408 of the Drug Abuse Office and Treatment Act of 1972, requiring that Federally-connected drug abuse patient records generally be kept confidential, has been correctly described in an interpretative regulation, 21 C.F.R. 1401.61 and 1401.62, which was upheld in People v. Newman, 32 N.Y. 2d 379, [reversing] 336 N.Y.S. 2d. 127, 298 N.E. 2d 651 (1973); certiorari denied, [414] U.S. [1163], 94 S. Ct. 927, [39 L. Ed. 2d 116] (1974). For that reason, among others, section 303(d) of the Senate amendment expressly continues the effectiveness of the current regulation promulgated by the Director of the Special Action Office for Drug Abuse Prevention. Thus, although section 502(c) of the Comprehensive Drug Abuse Prevention and Control Act of 1970 is not explicitly referred to in this legislation, the congressional intent is clear that the authority conferred by that section was not modified by Pub. L. 92-255, and is not intended to be modified by the bill now before the House.

(b) Sections 2.24 and 2.61 restate, in substance, the interpretative rules regulations) referred to in the passage quoted in paragraph (a) of this section, modified to reflect the amendment made to section 303(a) of the Public Health Service Act (42 U.S.C. 242(a)) by Pub.

Subpart C---Disclosures With Patient's Consent

§ 2.31 Written consent required .----Rules.

(a) Form of consent. Except as otherwise provided, a consent for a disclosure under this part must be in writing and must contain the following:

(1) The name of the program which is to make the disclosure.

(2) The name or title of the person

27811

(3) The name of the patient.

(4) The purpose or need for the disclosure.

(5) The extent or nature of information to be disclosed.

(6) A statement that the consent is subject to revocation at any time except to the extent that action has been taken in reliance thereon, and a specification of the date, event, or condition upon which it will expire withou' express revocation.

(7) The date on which the consent is signed.

(8) The signature of the patient and, when required under § 2.15, the signature of a person authorized to give consent under that section: or, when required under § 2.16, the signature of a person authorized to sign under that section in lieu of the patient.

(b) Duration of consent. Any consent given under this subpart shall have a duration no longer than that reasonably necessary to effectuate the purpose for which it is given.

(c) Disclosure prohibited with deficient consent. No program may disclose any information on the basis of a consent form-

(1) which on its face substantially fails to conform to any of the requirements set forth in paragraph (a), of this section, or

(2) which is known or in the exercise of reasonable care should be known. to the responsible personnel of the program to be materially false in respect to any item required to be contained therein pursuant to paragraph (a) of this section

(d) Falsification prohibited. No person may knowingly make, sign, or furnish to a program any consent form which is materially false with respect to any item required to be contained therein pursuant to paragraph (a) of this section.

§ 2.31-1 Written consent required .----Basis and purpose.

(a) The use of a consent form containing all of the elements specified in § 2.31(a) is necessary to assure compliance with the requirements of this subpart. Under § 1401.21 of the previous regulations, a much more abbreviated form was permissible, because the circumstances under which any consent could be given were very strictly limited. Now that the authorizing legislation permits disclosure with consent "to such extent, under such circumstances, and for such purposes as may be allowed under regulations," the consent form should show on its face information sufficient to indicate compliance with the regulations.

(b) Sections 2.31(b), 2.31(c), and 2.31 (d) are an exercise of the general rulemaking authority in subsection (g) of the authorizing legislation, Section 2.31 (c) imposes a legal liability on programs and their personnel for disclosure of information on the basis of a materially

° - 16 -

deficient consent, and § 2.31(d) imposes liability on any person who submits a falsified consent form to a program.

§ 2.32 Prohibition on redisclosure.-Rules.

Whenever a written disclosure is made under authority of this subpart, except a disclosure to a program or other person whose records pertaining to the patient are otherwise subject to this part, must contend. the disclosure shall be accompanied by a written statement substantially as follows: "This information has been disclosed to you from records whose confidentiality is protected by Federal law. Federal regulations (42 CFR Part 2) prohibit you from making any further disclosure of it without the specific written consent of the person to whom it pertains, or as otherwise permitted by such regulations. A general authorization for the release of medical or other information is NOT sufficient for this purpose." An oral disclosure may be accompanied or followed by such a notice.

A person who receives information from patient records and has been notified substantially in accordance with paragraph (a) of this section is prohibited from making any disclosure of such information except with the specific written consent of the person to whom it pertains, or as otherwise permitted under this part.

(c) Restriction on redisclosure. Whenever information from patient records is needed by any person, such information must be obtained directly from the program maintaining such records and not from another person to whom disclosure thereof has been made, except where the initial disclosure was intentionally and expressly made for the purpose of redisclosure (as for example in disclosure) was made, the information the case of an employment agency), or the information is no longer available from the program and redisclosure is not prohibited by any other provision of this part.

§ 2.32-1 Prohibition on redisclosure. Basis and purpose.

(a) Section 2.32 is intended to provide of the previous regulations, expanded to ment and the dosage thereof, with a reasonable protection against redismake explicit reference to nonmedical relevant dates, to a permissible central closure of information disclosed with counselling and other treatment and reregistry with respect to any patientconsent in accordance with this subpart. habilitative services. (1) When the patient is accepted for (b) Section 2.33(b) clarifies the cor-There is, of course, no problem where treatment. responding provision in § 1401.22(a) of the information becomes part of a record (2) When the type or dosage of the which is itself subject to this part because the previous regulations by specifying drug is changed, and it is maintained in connection with the how and through whom oral consent can (3) When the treatment is interbe given, and limiting the disclosure to performance of a covered substance rupted, resumed, or terminated. abuse prevention function. The difficulty that necessary to determine appropriate (e) Disclosures with respect to applicaarises when the disclosure is made to medication. tions. When any person applies to a prothose whose records are not otherwise gram for maintenance treatment or de-§ 2.34 Prevention of certain multiple affected by this part. To attempt to make toxification treatment, then for the purenrollments .--- Rules. all of the provisions of this part applipose of inquiring whether such person (a) Definitions. For the purposes of cable to such recipients with respect to is currently enrolled in another program this section and § 2.55such information might raise serious for such treatment, the program may (1) The terms "administer", "con-trolled substance", "dispense", "main-tenance treatment", and "detoxification problems of legality, administrative feasifurnish patient identifying information bility, and fairness, but where they are with respect to such persongiven actual notice that specific patient (1) To any permissible central registreatment" shall respectively have the consent is normally required for redismeanings defined in paragraphs (2), (6), try of which the program is a member, closure, we think they can and should be bound by it. (10), (27), and (28) of section 102 of the and

RULES AND REGULATIONS

torily covered because they should rarely be made to any recipient with whom the program does not have a continuing relationship. Where such a relationship exists or the program is otherwise satis-(a) Notice to accompany disclosure. fied that the recipient understands and will respect the confidential nature of the information supplied there seems no need to add to the already heavy load of paperwork with which programs

> § 2.33 Diagnosis, treatment, and rehabilitation .- Rules.

(a) Disclosure authorized. Where consent is given in accordance with § 2.31, disclosure of information subject to this part may be made to medical personnel or to treatment or rehabilitation programs where such disclosure is needed in order to better enable them to furnish services to the patient to whom the information pertains.

(b) Traveling, incarcerated, or hospitalized patients on medication. Where a patent on medication is at a distance (b) Consent required for redisclosure. from his normal residence or treatment program or is incarcerated or hospitalized, or is otherwise unable to deliver a written consent to his treatment program at the time the disclosure is needed, confirmation of the patient's status and information necessary to appropriately continue or modify his medication may be given to medical personnel in a position to provide services to the patient upon the oral representation of any legal, administrative, supervisory, or such personnel that the natient has requested medication and consented to such disclosure Any program making a disclosure in accordance with this paragraph shall make a written memorandum showing the name of the patient. or the patient's case number assigned by the program, the date and time the disclosed, and the names of the individuals by whom and to whom it was made.

> § 2.33-1 Diagnosis, treatment, and rehabilitation .- Basis and purpose.

(a) Section 2.33(a) is a restatement of the policy set forth in § 1401.22(a)

(b) Oral disclosures are not manda- Controlled Substances Act (21 U.S.C. 802)

(2) The term "program" means a program which offers maintenance treatment or detoxification treatment.

(3) The term "permissible central registry" means a qualified service organization which collects or accepts. from two or more programs (referred to hereinafter as member programs) all of which are located either within a given State or not more than 125 miles from the nearest point on the border of such State, patient identifying information about persons applying for maintenance treatment or detoxification treatment for the purpose of enabling the member programs to prevent any individual from being concurrently enrolled in more than one such program.

(b) Use of central registries prohibited except as expressly authorized. The furnishing of patient identifying information by a program to any central registry which fails to meet the definition of a permissible central registry set forth in paragraph (a) (3) of this section is prohibited, and the furnishing of patient identifying information to or by any central registry except as authorized in this section is prohibited. Information pertaining to patients held by a central registry may be furnished or used in accordance with paragraphs (e), (f), and (g) for the purpose of preventing multiple enrollments, but may not be otherwise furnished or used in connection with other action with respect to any patient.

(c) Safeguards and procedures required. To minimize the likelihood of disclosures of information to impostors or others seeking to bring about unauthorized or improper disclosure, any communications carried on by programs pursuant to this section must be conducted (1) by authorized personnel designated in accordance with § 2.17(b), and (2) in conformity with procedures established in accordance with that section.

(d) Disclosures with respect to patients in treatment. A member program may supply patient identifying information and information concerning the type of drug used or to be used in treat-

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

- 17 -

not more than 200 miles distant and which is not a member of any central registry of which the inquiring program is a member.

(f) Program procedure in case of apparent concurrent enrollment. When an inquiry pursuant to paragraph (e) (2) is made of another treatment program and to serve as a conduit for otherwise proits response is affirmative, the two programs may engage in such further com- narily, the attorney-client privilege munication as may be necessary to establish whether an error has been made, and ject to waiver by the client, whereas this if none, the programs should proceed in accordance with sound clinical practice and any applicable regulations pertain- given information about any patient, this ing to the type of treatment involved.

(g) Registry procedure in case of apparent concurrent enrollment. When an inquiry pursuant to paragraph (e) (1) is made of a permissible central registry and its response is affirmative, it may advise the inquiring program of the name, address, and telephone number of the other program, or it may advise the other program of the identity of the patient and the name, address, and telephone number of the inquiring program, or it may do both, and in any case the two programs may then communicate as provided in paragraph (f) above.

(h) Advice to patients. When the policles and procedures of any program involve any disclosures pursuant to this section, before any patient is accepted for or continued in treatment (other than detoxification treatment) after September 30; 1975, written consent in accordance with § 2.31 shall be obtained. Such consent shall set forth a current list of the names and addresses either of any § 2.37 Third-party payers and funding programs or of any central registries to which such disclosures will be made. Notwithstanding the requirement of § 2.31 (a) (2), such consent shall be effective with respect to any other such program thereafter established within 200 miles. or any registry serving such programs, and shall so state. Such consent shall be effective for as long as the patient remains enrolled in the program to which it is given.

§ 2.34–1 Prevention of certain multiple enrollments .--- Basis and purpose.

Section 2.34 is based upon § 1401.43 of the previous regulations. It was omitted from the August 22, 1974 draft, but comments on the omission made it clear that in certain areas of the country, central registries are a functional component of performance of an alcohol or drug abuse tions to guide their operations are needed

§ 2.35 Legal counsel for patient.-Rules.

When a bona fide attorney-client relationship exists between an attorney-atlaw and a patient, disclosure of any information in the patient's records may be made to the attorney upon the written application of the patient endorsed by the attorney. Information so disclosed may not be further disclosed by the attorney.

§ 2.35-1 Legal counsel for patient.-Basis and purpose.

the statement of the policy embodied in should clarify the question of coverage, program is made a condition of such in-

(2) To any other program which is § 1401.25 of the previous regulations. Its and where coverage exists, provide a purpose is to assure the availability to standard which will minimize the likelithe attorney, with his client's consent, of hood of violations. See also $\frac{1}{2}.12-1(g)$. any information needed as a basis for advice and counsel. The purpose of the prohibition on further disclosure by the attorney is to guard against the possibility that the attorney might be forced hibited disclosures to third parties. Ordiwould suffice, but that privilege is subprohibition is not. Where there is a need for disclosure, to a third party of any prohibition in no way affects the availability of other sections of this part to authorize such disclosure by the program.

> § 2.36 Patient's family and others .----Rule.

Where consent is given in accordance current or past status in a treatment program may be furnished to any person with whom the patient has a personal relationship unless, in the judgment of the person responsible for the patient's treatment, the disclosure of such information would be harmful to the patient.

§ 2.36-1 Patient's family and others .--Basis and purpose.

Section 2.36 expresses the same policy as was embodied in \$ 1401.27 of the previous regulations, broadened to reflect the expanded authority for consensual disclosure under the authorizing legislation.

sources .- Rules.

(a) Acquisition of information. Disclosure of patient information to thirdparty payers or funding sources may be made only with the written consent of the patient given in accordance with \$ 2.31 and any such disclosure must be § 2.38-1 Employers and employment limited to that information which is reasonably necessary for the discharge of the legal or contractual obligations of

the third-party payer or funding source. funding source or third-party payer maintains records of the identity of recipients of treatment or rehabilitation services for alcohol or drug abuse such records are, under the authorizing legislation, maintained in connection with the the treatment system, and that regula- prevention function and are subject to gust 22, 1974 draft that disclosures to the restrictions upon disclosure set forth in this part.

> § 2.37-1 Third-party payers and funding sources .--- Basis and purpose.

Section 2.37 is based upon the general authority to prescribe regulations to carry out the purposes of the authorizing legislation. The great diversity of contractual arrangements and legal require- ployer of a genuine effort by the emments under which the operations of third-party payers and funding sources are carried on precludes the prescription of detailed records management instructions in these regulations, even if that were otherwise desirable. The general Section 2.35 simplifies and broadens principles set forth in \$ 2.37, however, ticipation by an individual in a treatment

§ 2.38 Employers and employment agencies .- Rules.

(a) Disclosure permitted. Where consent is given in accordance with § 2.31. a program may make disclosures in accordance with this section.

(b) Eligible recipients. A program may make disclosures under this section to public or private employment agencies. employment services, or employers.

(c) Scope of disclosure. Ordinarily, disclosures pursuant to this section should be limited to a verification of the patient's status in treatment or a general evaluation of progress in treatment. More specific information may be furnished where there is a bona fide need for such information to evaluate hazards which the employment may pose to the with § 2.31, information evaluating his patient or others, or where such information is otherwise directly relevant to the employment situation.

> (d) Criteria for approval. A disclosure under this section may be made if, in the judgment of the program director or his authorized representative appointed as provided in § 2.17(b), the following criteria are met:

(1) The program has reason to believe, on the basis of past experience or other credible information (which may in appropriate cases consist of a written statement by the employer), that such information will be used for the purpose of assisting in the rehabilitation of the patient and not for the purpose of identifying the individual as a patient in order to deny him employment or advancement because of his history of drug or alcohol abuse.

(2) The information sought appears to be reasonably necessary in view of the type of employment involved.

agencies .- Basis and purpose.

Section 2.38 is based on the rulemaking power conferred by subsection (b) (1) of the authorizing legislation, and is (b) Prohibition on disclosure. Where a adapted from § 1401.26 of the previous regulations. Its purpose is to allow disclosures reasonably necessary and appropriate to facilitate the employment of patients and former patients, while protecting patients against unnecessary or excessively broad disclosures. It was urged in a comment received on the Auemployers be flatly prohibited on the ground that the employer's sole legitimate concern is with on-the-job performance. While we are not unsympathetic to this view, a countervailing consideration is that in the case of an employee or applicant who is known by the employer to have a problem with drugs or alcohol, knowledge by the employee to deal with it can make the difference between a job and no job.

> § 2.39 Criminal justice system referrals.-Rules.

(a) Consent authorized. Where par-

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

27813

was unsatisfactory. Thus, if such an evalposes here involved can be made by prouation were all that could be communigrams. We have been urged to make an cated by a program about a particular exception from the requirement of § 2.31 in the case of parolees and probationers, patient's conduct during the period he was in treatment, a condition requiring but such an exception would be whony satisfactory participation in a treatment unsupported by the authorizing legislaprogram would to all intents and purtion. In fashioning these regulations, it is poses become unenforceable. Moreover, if not our privilege to adorn a tabula rasa it were held to be enforceable, the operaaccording to our own predilections: tive decision on the revocation issue rather, it is our duty to interlineate a would then be made by the program, arstatute with fidelity to its spirit, its guably exacerbating rather than alleviatterms, and its purposes. vision. ing its role-conflict problem. It may thus (b) Duration of consent. Where con-§ 2.40 Situations not otherwise provided for.--Rules. some degree of role-conflict is inherent (a) Criteria for approval. In any sitin the situation of any program which uation not otherwise specifically proaccepts criminal justice referrals. If so, vided for in this subpart, where consent the issue then becomes that of finding is given in accordance with § 2.31, a prothe most constructive way to handle the gram may make a disclosure for the conflict, rather than a sterile and futile benefit of a patient from the records of effort to avoid it altogether. that patient if, in the judgment of the (c) We are persuaded that in many program director or his authorized rep-(1) Arrested, when such person is instances a prohibition on free comresentative appointed as provided in munication between probation officers § 2.17, all of the following criteria are and drug abuse program counsellors met:

dividual's release from confinement, the treatment program to the effect that a or not referred by the criminal justice disposition or status of any criminal pro- patient's status or progress in treatment system, before disclosures for the purceedings against him or the execution or suspension of any sentence imposed upon him, such individual may consent to unrestricted communication between any program in which he is enrolled in fulfillment of such condition and (1) the court granting probation, or other posttrial or pretrial conditional release, (2) the parole board or other authority granting parole, or (3) probation or parole officers responsible for his supersent is given for disclosures described in be the part of wisdom to confess that paragraph (a) of this section, such consent shall expire sixty days after it is given or when there is a substantial change in such person's status, whichever is later. For the purposes of this section, a substantial change occurs in the status of a person who, at the time such consent is given, has beenformally charged or unconditionally released from arrest:

27814

commenced:

(2) Formally charged, when the would have profoundly deleterious effects (1) There is no suggestion in the charges have been dismissed with prejuon the rehabilitative process. Many prodice, or the trial of such person has been written consent or the circumstances bation officers bring to their work a high surrounding it, as known to the program, degree of training, professionalism, and (3) Brought to a trial which has comthat the consent was not given freely, experience. They are under no illusion voluntarily, and without coercion. menced, when such person has been that they are dealing with a clientelle (2) Granting the request for disacquitted or sentenced. which will never stumble or relapse, and (4) Sentenced, when the sentence has if they have the information necessary closure will not cause substantial harm been fully executed. to intervene at an early stage of such to the relationship between the patient (c) Revocation of consent. An indiand the program or to the program's an episode, their intervention can often vidual whose release from confinement, make the difference between success and capacity to provide services in general. probation, or parole is conditioned upon failure for the client.

his participation in a treatment program bation, or parole.

(d) There is, however, nothing in these closure will not be harmful to the may not revoke a consent given by him regulations which precludes treatment patient. in accordance with paragraph (a) of this programs from entering into agreements (b) Circumstances deemed beneficial. section until there has been a formal or arrangements with agencies or insti-For the purposes of this section, the and effective termination or revocation tutions of the criminal justice system to circumstances under which disclosure of such release from confinement, proregulate or restrict the subject matter or may be deemed to be beneficial to a form of communications of information patient include, but are not limited to. (d) Restrictions on redisclosure. Any about patients. For example, such an those in which the disclosure may assist information directly or indirectly rearrangement might provide for free oral the patient in connection with any pubceived pursuant to this section may be communication between counsellors and lic or private claim, right, privilege, used by the recipients thereof only in probation officers, while restricting forgratuity, grant or other interest accruing connection with their official duties with mal written reports by the program to to, or for the benefit of, the patient or the respect to the particular individual with specified types of so-called hard data patient's immediate family. Examples of respect to whom it was acquired. Such such as attendance and urinalysis results. the foregoing include welfare, medicare, recipients may not make such informa-In view of widely differing conditions and unemployment, workmen's compensation available for general investigative attitudes in various parts of the country, tion, accident or medical insurance, pubpurposes, or otherwise use it in unrelated substantial variations in such arrangelic or private pension or other retirement proceedings or make it available for ments are not only expectable but debenefits, and any claim or defense asunrelated purposes. sirable. serted or which is an issue in any civil. § 2.39-1 Criminal justice system refer-(e) A further aspect of this matter, criminal, administrative or other prorals .--- Basis and purpose. which was not adequately considered or ceeding in which the patient is a party dealt with in the May 9 proposal, is the or is affected

(a) On the basis of extensive written impact which the rules laid down in § 2.40-1 Situations not otherwise procomment and oral communications re-§ 2.39 have on the bail decision. There is ceived on the subject matter of § 2.39 vided for.-Basis and purpose. a high correlation between the disposias proposed in the May 9, 1975 notice (a) Section 2.40 is based upon §1401.23 tion of the application for bail and the (designated as § 2.40 in that notice), we of the previous regulations, amended to type of sentence which may be meted have concluded that the latitude allowed reflect the expansion made by the change out upon conviction. The contrast beand the conditions imposed in § 2.39 as tween the recidivism rates for those who in the law with respect to the permissible set forth above are necessary and proper scope of consensual disclosures. receive treatment and supervision, as to effectuate the purposes of the authoragainst those who simply receive the (b) A strong case can be made for the izing legislation. proposition that § 2.40 should, in (b) From a legal standpoint, it seems punishment of incarceration, is a powereffect if not expressly, require a program highly doubtful whether, in a proceeding ful argument against restrictions which to make any disclosure requested by a to revoke probation or parole, the due would tend to narrow the circumstances patient. The discretion vested in the proprocess requirements laid down in Morunder which conscientious judges can gram, it can be argued, is at best an rissey v. Brewer, 408 U.S. 471, 92 S.Ct. grant bail. expression of overprotective paternalism, 2593, 33 L.Ed.2d 484 (1972) and Gagnon v. Scarpelli, 411 U.S. 778, 93 S.Ct. 1756, (f) It must be emphasized that § 2.39 and at worst, an invitation to programs 36 L.Ed.2d 636 (1973) could be met by in no way reduces the necessity to obtain to cover up material potentially eman unsupported general evaluation by a written consent from patients, whether barrassing to themselves. Bearing in

÷

RULES AND REGULATIONS

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

(3) Granting the request for dis-

27815

mind, however, that persons who have provision in the previous regulations closure or redisclosure of patient identiobtained the type of treatment to which with respect to patients who may be inthis part applies are more vulnerable to carcerated is now covered in § 2.33(b). pressures of various kinds than are paretain some responsibility on the part lowed is of the patient's endangered of the program to protect the best in- condition, not his identity as a drug or terests of its patients in this very sensi- alcohol abuse patient, and that the hutive area. This, like many other choices manitarian necessity of such notificawhich these regulations reflect, is a de- tion outweights its potential for accitermination which can be reviewed and dental violation of confidentiality revised from time to time in the light of experience.

Subpart D-Disclosures Without Patient Consent

§ 2.51 Medical emergencies.-Rules.

(a) In general. Disclosure to medical personnel, either private or governmental, is authorized without the consent of the patient when and to the extent necessary to meet a bona fide medical emergency.

(b) Food and Drug Administration. Where treatment involves the use of any drug, and appropriate officials of the Food and Drug Administration determine that the life or health of patients may be endangered by an error in the manufacture or packaging of such drug, disclosure of the identities of the recipients of the drug may be made without their consent to appropriate officials of the Food and Drug Administration to enable them to notify the patients or their physicians of the problem in order that corrective action may be taken.

(c) Incapacitated persons. Where a patient is incapacitated and information concerning the treatment being given him by a program is necessary to make a sound determination of appropriate emergency treatment, such information may be given without the patient's consent to personnel providing such emergency treatment.

(d) Notification of family or others. When any individual suffering from a serious medical condition resulting from drug or alcohol abuse is receiving treatment at a facility which is within the scope of this Part the treating physician may, in his discretion, give notification of such condition to a member of the individual's family or any other person with whom the individual is known to have a responsible personal relationship. Such notification may not be made without such individual's consent at any time such individual is capable of rational communication.

(e) Record required. Any program making an oral disclosure under authority of this section shall make a written memorandum showing the patient's name or case number, the date and time the disclosure was made, some indication of the nature of the emergency, the information disclosed, and the names of the individuals by whom and to whom it was disclosed.

§ 2.51-1 Medical emergencies.—Basis and purpose.

The provisions of § 2.51 are adapted from § 1401.42 of the previous regulations, and are based on subsection (b) (2) (A) of the authorizing legislation. The paragraph (a) and any subsequent dis- have emerged. One is that retrospective

Paragraph (d) of § 2.51 is based upon tients in general, it seems preferable to the theory that the disclosure there al-

> § 2.52 Research, audit, and evaluation .--- Rules.

(a) Research, audit, and evaluation. Subject to any applicable specific pro- or broad issues of public policy, while at vision set forth hereinafter in this sub- the same time safeguarding the personal part, the content of records pertaining to privacy of the individuals who are the any patient which are maintained in intended beneficiaries of the process or connection with the performance of a program under investigation. This subfunction subject to this part may be dis- part in particular, and this part as a closed, whether or not the patient gives whole, are intended to aid in carrying consent, to qualified personnel for the out that purpose. purpose of conducting scientific research, management audits, financial audits, or program evaluation, but such personnel may not identify, directly or indirectly, any individual patient in any report of such research, audit, or evaluation, or otherwise disclose patient identities in any manner. For the purposes of this subpart and for the purposes of subsection (b) (2) (B) of the authorizing legislation, the term "qualified personnel" means persons whose training and experience are appropriate to the nature research subjects. and level of the work in which they are engaged and who, when working as part of an organization, are performing such work with adequate administrative safe- One, a protection against improper govguards against unauthorized disclosures.

titving information. (1) Where a disclosure made to any

person pursuant to paragraph (a) of this section includes patient identifying in- Amendments. The protections afforded formation with respect to any patient, to patients by the authorizing legislation, such information may not be further dis- not to mention these regulations, go far closed, and may not be used in connection with any legal, administrative, su- required. pervisory, or other action whatsoever with respect to such patient, except as provided in paragraphs (b) (2) and (b) (3) of this section.

(2) The inclusion of patient identifying information in any written or oral communication between a person to making such disclosure does not constitute the identification of a patient in a report or otherwise in violation of paragraph (a).

(3) Where a disclosure is made pursuch person, such a risk exists and the to be considered in this process. situation cannot be dealt with solely by means of communications as described in paragraph (b) (2) of this section with- mation even for carefully guarded sciout intensifying or prolonging the risk entific research should be permitted only as compared with other means of dealing on a consensual basis, two dominant lines with it, then the initial disclosure under of argument, somewhat interrelated,

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

- 20 -

fying information for the purpose of reducing the risk to the patient involved shall be subject to the provisions of \$ 2.51.

§ 2.52-1 Research, audit, and evaluation.-Basis and purpose.

(a) General purpose. Subsection (a) of this section is adapted directly from subsection (b) (2) (B) of the authorizing legislation. The purpose of each is the same: To facilitate the search for truth, whether in the context of scientific investigation, administrative management,

(b) The succeeding sections of this subpart deal with problems which arise in connection with disclosures made for certain specific purposes which have been interpreted as falling within the general purposes embraced by § 2.52. Those sections will be best understood, however, in the light of some discussion of the underlying premises of the general rule, and its relationship to two other legal concepts: the right of privacy, and the duty to obtain informed consent from

(c) The Right of Privacy. So far as is relevant to this discussion, we may consider the right of privacy in two aspects. ernmental activity, is the right to be se-(b) Use of disclosures of patient iden- cure against unreasonable searches and seizures guaranteed by the Fourth Amendment, with some expansion from the penumbras of the Fifth and Sixth beyond those which are constitutionally

(d) The other aspect of the right of privacy, which has sometimes been described as the right to be left alone, is the notion that an individual has a right not to be hurt by intrusions into his essentially personal concerns, or to have essentially private information exploited whom a disclosure has been made pur- for commercial gain, whether or not the suan': to paragraph (a) and the program intrusion or exploitation is in connection with any possible governmental action against him. The courts have spoken of a right of privacy in a wide variety of contexts, but they have repeatedly and explicitly rejected the notion that anyone suant to paragraph (a) of this section has a right to go about his daily affairs to a person qualified to determine, on the encapsulated in an impenetrable bubble basis of such disclosure, the presence of a of anonymity. The courts have been caresubstantial risk to the health and well ful to weigh the competing interests, and being, whether physical or psychological, the social interest in valid research and of any patient, and, in the judgment of evaluation is clearly of sufficient moment

> (e) In defense of the position that disclosure of patient identifying infor-

27816

studies are of questionable value in any subject is that some disclosure or misuse the creation of four categories of activicase, and the other is that a sampling of information from which he could be ties. Three of them are specifically dealt technique involving informed consent on identified might result in embarrassment with in the succeeding sections of this subpart and need not detain us here: the. the part of the members of the sample lost opportunities, or other forms of can always be used to develop the inpsychological or social injury. While fourth is discussed below. formation sought. Neither line of argu-(n) Scientific research and evaluation. that possibility of harm could be rement will withstand careful scrutiny. duced by requiring consent to every re-Beyond the bare restatement of the au-(f) It is true, of course, that the thorizing legislation set forth in \$ 2.52. view of clinical records for research purthese regulations are deliberately silent efficacy of a given therapeutic agent can poses, a similar result can be achieved by with respect to purely voluntary scientific often best be evaluated by means of a the less restrictive method of limiting well-designed prospective study in which further disclosure of identifying inforresearch and program evaluation in the special recordkeeping procedures, special mation by the researcher. Given the apsense defined in § 2.11(g) (2). Testimony criteria for patient selection, and an and written comments received or the plicability of this alternative, equally appropriate control have all been estab-August 22, 1974 draft regulations were effective means for protecting a patient lished with a view to the purpose of the noteworthy in two respects. First,' no or subject from the possibility of a study. There are, however, many imporinstances of abuse on the part of persons harmful public disclosure, it is unreasontant investigations which simply do not acquiring patient identifying informaable to insist upon informed consent to lend themselves to such a format. Sometion under these circumstances, were every review of clinical records for the times the desirability or even the poscited. Second, while there was some wellpurposes of conducting legitimate resibility of a particular study does not search, particularly since such insistence founded criticism of the attempt in that suggest itself except in retrospect. draft to provide guidelines for determincould lead to the ultimate absurdity of Another important consideration is the prohibiting efforts to identify the nature ing what is scientific research and who fact that knowledge that an investigais qualified to do it, no usable alternaand source of an unknown plague simply tion is going on may influence the hetives-indeed. almost no alternatives at because the patients or researcher lacked all-were forthcoming. havior of patients, clinicians, or both. the clairvoyance to have consent forms Where such knowledge can influence the (0) In one of the written comments. signed prior to the onset of the make-up of a sample, it will normally do the writer cautioned against any assumpaffliction.

so in the direction of favorable outcomes. but to an unknown degree, thus tending to invalidate the results reported.

(g) While the sample technique has its uses, especially with populations that are formation is used, and there are reunmanageably large, it is often less difficult and expensive, and less likely to interfere with the actual conduct and outcomes of treatment or rehabilitation processes, to use the full population under study. Even more important than economy and administrative convenience in carrying out a study, there may be an overriding advantage in terms of eliminating any question as to the validity of the results of the study on the ground of bias in the selection of the sample.

(h) Informed Consent. The duty to obtain informed consent is obvious and compelling in situations where an individual is exposed to the possibility of harm, either physical or psychological, as a consequence of medical procedures, research, or similar activities. Where such a situation exists the person conducting the research or medical procedure violates his duty to the subject or patient if he proceeds without obtaining the voluntary informed consent from the individual or his legally authorized representative. Thus, in conducting an activity which places the subject or patient at risk the practitioner may not give precedence to a hidden agenda, even for so lofty a motive as the advancement of knowledge. In this regard, see the Department of Health, Education and Welfare's Protection of Human Subjects Regulations, 45 CFR Part 46. Those regulations are applicable to all Department of Health, Education and Welfare grants and contracts supporting research, development and related ac-

tivities involving human subjects. (i) It is apparent that the foregoing rationale for requiring informed consent does not apply to the same degree in situations involving the disclosure of clinical records for research in the form of follow-up or retrospective studies. Under these circumstances the risk to the

RULES AND REGULATIONS

(i) In sum, there are restraints on certain means of governmental acquisition of information about individuals which are operative irrespective of how the instraints on the uses of information which are independent of how or by whom it is acquired, but they do not and should not add up to the proposition that the use of information about a person is either morally or legally the absolute prerogative of that person to determine. (k) For all of these reasons, the authorizing legislation expressly provides that patient consent is not required with respect to disclosures for research, audit. and evaluation, nor does it prohibit individual patient identification in connection with such disclosures. While it is entirely appropriate to impose safeguards and procedures in connection with these activities, it would be wholly inappropriate to use the rulemaking process to impose an absolute requireactivities which by statute may be conducted without it.

(1) Classification of activities. It is clear that Congress intended a balancing of the social interest in the validity of the results of inquiry, on the one hand, with the individual interest in anonymity, on the other, all within the limits set by the legislation and the constitution. With that objective in mind, we may now turn to the various categories of activities which come within the purview of this subpart.

(m) These activities may be classified first, in regard to whether participation is voluntary from the standpoint of the program, and second, as to whether the personnel. objective is to ascertain compliance with predetermined standards (examinations as defined in § 2.54, and program evaluaascertain the validity of a given standard or hypothesis (scientific research; and program evaluation as defined in § 2.11

tion "that our major remaining problems in drug and alcohol abuse treatment are prevention of illicit diversion and protection of confidentiality," and suggested that we still have a problem in discovering, testing, and evaluating improved treatment techniques. To do this," he continued, "one should place minimalobstacles in the way of bona fide clinical and epidemiologic research!"

(p) The result of leaving the rule as it . is in the statute, without attempting to. sharpen its outlines or define its terms. will be to leave it for interpretation on a case-by-case basis by those who must apply it in practice: the researchers who seek the information, and the programs which supply it. This does not foreclose the possibility of amending the regulations on the basis of experience if it appears either that clinicians are becoming so cautious that research and evaluation studies are being choked off, or that abuses are occurring in the use of inment of patient consent with respect to formation disclosed. But until a need for more detailed regulation in this area is demonstrated, we think its imposition would do more harm than good.

§.2.53 Governmental agencies .--- Rules. .

(a) In general. Where research, audit. or evaluation functions are performed by or on behalf of a State or Federal governmental agency, the minimum qualifications of personnel performing such functions may be determined by such agency, subject to the provisions of this part, with particular reference to the organizational requirements and limitations on the categories of records subject to review by different categories of

(b) Financial and administrative records. Where program records are reviewed by personnel who lack either the tion as defined in § 2.11(g) (1)), or to responsibility for, or appropriate training and supervision for, conducting scientific research, determining adherence to: treatment standards, or evaluating treat-(g) (2)). The application of the fore- ment as such, such review should be congoing classifications logically results in fined as far as practicable to adminis-

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

- 21 -

RULES AND REGULATIONS

circumstances should such personnel be shown caseworker or counsellor notes, or similar clinical records. Programs should organize their records so that financial and administrative matters can be reviewed without disclosing clinical information and without disclosing patient identifying information except where necessary for audit verification.

(c) Scientific research and long-term evaluation studies. No State and no agency or political subdivision of a State may require, as a condition to funding, licensing, or otherwise, that any program furnish patient identifying information for the purpose of conducting scientific research or long-term evaluation studies unless the recipient of such information is legally required to hold such information in confidence, is prohibited from taking any administrative. investigative; or other action with respect to any individual patient on the basis of such information, and is prohibited from identifying, directly or indirectly, any individual patient in any report of such research or evaluation, or otherwise disclosing patient identities in any manner.

(d) Opinion and description to be furnished program. Before any patient may make, and no examiner may require, identifying information is required to be any disclosure of patient identifying insubmitted by a program under the cir- formation in connection with an examicumstances described in paragraph (c), the program shall be furnished—

(1) An opinion by the attorney general or other chief legal officer of the State to the effect that the conditions specified in paragraph (c) are fulfilled with respect to such program or with respect to all programs in such State similarly situated, and

(2) A description of the administrative procedures and physical limitations on access or other measures to provide for the security of the data, but such description shall not be in such detail as to furnish guidance for wrongful attempts to breach such security.

(e) Exclusiveness of procedures. No State or local governmental agency may require any treatment program to furnish patient identifying information to itself or any other recipient except in conformity with this section or § 2.54. No Federal agency may require any treatment program to furnish patient identifying information to itself or any other recipient except in conformity with this section (other than paragraph (d) (1) thereof) or § 2.54.

§ 2.53-1 Governmental agencies.—Basis and purpose.

Section 2.53 is an implementation of the authority contained in subsection (g) of the authorizing legislation to provide safeguards and procedures to effectuate the purposes of such legislation. It makes clear that whenever information is required of a program, whether by law or by the terms or conditions of a contract or grant. the procedures and safeguards required under this section are applicable.

in connection with examinations.-

(a) Definitions. For the purposes of this section-

(1) The term "examination" means any examination to which this section is made applicable by paragraph (b) of this section.

(2) The term "examiner" means any individual or any public or private organization, including any Federal, State, or local governmental agency, which conducts an examination to which this section applies.

(b) Applicability. This section applies to any examination of the records of a treatment program which is carried out for the purpose of or as aid to ascertaining the accuracy or adequacy of its § 2.54-1 Patient identifying informafinancial or other records, or the efficiency or effectiveness of its financial, administrative, or medical management, or its adherence to financial, legal, medical, administrative, or other standards, regardless of whether such examination is called an audit, an evaluation, an inspection, or by any other name.

(c) Statement required for disclosure of patient identifying information in connection with examination. No program § 2.55 Supervision and regulation of nation unless the examiner furnishes to the program a written statement-

(1) that no record of patient identifying information will be made or retained by or on behalf of the examiner in connection with the examination without notice to the program in accordance with paragraph (c)(2) of this section or

(2) setting forth the specific purpose for which a record of patient identifying information is being retained by or on behalf of the examiner, the location at which such information will be kept, and the name, official title, address, and telephone number of a responsible individual to whom any inquiries by the program about the disposition of such record should be directed.

(d) Disposition of record of natient identifying information in connection with examination. After any record of patient identifying information retained in connection with an examination has served its purpose, or within the time prescribed in paragraph (e) of this section. whichever is earlier, the examiner shall destroy or return to the program all records (including any copies thereof) containing patient indentifying information which have been in its possession in connection with such examination.

(e) Maximum time allowed for disposition. The action required by paragraph (d) shall be completed-

(1) Except as provided in paragraph (e) (2) of this section not more than two years after the record was acquired by or on behalf of the examiner, or

(2) Where the record is needed in connection with a formal legal proceeding against the program commenced or to be after the record was acquired, and writ- advance notice, but subject to the pro-

trative and financial records. Under no § 2.54 Patient identifying information ten notice to this effect is furnished to the program within two years after the record was acquired, not later than the termination of such proceeding.

27817

(f) Notice of final disposition. When an examiner disposes of records as required by paragraph (d) of this section, or not later than the time prescribed by paragraph (e) of this section, whichever is earlier, the examiner shall furnish to the program concerned a written statement-

(1) That there has been compliance with this section and with the provisions of this part prohibiting any disclosure of patient identifying information from records held by auditors or evaluators, or (2) Specifying the particulars in which there has been a failure of compliance.

tion in connection with examinations.-Basis and purpose.

Confidence on the part of treatment program personnel in the integrity of auditing and regulatory processes is important to the effective functioning of the treatment system. It is the purpose of § 2.54 to foster practices which will both justify and engender such confidence.

narcotic maintenance and detoxification programs.—Rules.

(a) Definition of "registrant". For the purposes of this section, the term 'registrant" means a person who (1) has pending an application for registration under section 303(g) of the Controlled Substances Act (21 U.S.C. 823 (g)), or (2) has been registered under such section and whose registration has not expired or been surrendered or revoked

(b) Drug Enforcement Administration. Duly authorized agents of the Drug Enforcement Administration shall have access to the premises of registrants for the purpose of ascertaining compliance (or ability to comply) with standards established by the Attorney General under section 303(g)(2) of the Controlled Substances Act (21 U.S.C. 823(g)(2))' respecting the security of stocks of narcotic drugs and the maintenance of records (in accordance with section 307 of the Controlled Substances Act, 21 U.S.C. 827) on such drugs. Registrants shall maintain such records separate from and in addition to patients' clinical records required to be maintained under 21 CFR 310.505 (d) (7) (iii), which shall not be available to such agents except as authorized under a court order in accordance with Subpart E of this part. Records maintained by registrants for the purposes of section 307 of the Controlled Substances Act (21 U.S.C. 827) need not identify patients by name, address, social security number, or otherwise except by an identifying number assigned by the registrant, but where such a system is used, the registrant shall maintain on a current basis a cross-index referencing each identifying number to the name and address of the patient to whom it refers. commenced not more than two years Upon request at any time and without

27818

visions of § 2.54, such agents shall be which is legally and administratively guards and procedures to assure that the granted immediate access to any such separate from any agency of State govstatutory prohibition is respected. index. Such agents may use names and ernment responsible for investigation of (c) In testimony and written comaddresses so obtained strictly for the purviolations of, or enforcement of, criminal ment on the August 22, 1974 draft of poses of auditing or verifying program law generally or criminal laws relating these regulations, it has been urged that records, and shall exercise all reasonable to controlled substances; (iii) whose access to patient identifying information precautions to avoid inadvertent disclospersonnel are qualified by training or by law enforcement personnel, even for ure of patient identities to third parties. experience to conduct inspections of the limited purposes allowed by statute Names and other identifying information health care facilities to ascertain comand regulation, should be prohibited exso obtained may not be compiled or used pliance with treatment standards: and cept pursuant to a court order obtained in any registry or personal data bank of (iv) whose personnel are by State law. under 21 U.S.C. 1175(b) (2) (C). We any description. or by published administrative directive believe that such a prohibition is (c) Food and Drug Administration. enforced by effective sanctions, required beyond our power to impose. to maintain the confidentiality of any (d) Section 307(b) of the Controlled information concerning the identity of Substances Act (21 U.S.C. 827) provides, in pertinent part, "Every * * * record

Duly authorized agents of the Focd and Drug Administration shall have access to the premises of registrants and to all patients which they may acquire in the records maintained by registrants, for the required under this section * * * shall course of their official duties. purpose of ascertaining compliance (or (2) Access. Duly authorized agents of be kept and be available, for at least two ability to comply) with standards es- a qualified State health agency shall years; for inspection and copying by tablished by the Secretary of Health, have access to the premises of registrants officers or employees of the United States. Education and Welfare under section 4 and to all records maintained by regisauthorized by the Attorney General." It of the Comprehensive Drug Abuse Pretrants, for the purpose of ascertaining is a well known principle of statutory vention and Control Act of 1970 (42 compliance (or ability to comply) with construction that amendments and re-U.S.C. 257a), sections 303(g)(1) and 303 treatment standards (including those peals by implication are not favored. In (g) (3) of the Controlled Substances Act relating to quantities of narcotic drugs People v. Newman, 32 N.Y.2d 379, 345 (21 U.S.C. 823(g)(1) and 823(g)(3)), which may be provided for unsupervised N.Y.S.2d 502, 298 N.E.2d 651 (1973). and sections 505 and 701(a) of the Fed- use by individuals in treatment) estabcert. denied 414 U.S. 1163, 94 S.Ct. 927. eral Food, Drug, and Cosmetic Act (21 lished under State law. Such access, and 39L. Ed. 2d 116 (1974), the United States U.S.C. 355 and 371(a)). When necessary the use of any information thereby opfiled amicus briefs with the Court of Apin the conduct of their duties, and subtained, shall be subject to the restricpeals of New York and with the United ject to the provisions of § 2.54, agents tions and limitations set forth in para-States Supreme Court, arguing that secmay use names and addresses of patients graph (c) of this section, and subject tion 408 of Pub. L. 92-255 (21 U.S.C. strictly for the purposes of auditing or to § 2.54. 1175) did not effect an implied amendverifying program records, and shall exment or repeal of the provisions of Pub. § 2.55-1 Supervision and regulation of ercise all reasonable precautions to avoid L. 91-513 (21 U.S.C. 872(c) and 42 U.S.C. narcotic maintenance and detoxificainadvertent disclosure of patient identi-242a(a)) which confer on the Attorney tion programs.-Basis and purpose. ties to third parties. Names and other General and the Secretary of Health. identifying information on patients ob-(a) Section 2.55 is addressed to the Education, and Welfare the power to tained pursuant to this section or by any general problem described in the followgrant the so-called research privilege other compulsory process may not be ing passage from the legislative history discussed in § 2.24. This position was compiled or used in any registry or per- of Pub. L. 93-282: expressly adopted by the New York sonal data bank of any description. Excourt. We cannot now take the incon-A major element of the task of fashioning cept as authorized under this paragraph new regulations pursuant to the express sistent position that section 408 of Pub. or by a court order granted under Subrulemaking authority conferred by this leg-L. 92-255 did indeed amend by implicapart E of this part, (1) such agents may islation will be to reconcile the sometimes tion section 307 of Pub. L. 91-513, parnot, either orally or in writing, except conflicting interests of research, audit, and ticularly in the face of a contrary conevaluation with rights of privacy and the in conversation with personnel of the temporaneous administrative interpretaconfidentiality of the relationship between registrant while on the premises of the tion by both the Special Action Office patient and clinician, Such a reconciliation registrant, identify any patient otherwise for Drug Abuse Prevention and the Debecomes particularly crucial where the functhan by reference to an identifying numpartment of Justice. In short, if the right tions of research, audit, or evaluation are ber assigned by the registrant, and (2) of access and copying conferred on Fedconducted by, a governmental agency with such agents may not remove from the regulatory powers and responsibility, and eral agents by 21 U.S.C. 827 is to be premises of the registrant any notes, the treatment involves the use of a drug amended to provide that it may only documents, or copies thereof which consuch as methadone which is in a research be exercised pursuant to a court order tain patient identifying information. status or which is readily susceptible of misin the case of maintenance and de-(d) State drug law enforcement agenuse or illicit diversion. toxification programs, that is a change Because of the difficulty and complexity which must be wrought by the Congress.

cies. Duly authorized agents of any State of the task, the rulemaking authority is in-tentionally cast in terms broad enough to drug law enforcement agency having jurisdiction and specific responsibility permit the limitation of the scope, content, by statute or otherwise for the enforceor circumstances of any disclosure under ment of criminal laws relating to consubsection (b), whether (b)(1) or (b)(2), in the light of the necessary purposes for which it is made or required. (Congressional trolled substances (as defined in the Controlled Substances Act) shall have access to the premises of any registrant Record, daily edition, May 6, 1974, page H3563) for the purposes (with respect to corresponding provisions, if any, of State (b) It has been the consistent interlaw) and subject to the restrictions and limitations set forth in paragraph (b)

pretation of the Special Action Office for Drug Abuse Prevention that the only of this section, and subject to § 2.54. provision of the authorizing legislation (e) State health authorities. which permits disclosures to compliance (1) Definition of "qualified State officers, whether of DEA, FDA, or state health agency". As used in this paraagencies, is subsection (b) (2) (B). That graph, the term "qualified State health subsection strictly prohibits any further agency" means an agency of State govdisclosure of names or other identifying ernment (i) which has express legal information concerning patients, and the responsibility to ascertain that regis- statutory prohibition has been buttrants under its jurisdiction comply with tressed by provisions of these regula- misfilings" may be something else appropriate treatment standards; (ii) tions, notably §2.54, providing safe- again.

RULES AND REGULATIONS

(e) In the case of inspections carried out by health supervisory agencies, we think that denial of access to any documents showing patient identifying information may have a serious adverse effect on the validity of the inspection process. Even if a program keeps its own records in terms of patient-identifying numbers assigned by the program, the patient file may contain-may, indeed, be required to contain-documents signed by the patient or originating outside the program. Where signatures, names, and addresses are all obliterated. it is impossible for the inspector to check, the file even for apparent internal con-. sistency. We believe that outright forgery is and will remain a rarity, but the temptation to cover improper or inade-

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

- 23 -

as an affirmative grant of jurisdiction to under section 408 of Pub. L. 92-255 pub- order under this subpart may authorize

HEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

<page-header><page-header><page-header><text><text><text>

the submission of testimony or other held or taken in the judge's chambers, afforded an opportunity to appear and evidence by the program or its per- unless the patient requests an open hearbe heard thereon. ing or the court determines that such (c) Criteria. A court may authorize sonnel. hearing is consistent with the public indisclosure of records pertaining to, a § 2.63-1 Limitation to objective data.--patient for the purpose of conducting terest and the proper administration of **Basis and purpose.** an investigation of or a prosecution for justice. In the three-year period subsequent a crime of which the patient is suspected (d) Good cause. No order shall be issued unless the record shows that good only if the court finds that all of the cause exists, and in assessing good cause, following criteria are met:

to the original enactment of 21 U.S.C. 1175, not a single occasion was reported to the Special Action Office for Drug Abuse Prevention on which an attempt and the need for disclosure against the was made to secure a (b) (2) (C) order authorizing the disclosure of a confidential communication by a patient to a counsellor or other member of the staff of a treatment program. In all of the comments and testimony received on the draft regulations published August 22, 1974, there was nothing to suggest any circumstances under which a court order authorizing such a disclosure would be effect upon successful treatment or re- material information or evidence of subeither desirable or appropriate. Yet the mere possibility that such an order might be issued is to some a source of anxiety which impairs the effectiveness of treatment. Such an ongoing negative effect clearly outweighs the remote theoretical possibility that some peculiar circumstance might arise in which judicial authorization for such a disclosure might thorizing disclosure shallbe sought. Accordingly, the limitation imposed by § 2.63 on the scope of (b) (2) (C) orders to preclude that possibility, and hence to eliminate its adverse influence on treatment services, appears power.

§ 2.64 Procedures and criteria in gencral-Rules.

(a) Identity of patient. Applications for court orders to authorize disclosure of records pertaining to a known patient shall not use the real name of the patient unless the petient consents thereto voluntarily and intelligently. In the case an order authorizing disclosure of recof an ex parte application initiated by ords to which this part applies may be the patient, the application should be made by any person who has a legally instituted in the name of a fictious person, such as Jon Doe, unless the patient requests otherwise. The same procedure should be followed in the case of a separate proceeding held in conjunction with a pending criminal or civil action. Any court order should identify the patient fictitiously, and the disclosure of the sets out procedures and criteria for the patient's real name should be communi- issuance of (b) (2) (C) orders in general, cated to the program in such manner as subject to the more specific provisions to protect the confidentiality of the pa- with respect to particular types of protient's identity.

(b) Notice. In any proceeding not tions of this subpart. otherwise provided for in this subpart. in which the patient or the program has not been made a party, each shall be given appropriate notice and an opportunity to appear in person or to file a responsive statement, deposition or other form of response consistent with local rules of procedure. The court shall give due consideration to any such statement, deposition or other response in exercising its discretion as to the existence of be, a present or former patient in a good cause and, if deemed necessary or desirable, consistent with local rules of

(b) Notice. Except where an order under § 2.66 is sought in conjunction procedure, it may order the program diwith an order under this section, any rector to appear and give direct testiprogram with respect to whose records (c) Hearings. All hearings and all evi- an order is sought under this section dence in connection therewith shall be shall be notified of the application and identify any photograph as being a pic-

RULES AND REGULATIONS

the court shall weigh the public interest patient relationship, and to the treatment services.

(e) Need for disclosure. If other competent evidence or sources of information deny the application.

(f) Adverse effects. If there is evidence habilitation of the patient or would impair the effectiveness of the program, or other programs similarly situated, in the treatment or rehabilitation of other patients, the application should be denied unless the court finds that the adverse effects are outweighed by other factors. (g) Content of order. Any order au-

(1) Limit disclosure to those parts of the patient's record deemed essential to outweighed by the public interest in aufulfill the objective for which the order was granted:

for the order: and

(3) Include any other appropriate measures to keep disclosure to a minimum for the protection of the patient, the physician-patient relationship and the treatment services.

(h) Applications not otherwise procognizable interest in obtaining such disclosure.

§ 2.64-1 Procedures and criteria in general.-Basis and purpose.

Section 2.64. in accordance with subsection (g) of the authorizing legislation. ceedings covered in the succeeding sec-

§ 2.65 Investigation and prosecution of patients.-Rules.

to any application by an investigative, law enforcement, or prosecutorial agency for an order to permit disclosure of patient records for the purpose of conducting an investigation or prosecution of an individual who is, or who is believed to program.

(1) The crime was extremely serious. such as one involving kidnapping, homiinjury to the patient, to the physician- cide, assault with a deadly weapon, armed robbery, rape, or other acts causing or directly threatening loss of life or serious bodily injury, or was believed to have been committed on the premises of the are available, the court should ordinarily program or against personnel of the program.

(2) There is a reasonable likelihood that disclosure would have an adverse that the records in question will disclose tantial value in connection with the investigation or prosecution.

(3) There is no other practicable way of obtaining the information or evidence.

(4) The actual or potential injury to the physician-patient relationship in the program affected and in other programs similarly situated, and the actual or potential harm to the ability of such programs to attract and retain patients, is thorizing the disclosure sought.

(d) Scope. Both disclosure and dis-(2) Limit disclosure to those persons semination of any information from the to be a proper exercise of rulemaking whose need for information is the basis records in question shall be limited under the terms of the order to assure that no information will be unnecessarily disclosed and that dissemination will be no wider than necessary. Under no circumstances may an order under this section authorize a program to turn over patient records in general, pursuant to a subvided for. In any case not otherwise pro- poena or otherwise, to a grand jury or vided for in this subpart, application for a law enforcement, investigative, or prosecutorial agency.

(e) Counsel. Any application to which this section applies shall be denied unless the court makes an explicit finding to the effect that the program has been afforded the opportunity to be represented by counsel independent of counsel for the applicant, and in the case of any program operated by any department or agency of Federal, State, or local Government, is in fact so represented.

§ 2.65-1 Investigation and prosecution of patients—Basis and purpose.

(a) The need for objective criteria for the issuance of court orders in connection with investigation or prosecution of patients seems particularly pressing. In the absence of such criteria, the assur-(a) Applicability. This section applies ance of confidentiality otherwise provided for by the authorizing legislation may be felt to be of little value.

(b) It has not been found possible to frame entirely satisfactory rules for the scope of orders under § 2.65, but an illus-tration may be helpful. Where a witness to a crime is believed capable of identifying a suspect by appearance, and the criteria set forth in § 2.65(c) are met, and the program has photographs of its patients, the witness alone may be permitted to view the photographs, with no names attached. If the witness failed to

FEDERAL REGISTER, VOL. 40, NO. 127-TUESDAY, JULY 1, 1975

- 25 -

ture of the suspect, that would end the records in question shall be limited under in any program be authorized for more that the terms of the order to assure that in any program be authorized of 12 con-tion, the program would be authorized patient identities will be protected to the secutive months. to give any information in its possession as to the suspect's identity and whereabouts to appropriate authorities.

(c) It is not the purpose of this section to substitute a mechanical formula for judicial discretion, but rather to provide criteria which define the area within which discretion is to be exercised. The reason for including all crimes committed on program premises or against program personnel is not any special solicitude for programs as opposed to other victims of crime, but is rather the result of the special difficulties which the broad definition of "records" in § 2.11(0) creates for program personnel as complaining witnesses.

(d). In regard to § 2.65(e), experience has demonstrated that independent counsel may be of crucial importance. The leiding case construing 21 U.S.C. 1175, People v. Newman, 32 N.Y.2d 379, 345 N/Y.S.2d 502, 298 N.E.2d 651 (1973); certicrari denied, 414 U.S. 1163, 94 S.Ct. 927, 39 L. Ed.2d 116 (1974), would never have been presented to the courts but for the fact that legal counsel for Dr. Newman was furnished on a pro bono publico basis by a private law firm. In an entirely different case, a United States District Court appears to have issued a wholly inappropriate order under 21 U.S.C. 1175 in a case in which the treatment program involved was operated by an agency of the United States Government, and either was unrepresented, or was represented by the same attorney who represented the agency seeking the order. It is possible, of course, that the order would have been issued in any event, but it seems clear that there was no adequate presentation to the court of arguments or testimony in opposition. It is difficult to see how the purposes of subsection (b) (2) (C) of the authorizing legislation can be carried out if there is inadequate presentation of the issues to the courts which must decide them.

§ 2.66 Investigation and prosecution of programs.-Rules.

(a) Applicability. This section applies to any application by an administrative, regulatory, supervisory, investigative, law enforcement, or prosecutorial agency for principal, agent, or employee thereof is an order to permit disclosure of patient engaged in serious criminal misconduct. records or the making of copies thereof and that other means of securing evi-(including patient identifying information) for the purpose of conducting an not available or would not be effective. investigation or an administrative or judicial proceeding with respect to any

section may, in the discretion of the court, be granted without notice, but upon the implementation of any order so granted, the program shall be afforded the expiration of such 6ff-day period, an opportunity to seek the revocation or the applicant may apply for an order examendment of such order.

(c) Scope. Both disclosure and dis- period not to exceed 60 days, but in no semination of any information from the event may the use of an undercover agent

maximum practicable extent, and that names and other identifying characteristics of patients are expunged from any documents placed in any public record. No information obtained pursuant to an order under this section may be used to an application for an order under § 2.65.

§ 2.66-1 Investigation and prosecution of programs—Basis and purpose.

The principal purpose of § 2.66 is to enable a regulatory agency whose inspection or other source of information has disclosed a need for follow-up, or which has been refused access to patient records, to obtain the necessary authorization for access and copying. There may also be rare instances, such as those involving financial fraud, tax evasion, or other offenses where access by other investigative agencies is necessary, subject to the requirements and protections of this part.

§ 2.67 Undercover agents and informants-Rules.

(a) Applicability. This section applies to any application by an administrative, regulatory, supervisory, investigative, or law enforcement agency for an order to permit such agency to have an undercover agent or informant in a program under circumstances which would otherwise be prohibited under § 2.19.

(b) Notice. An order under this section may be granted without notice where the criminal conduct for the investigation of which it is granted is believed to be carried on by the program director or by any employee or agent of the program with the knowledge of the program director or under such circumstances that in the exercise of reasonable care the program director should know of such conduct. Under any 42 of the Code of Federal Regulations. other circumstances, an order under this section may be granted only after the mit written comments, views, or arguprogram director has been afforded notice and opportunity for hearing.

(c) Criteria. An order under this section may be granted only where there is reason to believe that a program or any dence of such criminal misconduct are

(d) Scope. An order granted pursuant to this section may authorize the use by program or any principal, agent, or em- the applicant of an undercover agent ployee thereof in his capacity as such. or informant, either as a patient or as (b) Notice. An application under this an employee, of the program in question.

(e) Time periods. An order under this section may not authorize the use of an undercover agent for an initial period exceeding 60 days. At any lime prior to tending such period for an additional

27821

(f) Duty of agent. Except to the extent expressly authorized in an order under this section, which shall be limited

to disclosure of information directly related to the purpose for which the order is granted, an undercover agent or inconduct any investigation or prosecution formant shall for the purposes of this of a patient, or be used as the basis for part be deemed an agent of the program within which he is acting as such, and as such shall be subject to all of the prohibitions of this part applicable to disclosures of any information which he may acquire.

> § 2.67-1 Undercover agents and informants-Basis and purpose.

> The legal rationale underlying this section has been set forth in § 2.19-1. It is expected that this section will find its principal and perhaps its exclusive application in the area of drug law enforcement. Experience has demonstrated that medical personnel, no matter how credentialed, can engage in the illicit sale of drugs on a large scale, and that the use of undecover agents and informants is normally the only effective means of securing evidence sufficient to support a successful prosecution.

[FR Doc.75-17169 Filed 6-27-75;9:38 am]

Title 21-Food and Drugs

CHAPTER III-SPECIAL ACTION OFFICE FOR DRUG ABUSE PREVENTION

PART 1401-CONFIDENTIALITY OF DRUG ABUSE PATIENT RECORDS

Revocation of Part

On May 9, 1975, there was published in the FEDERAL REGISTER (40 FR 20542) a notice of proposed rulemaking proposing the revocation of Part 1401 of Title 21 of the Code of Federal Regulations by reason of the proposed incorporation of its subject matter in a new Part 2 of Title

Interested persons were invited to subments with respect to the proposed revocation, within 30 days of the date of publication of that notice. None were received, except to the extent that they were implicit in those submitted on the proposed new Part 2 of Title 42 of the Code of Federal Regulations, which were duly considered.

Accordingly; pursuant to the authority of section 408 of the Drug Abuse Office and Treatment Act of 1972, as amended by Fub. L. 93-282 (21 U.S.C. 1175), and under the authority delegated to the General Counsel (39 FR 17901, May 21, 1974), Part 1401 of Title 21 of the Code of Federal Regulations is revoked, effective August 1, 1975.

Dated: June 25, 1975.

GRASTY CREWS, II, General Counsel, Special Action Office for Drug Abuse Prevention.

[FR Doc.75-17170 Filed 6-27-75;9:38 am]

1 2 3	NEIL F. HORTON EARL D. OSBORN JOHNSTON & KLEI 1221 Broadway, Oakland, Califo
-1	Telephone: (4)
5	MARGARET C. CRO ALAN L. SCHLOSS
6	AMITAI SCHWART2 American Civil
7	Foundation of 814 Mission St San Francisco,
8	Telephone: (4
9	Attorneys for F
10	
11	IN THE SU
12	CITY
13	
14	JOHN DOE I and
15	individually an of all others s situated,
16	
17	v.
18	T T CODDOGK
19	J. L. COPPOCK, and in his offi as Chief of Pol
20	City of San Mat Department; EAM
21	individually an
22	official capaci Sergeant of the Mateo Police De
23	CITY OF SAN MAT DEPARTMENT; CHA
24	individually and official capacit
25	Police of the Sa Police Department
26	DEAN, INdividua.
27	his official cap inspector with Francisco Police
28	the SAN FRANCISC DEPARTMENT, and through ROE XX,
-	Sincough ROE AA,

- 26 -

IN Twentieth Floor ornia 94612 15) 452-2133OSBY SER 1 Liberties Union Northern California treet , California 94103 415) 777-4880 Plaintiffs UPERIOR COURT OF THE STATE OF CALIFORNIA Y AND COUNTY OF SAN FRANCISCO G. JOHN DOE II, d on behalf imilarly Plaintiffs, No. individually CLASS ACTION cial capacity ice of the COMPLAINT FOR DECLARATORY eo Police AND INJUNCTIVE RELIEF ION RYAN, d in his ty as a City of San partment, the EO POLICE RLES R. GAIN, d in his ty as Chief of an Francisco nt; MARVIN lly and in pacity as an the San e Department; CO POLICE ROE I TAB B Defendants.

- 27 -

Plaintiffs allege:

. 1

-. 7

- 8-

9

10

11

12

2

FIRST CAUSE OF ACTION

PARTIES

1. Plaintiffs JOHN DOE I and JOHN DOE II on January 1, 1979, and on February 6-8, 1979, were enrolled as patients in the Methadone Maintenance and Levo Alpha Acetyl Methodol (L.A.A.M.) research program operated at Ward 93 of San Francisco General Hospital. For reasons that will be set out more completely below, JOHN DOE I and JOHN DOE II reasonably believe that mere disclosure of their names or identities in this complaint will directly abridge the rights and interests which they seek to vindicate by bringing this action.

13 2. Plaintiffs JOHN DOE I and JOHN DOE II bring this 14 action on behalf of themselves and on behalf of all others similarly situated. The class which plaintiffs represent is composed 15 of all persons enrolled as patients on January 1, 1979, in the 16 L.A.A.M. program operated at Ward 93 of San Francisco General Hospital whose identity was ascertained by the defendants pursuant 18 19 to a search warrant executed at Ward 93 on February 8, 1979. The 20 persons in this class are so numerous, consisting of approximately 21 35 individuals, that joinder of all such persons is impractical 22 and disposition of their claims in a class action will benefit 23 the parties and the court.

There is a well-defined community of interest in the questions of law and fact involved in this case affecting the parties to be represented by JOHN DOE I and JOHN DOE II in that all issues of law and fact are identical with respect to the entire class. As set out more completely below, the issues of

- 28 -

)	law.and fact concern
2	cuting a search warra
3	held in Ward 93 of Sa
. í	single state of fact:
5	the class to relief.
6	are typical of those
7	adequately represent
8	3. Defenda
9	the City of San Mate
0	trative officer of t
1	has primary responsi
2	the direction and co
3	San Mateo Police Dep
4	peace officers. Def
5	his official capacit
6	4. Defend
7	for the City of San
8	investigator. Defen
9	official capacity as
0	Department.
1	° 5. Defend
22	component of the Cit
3	police, a police for
14	6. Defend
5	the City and County
26	chief administrative
27	ment and has primary

6. Defendant CHARLES R. GAIN is the Chief of Police of the City and County of San Francisco. He is the duly appointed chief administrative officer of the San Francisco Police Department and has primary responsibility for the development of policies and the direction and control of subordinate employees of

상품에서 가장 가장이 있었다.

n defendants' conduct in obtaining and exerant for patient records of the entire class San Francisco General Hospital. Proof of a ts will establish the right of each member of . The claims of JOHN DOE I and JOHN DOE II e of the class and plaintiffs will fairly and t the interests of the class.

dant J. L. COPPOCK is the Chief of Police of eo. He is the duly appointed chief administhe City of San Mateo Police Department and ibility for the development of policies and ontrol of subordinate employees of the City of partment, including numerous duly appointed fendant COPPOCK is sued individually and in ty.

dant EAMON RYAN is a duly appointed Sergeant Mateo Police Department who is assigned as an ndant RYAN is sued individually and in his s a Sergeant of the City of Son Mateo Police

dant CITY OF SAN MATEO POLICE DEPARTMENT is a ty of San Mateo, consisting of a chief of rce, clerks and other employees.

- 29 -

the San Francisco Police Department, including numerous duly appointed peace officers. Defendant GAIN is sued individually and in his official capacity.

7. Defendant MARVIN DEAN is a duly appointed police officer for the San Francisco Police Department who is assigned as an inspector with the Narcotics Squad. Defendant Dean is sued individually and in his official capacity as an inspector with the San Francisco Police Department.

9 8. Defendant SAN FRANCISCO POLICE DEPARTMENT is a com10 ponent of the City and County of San Francisco, consisting of a
11 police commission, a chief of police, a police force, clerks and
12 other employees. 0

9. The true names and official capacities of defendants 13 designated as ROE I through ROE XX, inclusive, are unknown to 14 plaintiffs who therefore sue these defendants by such fictitious 15 names. Plaintiffs are informed and believe and on that basis 16 allege that each of the defendants designated herein as a ROE is 17 responsible in some manner for the practices, conduct, and acts 18 sought to be declared unlawful and restrained and prevented by 19 this action .- Plaintiffs will seek leave of the court to amend 20 their complaint to show, the true names and capacities of these. 21 defendants when they have been ascertained. 22

23

5

- 6

-8-

STATEMENT OF FACTS

10. San Francisco General Hospital, an agency of the
City and County of San Francisco under the direction of the San
Francisco Department of Health, operates in conjunction with the
University of California, a state university, a Levo Alpha
Acetyl Methodol (L.A.A.M.) research program at Ward 93 of San

- 30 -

on February 6-8, 1979. drug use.

8

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

13. One objective of the L.A.A.M. treatment program is to keep strict confidentiality of patient records in order to encourage voluntary participation in the program and in order to allow patients who successfully complete the program to lead normal lives afterward without the stigma that would attach to them if it were known that they had participated in the program. At the time of their enrollment in the program JOHN DOE I and JOHN DOE II reasonably believed that their patient records and their identity would be confidential and not released by hospital administrators or physicians except in circumstances required by federal law. Plaintiffs' participation in the program was in substantial part based upon assurances of confidentiality. 14. On February 6, 1979, an employee at San Francisco General Hospital connected with the L.A.A.M. program was contacted

Francisco General Hospital. The L.A.A.M. program was in existence on February 6-8, 1979.

11. The L.A.A.M. program is a drug abuse research program which is funded by the National Institute of Drug Abuse, an agency of the United States government, and which involves the use of L.A.A.M., an experimental form of methadone, on volunteer patients addicted to opiate drugs.

12. In connection with the performance of the L.A.A.M. program, San Francisco General Hospital maintains records of the identify, diagnosis, prognosis, and treatment of patients enrolled in the program. The records include hospital identification photos of the L.A.A.M. patients, their names, addresses, dates of birth, and medical histories, including histories of

- 31 -

by telephone by Defendant Inspector DEAN who told the employee that he (Dean) had information obtained from an informant that the perpetrators of a triple homicide two days earlier at a Payless Drugstore in the City of San Mateo might be white males enrolled in the L.A.A.M. program. Inspector Dean asked the employee for the names, dates of birth and addresses of all white male L.A.A.M. program members. The employee responded that under federal law the names of the patients could not be divulged.

15. Later in the day of February 6, 1979, defendant Inspector Dean contacted by phone an employee at Ward 93 at San Francisco General Hospital. Inspector Dean requested the names, dates of birth and addresses of Latino in addition to white male L.A.A.M. patients and told the employee that the police intended to obtain a search warrant to seize the requested information.

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

16. On February 7, 1979, the Municipal Court for the City and County of San Francisco (Judge John E. Dearman), was requested by application and affidavit signed by defendant RYAN to issue a search warrant for property at Ward 93, San Francisco General Hospital, 1001 Potrero Street, San Francisco. A copy of the affidavit is attached hereto as part of Exhibit A and is incorporated herein by reference as if set forth in full. The property included patient rosters, rolls, and records, including names, addresses, and dates of birth of patients in the L.A.A.M. program at San Francisco General Hospital.

25 17. On February 7, 1979, the requested search warrant
 26 was issued by the Municipal Court for the City and County of San
 27 Francisco (Judge John E. Dearman), and is attached hereto as part
 28 of Exhibit A and is incorporated herein by reference. The warrant

· •			2.1
	THE	PE	OPLE
	POL	ICE	MAN
	COU	NTY	OF
			coof
	bef	ore	me
	appe	eari	ing
1	tor	bel	.iev
	93,	Sar	1 Fr
	Stre	et,	Sa
	prop	pert	y o
	Pati	ent	ro
	name in t	es,	add
5. A.	in t	he	Lam
	Hosp	ita	1.
		11	
	and	tha	t sa
	sion	s o	f Se
	here	wit	h:
	•		
			*
	d. /	<u>XXX</u>	7 <u>St</u>
•			ar
			a
		•	th
			CO
		YOU	J AR
	of th	ne p	rem
	said	art	:icl

2

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27 .

28

said article same or any before me or to Section 1
18. No repre
or its L.A.A.M. progra
cisco, or of the Unive
present during the pro
requested and issued,
and no opportunity was
and be heard by the co
19. On Februa
some of which are decid

- 32 -

provides in relevant part:

E OF THE STATE OF CALIFORNIA, TO ANY OR PEACE OFFICER IN THE CITY AND SAN FRANCISCO, STATE OF CALIFORNIA:

f by affidavit having been made this day by Sgt. E. Ryan, San Mateo P.D. and it therefrom that there is probable cause ving that there is now located at Ward rancisco General Hospital, 1001 Potrero an Francisco, California, certain personal or things consisting of the following:

sters, rolls, and records, including resses, and dates of birth of patients [sic] Program at San Francisco General

aid property comes within the proviection 1524 of the Penal Code as noted

ubdivision 4 (Property or things re evidence which tends to show felony has been committed or hat a particular person has ommitted it)

THEREFORE COMMANDED to make a search nises or person described above for the es and property, and if you find the part thereof to bring it forthwith or retain it in your custody according 1536 of the California Penal Code.

esentative of San Francisco General Hospital am, or of the City and County of San Franersity of California, or of plaintiffs was occeedings at which the search warrant was plaintiffs had no notice of the proceedings,

given to the Hospital or program to appear ourt.

19. On February 8, 1979, at least six police officers, which are designated as Roes herein, entered Ward 93 of

- 33 -

San Francisco General Hospital, and presented to Dr. David Deitch, ٦, the Chief of Substance Abuse Service of San Francisco General Hospital, a copy of the search warrant, and informed Dr. Deitch 3 that if the records and information designated in the warrant were not provided they would seize every document on the ward. Dr: Deitch advised the police officers that he lacked legal 6 authority to provide the records and asked them to wait for the arrival of Mr. Frank Puglisi, the hospital administrator. When 9 Mr. Puglisi arrived, Dr. Deitch asked to call an attorney, and was told he was being placed in detention. Mr. Puglisi, with the 10 11 aid of ward personnel, provided the police officers with the 12 requested information, which included names, addresses and dates 13 of birth of all L.A.A.M. enrollees as of January 1, 1979, includ-14 ing plaintiffs, and photographs of the hospital identification 15 photographs of the L.A.A.M. enrollees, including the hospital 16 identification photographs of plaintiffs. The police officers 17 then released Dr. Deitch from detention and left the premises. 18 20. On February 16, 1979, Defendant Sergeant RYAN 19 executed a Return and Inventory on Search Warrant which was filed 20 with the Municipal Court in and for the City and County of San 21 Francisco (Judge John E. Dearman). The Return and Inventory is 22 attached hereto marked Exhibit B and is incorporated herein by 23 reference as if set forth in full. It provides in relevant part:

> I, the undersigned make this return to the within search warrant. On February 7, 1979, I received said warrant, and under its authority I diligently searched the below listed premises on (date) February 7, 1978 [sic] and there I discovered the matter described in the inventory.

> > - 34 -

24

25

26

27

28

)

5

6

8

9

10

11

12

13

14

15

16

17

18

10

20

21

22

23

24

25

26

27

28

ana gataba rants.

Premises searched: Ward 93, San Francisco General Hospital Records of patients in the LAAM Program

INVENTORY: Photocopy list of 35 individuals identified by clinic staff as being enrolled in the LAAM methadone program. Information given to this officer by clinic personnel included names, addresses, and dates of birth. of the enrollees. Photographs were taken of hospital identification photos of the LAAM patients

J, the officer by whom this search warrant was executed, do swear that the above inventory contains a true and detailed account of all the property taken by me on the warrant. The property seized will remain in the custody of the San Mateo Police Department subject to further order of this Court or other court of proper jurisdiction.

/s/ E. Ryan Sergeant E. Ryan, #18, San Mateo P.D.

21. Plaintiffs are informed and therefore allege, on information and belief, that the information and property taken from Ward 93 pursuant to the search warrant has not been returned in spite of requests by officials of the L.A.A.M. program and San Francisco General Hospital, and that any and all copies of the information and property have not been destroyed.

22. Plaintiffs are informed and therefore allege, on information and belief, that defendants intend to seek similar search warrants for the records of the L.A.A.M. program at San Francisco General Hospital, or other drug abuse programs, whenever similar circumstances exist, and that defendants will follow similar procedures as outlined herein in seeking such search war-

23. At the time the defendants sought the search warrant and seized from Ward 93 the property and information regarding

- 35 -

			5 3		Education and Welfar
	1.	L.A.A.M. patients, defendants RYAN, DEAN and several ROES had)	purposes of the Act.
: : :	2	knowledge of federal statutes and regulations protecting the		3	ments applicable to:
	3	confidentiality of patient records in federally assisted drug			· · · any
	ai a	abuse programs.		5	order to p
	i.	. 24. The Drug Abuse Office and Treatment Act of 1972		6	for the pu or prosecu
	6	provides:		7	is believe in a progr
	7	Records of the identity, diagnosis,		8	The regulations requ
	8	prognosis, or treatment of any patient which are maintained in connection with the		9	also involving a pro
	9	performance of any drug abuse prevention function conducted, regulated, or directly		10	records are sought "
1	0	or indirectly assisted by any department or agency of the United States shall			afforded an opportun:
1	1	[exception] be confidential and be disclosed only for the purposes and under the circum-		12	C.F.R. §2.65(b). In
1	2	stances expressly authorized under subsection (b) of this section. (21 U.S.C. §1175(a))		13	Court must find that:
	3	Section (b)(2)(c) provides that such records can be disclosed		14	There
ł		without the written consent of the patient:		15	records in information
1	5	If authorized by an appropriate order of		16	in connecti cution. (4
1	6	a court of competent jurisdiction granted after application showing good cause therefor. In		17	There
	7	assessing good cause the court shall weigh the public interest and the need for disclosure		18	taining inf §2.65(c)(3)
	8	against the injury to the patient, to the physician-patient relationship, and to the		19	The ac
	9	treatment services. Upon the granting of such order, the court, in determining the extent to		20	physician-p affected an
	20	which any disclosure of all or any part of the record is necessary, shall impose appropriate		21	situated, a
	21	safeguards against unauthorized disclosure.		22	to the abil and retain
-	יי <u>יי</u>	(21 U.S.C. §1175(b)(2)(c))		23	public inte sought, (4
	3	Section (c) provides:		2.1	The regulation
	-) !1	Except as authorized by a court order granted under subsection (b)(2)(c) of this		25	The regulations place disclosure:
		section, no record referred to in subsection (a) of this section may be used to initiate		26	
	25 26	or substantiate any criminal charges against a patient or to conduct any investigation of		27	Both d: information
		a patient. (21 U.S.C. §1175(c))		- 28	be limited u assure that
)		Pursuant to authority granted by the Drug Abuse Office and Treat-		•	disclosed ar
2	28	ment Act of 1972 (21 U.S.C. §1175(g)), the Secretary of Health,			
		- 36 -			

· · · ·

fare has prescribed regulations to carry out the ct. The regulations set forth specific require-

by application by an investigative, law ment, or prosecutorial agency for an permit disclosure of patient records purpose of conducting an investigation ecution of an individual who is, or who eved to be, a present or former patient ogram. (42 C.F.R. §2.65)

quire in connection with investigations not rogram itself, that the program from which the "shall be notified of the application and unity to appear and be heard thereon". 42 In addition, before authorizing disclosure, the at:

re is a reasonable likelihood that the in question will disclose material ion or evidence of substantial value ction with the investigation or prose-(42 C.F.R. §2.65(c)(2))

re is no other practicable way of obinformation or evidence. (42 C.F.R. (3))

actual or potential injury to the -patient relationship in the program and in other programs similarly and the actual or potential harm pility of such programs to attract in patients, is outweighed by the sterest in authorizing the disclosure (42 C.F.R. §2.65(c)(4))

ce limitations on the scope of any authorized

 \mathfrak{L}^{*}

disclosure and dissemination of any on from the records in question shall d under the terms of the order to at no information will be unnecessarily and that dissemination will be no

- 37 -

- 27

wider than necessary. Under no circumstances may an order under this section authorize a program to turn over patient records in general, pursuant to a subpoena or otherwise, to a grand jury or a law enforcement, investigative, or prosecutorial agency. (42 C.F.R. §2.65(c)(4)(a))

Finally, the regulations provide as to the appearance of counsel:

2

6

7

8

9

10

18

Any application to which this section applies shall be denied unless the court makes an explicit finding to the effect that the pro- . gram has been afforded the opportunity to be represented by counsel independent of counsel for the applicant, and in the case of any program operated by any department or agency of Federal, State, or local Government, is in fact so represented. (42 C.F.R. §2.65(c)(5))

11 25. The statute and regulations set out in paragraph 23 12 apply to the L.A.A.M. Program of Ward 93. None of the statutory and regulatory safeguards set forth in the previous paragraph 13 14 were observed by defendants in seeking and issuing the warrant, 15 or in seizing the property and information at Ward 93. Both the 16 warrant, and the subsequent search and seizure, therefore, were 17 invalid.

REQUISITES FOR RELIEF

19 26. Plaintiffs and the class they represent have been 20 and will continue to be irreparably injured by defendants' actions 21 in seizing and retaining the patient records of the L.A.A.M. 22 program from Ward 93, and copies thereof, in that plaintiffs' 23 protected rights of privacy and confidentiality have been violated 24 without legal justification. Moreover, Plaintiffs reasonably 25 fear that defendants will make future use of the information and 26 property invalidly seized pursuant to the search warrant and that 27 their identities will be revealed as participants in the L.A.A.M. 28 program. Such fears, and the fear of future similar violations

of plaintiffs' rights by defendants, have a chilling effect upon plaintiffs' willingness to participate in the L.A.A.M. program or other drug abuse programs, if necessary, or to seek medical assistance for drug-related conditions. Plaintiffs believe that information gathered for one purpose will make its way into police records which will adversely affect them in the future, including but not limited to future employment possibilities. 27. Plaintiffs (and the class they represent) have no adequate remedy at law for the injuries they have and will continue to suffer. 28. An actual controversy has arisen and now exists between plaintiffs and the class they represent and the defendants concerning their respective rights and duties in that plaintiffs contend that the search warrant and subsequent seizure, retention, and use of L.A.A.M. patient records was and is invalid and unlawful. Plaintiffs allege on information and belief that defendants believe that their actions were and are in all respects valid, and that the search warrant and subsequent seizure, retention and

use of L.A.A.M. patient records is lawful. 29.- Defendants' actions in obtaining the search warrant, in carrying out the subsequent search and seizure, and in retaining the records seized were and are unlawful in that defendants' actions violated the Drug Abuse Office and Treatment Act of 1972 (21 U.S.C. §1175(g)), and its implementing regulations (42 C.F.R.

§2.1 et seq.).

2

3

S 5

6

0

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

30. Plaintiffs refer to paragraphs 1 through 23 and 26 through 28 of the first cause of action and by reference incorp-

- 38 -

SECOND CAUSE OF ACTION

- 39 -

orate them as part of this cause of action.

1

2

ŝ

6

ີ7

8

9

10

11

12

13

14

15

16

17

23

24

25

26

27

28

31. Section 11977 of the Health and Safety Code of the Sfate of California provides in part:

> (a) Except-as otherwise provided in this subdivision (b), records of the identity, diagnosis, prognosis, or treatment of any patient which are maintained in connection with performance of any narcotic and drug abuse program shall be confidential and shall be disclosed only for the purposes and under the circumstances expressly authorized by this section. The content of any record referred to in this section may be disclosed in accordance with the prior written consent of the patient with respect to whom such record is maintained, or:

(3) If authorized by an appropriate order of a court of competent jurisdiction granted after application showing good cause therefor. In assessing good cause, the court shall weigh the public interest and the need for disclosure against the injury to the patient, to the physicianpatient relationship, and to the treatment services program. Upon the granting of such order, the court, in determining the extent to which any disclosure of all or any part of any record is necessary, shall impose appropriate safeguards against unauthorized disclosure.

Section 1282 of Title 9 of the California Administrative 18 Code sets forth additional confidentiality requirements for 19 20 methadone treatment programs, and incorporates the safeguards and protections contained in the federal statutes and regulations. 21 Section 1282 provides in part: 22

> All information and records obtained in the course of providing services to patients in a program shall be subject to the confidentiality and disclosure provisions contained in Article 7 (commencing with § 5325) of Chapter 2 of Part 1 of Division 5 of the Welfare and Institutions Code, and as is required by the applicable statutes and regulations of the Federal Government . . .

32. The statute and regulation set out in paragraph 29

apply to the L.A.A.M. program at Ward 93. None of the safeguards

- 40 -

lawful.

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

33. Plaintiffs refer to paragraphs 1 through 23 and 26 through 28 of the first cause of action and by reference incorporate them as part of this cause of action. 34. Article 1, Section 1, of the California Constitution provides that "privacy" is an "inalienable right" of all people. Defendants' seizure of patient records regarding participation in the L.A.A.M. program was an unreasonable governmental intrusion into plaintiffs' personal and objectively reasonable expectation of privacy. The seizure and retention of patient information therefore was and is illegal in violation of Article 1, Section 1, of the California Constitution.

35. Plaintiffs refer to paragraphs 1 through 23 and 26 through 28 of the first cause of action and by reference incorporate them as part of this cause of action. 36. Article 1, Section 13, of the California Constitution provides: "The right of the people to be secure . . against unreasonable seizures.and searches may not be violated " The ex parte issuance of the search warrant under the circumstances described herein was per se unreasonable. The warrant and subsequent search therefore violated Article 1, Section 13, of the California Constitution, and were illegal.

contained in said statute and regulation were observed by defendants in seeking and issuing the warrant, or in seizing the property and information at Ward 93. Both the warrant and the subsequent search and seizure, therefore, were invalid and un-

THIRD CAUSE OF ACTION

FOURTH CAUSE OF ACTION

- 41 -

FIFTH CAUSE OF ACTION

37. Plaintiffs refer to paragraphs 1 through 23 and 26 through 28 of the first cause of action and by reference incorporate them as part of this cause of action.

38. The Fourth Amendment of the United States Constitution provides: ". . . no Warrants shall issue, but upon probable cause, supported by oath or affirmation, and particularly describing the place to be searched, and the persons or things to be seized." Article 1, Section 13, of the California Constitu-10 tion provides: " . . . a warrant may not issue except on prob-11 able cause, supported by oath or affirmation, particularly des-12 cribing the place to be searched and the persons and things to be 13 seized." The affidavit in support of the warrant in this case 14 was insufficient to establish probable cause. The warrant and 15 subsequent search therefore violated the Fourth Amendment of the 16 United States Constitution, and Article 1, Section 13, of the 17 California Constitution, and were illegal.

18

2

. 3.

-1

5

6

7

8

9

PRAYER

19 WHEREFORE, plaintiffs pray judgment against defendants, 20 and each of them, as follows:

21 1. For an injunction, enjoining defendants, and each 22 of them, and their agents, servants, and employees, and all other 23 persons acting under, in concert with, or for them:

24 a. From making any copies of the records, docu-25 ments, or information taken from Ward 93 of San Francisco General 26 Hospital pursuant to the search warrant, or information derived 27 therefrom, or disseminating or disclosing to any persons or 28 entities such records, documents, information, or information

- 42 -

derived therefrom, or making any other use of such records, b. To return to Ward 93 of San Francisco General

documents, or information, or information derived therefrom; Hospital all records, documents and information seized therefrom pursuant to the search warrant, and any notes, memoranda or other writings which disclose the identity of patients on the L.A.A.M. program, and to make proof of compliance upon counsel for plaintiffs:

2

-4

5

6

8

10

11

12

13

14

15

16

17

18

20

22

23

-24

25

26

27

28

° 21

62 -

c. To destroy any and all previous copies, including photographs, of any and all records, documents, or information obtained from Ward 93 of San Francisco General Hospital pursuant to the search warrant, and to make proof of the date and method of destruction upon counsel for plaintiffs;

d. To submit to the court and plaintiffs' counsel a detailed listing of each and every person and agency to whom records, documents, or information seized from Ward 93 of San Francisco General Hospital pursuant to the search warrant, and all information derived therefrom, has been previously disseminated or disclosed, including the dates of such dissemination or disclosures, and to take all necessary affirmative steps which will lead to the return or destruction of all records, documents, information, and information derived therefrom which is in the possession, custody, or control of persons and agencies that are not expressly named as defendants herein, and to make proof of compliance upon counsel for plaintiffs; e. From taking any action either directly or in-

directly against plaintiffs and the class they represent for their participation in the L.A.A.M. program at Ward 93 of San

- 43 -

Francisco General Hospital or for bringing this law suit.

7

-1

11

14

15

16

38

19

20

21

22

23

24

25

2. That this court declare that the issuance of search warrants, seizures pursuant thereto, and the lack of any safeguards against unauthorized disclosure of the information seized, in the circumstances of this action, is illegal and void 5 6 in that issuance of such warrants violate applicable federal statutes and regulations, applicable California statutes and 7 8 regulations, the Fourth Amendment of the United States Constitu-.9 tion, and Article 1, Sections 1 and 13, of the California Consti-10 tution.

3. For costs of suit herein.

12 4. For plaintiffs' attorneys' fees pursuant to Code of 13 Civil Procedure §1021.5 and California law.

5. For such other and further relief as the Court may deem equitable and proper.

> Dated: _____, 1979**.**

Attorneys for Plaintiffs

Attorneys for Plaintiffs

VERIFICATION

I, the undersigned, say:

26 I am one of the attorneys for the plaintiffs. The 27 named plaintiffs, John Doe I and John Doe II, reasonably believe 28 that mere disclosure of their names or identitites in this comtrue and correct.

1

5

6

7

8

0

10

11

12

13

14

15

.16

17

18

19

20

21

22

23

24

25

26

.27

28

plaint will directly abridge the rights and interests which they seek to vindicate by bringing this action, and for that reason they are unable to verify this complaint.

I have read the complaint and I am informed and believe the matters stated therein to be true, and on that ground allege that the matters stated therein are true.

I declare under penalty of perjury that the above is

Executed on August 27, 1979, at Oakland, California.

Earl D. Osborn

STATE OF CALIFORNIA) SS SEARCH WARRANT		
THE FEOPLE OF THE STATE OF CALIFORNIA, TO ANY POLICEMAN OR PEACE OFFICER IN THE CITY AND COUNTY OF SAM FRANCISCO, STATE OF CALIFORNIA:	1	Your affia
Proof by affidavit having been made this day before me by Sqt. E. Rvan,	2	officer for the City
San Mateo P.D. and it appearing therefrom that there is probable cause for believing that there is now located at Ward 93, San Francisco General Hosnital, 1001 Potrero Street, San Francisco	3	has been assigned as
San Francisco, California, certain personal property or things consisting of the	4	Your affia
following: Patient rosters, rolls, and records, including	5	cases. Your affiant
names, addresses, and dates of birth of patients	6	hundred heroin users
	7	attended classes con
· · · · · · · · · · · · · · · · · · ·	8	Justice, Drug Enforce
	9	Bureau of Investigat:
and that said property comes within the provisions of Section 1524 of the Penal	10	in-service courses r
Code as noted herewith:	11	Your affiant has into
a. <u>Subdivision 1</u> (Stolen property) b. <u>Subdivision 2</u> (Property or things used as a means of committing a felony)	12	and questioned them a
c. // Subdivision 3 (Property or things in possession with intent to use it to	13	their crines. Your :
commit public offense or to conceal it from discovery)	14	field of narcotics in
d. XXX Subdivision 4 (Property or things are evidence which tends to show a felony has been committed	15	The followi
or that a particular person has committed it)	16	Homicide Sergeant Lar
YOU ARE THEREFORE CONMANDED to make a search of the premises or person described above for the said articles and property, and if you find the same or	17	scene shortly after t
any part thereof to bring it forthwith before me or retain it in your custody according to Section 1535 of the California Penal Code.	18	Between the
GIVEN UNDER MY HAND and dated February 7, 19 79.	19	evening of 4 February
CPD	20	juveniles, were murde
Judge of the Municipal Court	21	All three persons wer
GOOD CAUSE APPEARING, YOU ARE HEREBI AUTHORIZED TO serve this warrant during the nighttime.	22	the Payless Drug Stor
Judge of the Municipal Court	23	Suspect(s) in the cas
In and for the City and County of San Francisco State of California	24	.38 calibre revolver
1531 P.C. Announce that you are a peace officer with search Warrant	25	victics survived long
is authorized	- 26	suspact(s) description
1534 P.C. Warrant must be executed within 10 days	0	
003-C	4617A3, JR.	
- 46 - Jan 1997 - Jan 1	ALTONALY	

EVHIBIT "A"

int, Sergeant Eacon Ryan, has been a police r of San Mateo for the past twenty years and an investigator for the past five years. nt has investigated hundreds of narcotics has been involved in the arrest of over two within the past year. Your affiant has ducted by the United States Department of ement Administration, the State of California, ion and Narcotics Enforcement, and numerous elated to homicide and narcotics investigations erviewed many admitted burglars and robbers regarding the manner in which they commit affiant has testified as an expert in the n the courts of the County of San Mateo. ing information was related to affiant by rry Eeissel who was personally present on the the events occurred.

e hours of 7:00 P.M. and 8:00 P.M. on the y, 1979, three males, two of which were ered in the City and County of San Mateo. re working, at the time of their demise, at re, 656 Concar Drive, in San Mateo. se executed all three employees by firing a into the back of their heads. None of the g enough to supply responding officers with a on. Suspect(s) removed approximately 550,000

- 47 -

in store receipts from the premises. Numerous latent fingerprint: 1 were obtained by law enforcement personnel from the scene of the 2 robbery/nomicides.

Within the past 48 hours your affiant was contacted by Detective Ted Spyrow of the Concord Police Department. He stated that he had just received a telephone call from an anonymous citizen informant, regarding the aforementioned nurders. The informant, hereinafter designated in this affidavit as the "CI", 8 stated that an acquaintance of his had a conversation with the 9 perpetrators of the robbery/homicide that occurred at Payless 10 Drugs in San Mateo. The CI stated that the perpetrators had 11 talked with his acquaintance approximately four days prior to the 12 robbery/honicides in San Mateo. At that time, the CI relates the 13 the suspects were attempting to obtain guns to "do the Payless 14 Store in San Mateo". CI states that the suspects related they 15 would "clean out the drugs" and attempted to solicit his 16 acquaintance to participate in the robbery, however CI states the 17 his acquaintance was not interested. 13

CI further relates that the suspects had this conversa-19 tion with his acquaintance at the "Lan Clinic at San Francisco 20 General Hospital". CI states that his acquaintance and the 21 perpetrators of the homicides are patients of the Lan Clinic. 22

Affiant has checked the City of San Mateo for Paylass Stores. The only Payless Store in the City is the one in which the triple nurder robbery occurred.

- 48 -

Your affiant has received information from the San

without a court order. Francisco General Hospital.

2

Ŀ

:9.

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

EPH PASITAL JA

STRICT ATTONNEY

------STA . CT A 17 - 5- 4 7

23

24

25

26

3

Francisco General Hospital indicating that the Lam Clinic is a Rethedone Maintenance Program for heroin addicts. The Lan Program is located on Ward 93 of San Francisco General Hospital. Your affiant contacted the Director of the Lam Program on 6 February, 1979. Your affiant requested a list of the patients presently enrolled in the Lam Program. Your affiant was informed that this information is confidential and cannot be revealed

Fron the information contained in this affidavit, as supplied by a citizen informant, your affiant has reasonable caus to believe that the identity of the robbery/murder suspects may be contained on the patient rolls of the Lan Program at the San

. On February 6, 1979, your affiant was present when Inspector Marvin Dean of the San Francisco Police Department contacted Arthur Weinberg, Director of the Lam Clinic by telephon Dean informed Weinberg that your affiant would be attempting to obtain a search warrant on February 7 in order to determine the identity of all patients at the Lan Clinic./

On February 7, your affiant telephoned Arthur Weinberg as required by 42 Code of Federal Regulation 2.65(e), and

informed him that your affiant would be contacting a Judge of the San Francisco Municipal Court later during the morning of

February 7 to request a search warrant, and that a representating or estorney of the Les Glinic was invited to also appear if

sized by the Glimic.

SAN FRANCISCO, STATE OF CALIFORNIA Accordingly, your affiant prays that a search warrant b: STATE OF CALIFORNIA) SS RETURN AND INVENTORY CITY AND COUNTY OF SAN FRANCISCO) ON SEARCH WARRANT 1 issued for the records and patient rosters and/or rolls of the Lat 2 Program located on Ward 93 of the San Francisco General Hospital. 3 1001 Potrero Street, San Francisco, in order that your affiant night ascertain the identity, dates of birth, and last known 5 addresses of the program participants, so that your affiant can 6 attempt to compare the latent prints collected from the scene to INVENTORY: Photocopy list of 35 individuals identified by clinic fingerprints of those persons whose names are contained on the 8 rolls of the Lan Program and to possibly contact or interview Q those persons. 10 11 • • • nhotos of the LAAM natients 12 13 14 15 1. 16 17 I, the officer by whom this search warrant was executed, do 18 19 will remain in the custody of the San Mateo Folice Department subject to further order of this Court of other court of proper 20 21 Inventory approved. Subscribed and sworn to before me---22 this ______ day of February 1979 1. . . 23 Judge of the Municipal Court 24 City and County of San Francisco, State of California 25 1535 p.m. Leave copy of warrant and list of property taken with person 26 no one there. after being sworn; file with Clerk. SEPH PPEITAS, JA, - 50 -STAICT ATTUATEY 003-C - 51 -

I, the undersigned make this return to the within scarch warrant. On February 7, 1979 , I received said warrant, and under its authority I diligently searched the below listed ... premises on (date) . February, 7, 1978 ... and there I discovered the matter described in the inventory. Premises searched: ----Ward-93, San Francisco General Hospital ----Records of patients in the LAAM Program staff as being enrolled in the LAAM methadone program. Information given to this officer by clinic personnel included names, addresses, and dates of birth of the enrollees. Photographs were taken of hospital identification

swear that the above inventory contains a true and detailed account of all the property taken by me on the warrant. The property seized

Serneant E. Avan, #18, San Mateo P.D.

person from whom it was taken; leave receipt on premises if 1537 p.m. Return warrant to Judge; sign inventory in Judge's presence

DEPARTMI OF HEALTH, EDUCATION, AN WELFARE

77-29

December 8, 1977

Mr. Richard D. Bybee, Staff Attorney Office of the Attorney General P.O. Box 11549 Columbia, South Carolina 29211

Dear Mr. Bybee:

We regret the delay in responding to your letter of October 13, 1977, which requests clarification of our October 3, 1977, response to your inquiry of August 16, 1977, regarding the effect of the "Confidentiality of Alcohol and Drug Abuse Patient Records" regulations, 42 CFR Part 2.

Your latest inquiry questions the effect of the confidentiality regulations where a law enforcement official seeking to serve an arrest warrant knows that a particular individual is residing at an alcohol or drug abuse treatment center. In addition, you question whether the confidentiality regulations would permit a disclosure to law enforcement officials of the fact that a particular individual is not and has never been a patient in the alcohol or drug abuse treatment program from which the information is sought.

In response to your first question, we call your attention to the following paragraph in the May 10, 1976, opinion letter to Mr. Gardner which was enclosed in our October 3, 1977, letter to you:

"In the situation which you have presented, it is our conclusion that the regulations do not authorize you to assist a law enforcement officer by identifying, either directly or indirectly, any individual who is or has ever been a patient in the program. This applies regardless of the fact that a law enforcement officer may have a valid warrant for the arrest of the individual, As indicated above, section 2.13(e) provides, in pertinent part, that 'any improper or unauthorized request for any disclosure of records or information subject to this part must be met by a noncommittal response.

⊐ off	FOULT	Shouldhebe	maire	lonnot	Ionced)	torite:	1VEI and	SURNAME	0.
2	GH	Tannan	12/11/		u.			4	
			heriout						
Y									1

- 52

Page 2 - Mr. Richard D. Bybee

it should be explained that Federal regulations pro-... hibit the treatment program and its personnel from disclosing any information about a patient ... unless a court order authorizes such disclosure pursuant - to Subpart E of the regulations." (Emphasis in original.)

Thus, the confidentiality regulations do not permit alcohol or drug abuse treatment program personnel to assist a law enforcement officer in identifying or locating a patient where such action would result in a disclosure identifying the patient as an alcohol or drug abuser unless an authorizing court order has been entered in accordance with Subpart E of the regulations. This conclusion applies even though the law enforcement official knows that the patient is present. (Section 2.13(b) provides that the regulatory prohibitions apply irrespective of whether the person seeking disclosure already has the information sought.) The regulations do not. require that the personnel of a treatment program forcibly restrain or otherwise take action to prevent a law enforcement official from serving an arrest warrant, but rather restrict any disclosure of information whether / recorded or not. which would identify an individual as an alcohol or drug abuser. While the confidentiality regulations do not prohibit, or require treatment program personnel to prohibit, a law enforcement official from locating an individual for the purpose of serving an arrest warrant, we believe this situation should be avoided since it may result in disclosures of information which would identify other patients or be disruptive of the operation of the program. Accordingly, we urge that treatment programs provide information to-local law enforcement agencies regarding the restrictions of the regulations and seek to enter into cooperative arrangements which will permit, to the extent possible, a reconciliation of the interests of the law enforcement agencies and of the interests of the program in protecting the confidentiality of its patient records.

In response to your second question, it is our opinion that alcohol or drug abuse treatment program personnel may advise law enforcement officials or other persons that a particular individual is not and has never been a patient. In that المريسية أأمين فتوادع المراجع المراجع

- 53 -

Page 3 - Mr. Richard D. Bybee

situation, the confidentiality regulations would not be applicable because there would be no "record" 1/ of the identity, diagnosis, prognosis, or treatment of any "patient" 2/ maintained in connection with the performance of any alcohol abuse or drug abuse prevention function which is directly or indirectly federally assisted as set forth in § 2.12(a) of the regulations. Thus, where an individual has never been a patient as defined by the regulations, there would be no "record" subject to the regulatory restrictions on disclosure.

It has been suggested that requests for records or informa-... tion to which the confidentiality regulations would not otherwise apply must be responded-to in the same manner as requests for records and information which are subject to * +# ± * • • • the regulations, in order to avoid an implicit identification and disclosure of alcohol or drug abuse patient records (implicit and negative disclosures are prohibited by § 2.13(e)). 3/ However, it is our opinion that the confidentiality regulations do not prohibit a disclosure that an individual is not and has never been an alcohol or drug abuse patient, even though the request for information

.

Section 2.11(o) of the confidentiality regulations 1/ defines the term "records" to include "any information, whether recorded or not, relating to a patient received or acquired in connection with the performance of any alcohol abuse or drug abuse prevention function, whether such receipt or acquisition is by a program, a. qualified service organization, or any other person. (Emphasis added.)

Section 2.11(1) of the confidentiality regulations ... defines the term "patient" to mean.

"Any individual (whether referred to as a " patient, client, or otherwise) who has applied for or been given diagnosis or treatment for drug abuse or alcohol abuse and includes any . individual who, after arrest on a criminal charge, is interviewed and/or tested in connection with drug or alcohol abuse preliminary to a determination as to eligibility to participate in a treatment or rehabilitation program.".

See: Helms, A Guide to the New Federal Rules Governing the Confidentiality of Alcohol and Drug Abuse Patient ... Records, Medical Record News, August, 1976, 7 at 13. (Copy enclosed).

- 54 -

Page 4 - Mr. Richard D. Bybee

regarding that individual may accompany a request for information regarding an alcohol and drug abuse patient to which a noncommittal response must be made. We reach this conclusion because: (1) as noted above, it is clear that the applicability provisions of the regulations encompass only alcohol or drug abuse patient records which are maintained in connection with the performance of any federally assisted · · · · alcohol abuse or drug abuse prevention function; (2) the implicit and negative disclosures section of the regulations ... (§ 2.13(e)) does not clearly prohibit such a disclosure, but ... rather provides that "[a]ny improper or unauthorized request for any disclosure of records or information subject to this part must be met by a noncommittal response (Emphasis added.); and (3) violations of the regulations are subject to a criminal penalty (see §§ 2.14 and 2.14-1) and, therefore, the same strict rule of construction as is applied to statutes defining criminal action must be applied to the regulations. 4/ Please contact us if you have any further questions on this

Enclosure cc: Ms . Susan Greene.

See: <u>M. Kraus & Bros. v. United States</u>, 327 U.S. 614, 621-622, 66 S. Ct. 705, 707-708 (1946).

It is, of course, well settled that criminal statutes are to be construed narrowly and that any ambiguity ... must be resolved in favor of lenity. See, e.g., United States v. Emmons, 410 U.S. 396, 411, 93 S. Ct. 1007, 1015 (1973); Rewis v. United States, 401 U.S. 808, 812, 91 S. Ct. 1056, 1059 (1971); United States v: Bass, 404 U.S. 336, 347-49, 92 S. Ct 515, 522-23 (1971).

matter, Sincerely yours,

Robert B. Lanman Senior Attorney Public Health Division

Prepared by: GH, LANMAN: ack, 12/8/77, 443-3096

- 55 -

JUN 7 1977

E. Kontz Bennett, Seulor, Esq. Bennett, Pedrick & Bennett P.O. Box 178 Waycross, Georgia 31501

Dear Mr. Bennett!

This is in response to your request of May 9, 1977, for a legal opinion on the effect of the federal alcohol abuse and drug abuse confidentiality statutes, 42 U.S.C. 4582 and 21 U.S.C. 1175, and the NEW regulations implementing those statutes, 42 CFR Part 2 1/ (copy enclosed), on a hospital's obligation under state or local law or court order to report instances of "possible alcohol and drug abuse" to law enforcement officials.

The Ware County Grand Jury presentment which you enclosed recommends that the District Attorney's Office act to require Memorial Hospital to report "all drug and or criminal cases to local authorities." Further, your letter indicates that the Assistant District Attorney believes that "the Hospital should report to local law enforcement officials the names of persons coming to its attention where drug or alcohol abuse are possibly involved."

These statutes and regulations pertain to the confiden-1/ tiality of alcohol and drug abuse patient records and are not, as your letter indicates, part of the federal "Privacy Act," 5 U.S.C. 552a, which pertains to a broad range of records about individuals which are maintained by the Federal Covernment, or by Federal contractors in the performance of an agency function.

- 56 -

Page 2 - E. Kontz Bennett, Senior, Esq.

Pursuant to the authority of subsection (g) of 21 U.S.C. 1175 and 42 U.S.C. 4582, 2/ the HEW confidentiality regulations restrict the disclosures that may be made from records 3/ of the identity, diagnosis, prognosis, or treatment of any patient, 4/ which are maintained in connection with the

Subsection (g) of the confidentiality statutes gives broad authority for the prescription of regulations to carry out their common purposes, providing in pertinent part: >

> "The regulations may contain such definitions, and may provide for such safeguards and procedures, including procedures and criteria for the issuance and scope of orders under subsection (b) (2) (C), as in the judgment of the Secretary are necessary or proper to effectuate the purposes of this section, to prevent circumvention or evasion thereof, or to facilitate compliance therewith."

The regulations define "records" broadly in § 2.11(o) as: 3/

> "Any information, whether recorded or not, relating to a patient, received or acquired in connection with the performance of any alcohol abuse or drug abuse prevention function, whether such receipt or acquisition is by a program, a qualified service organization, or any other person."

Section 2.11(i) of the regulations defines "patient" as: 41

"... any individual (whether referred to as a patient, client, or otherwise) who has applied for or been given diagnosis or treatment for drug abuse or alcohol abuse and includes any individul who, after arrest on a criminal charge, is interviewed and/or tested in connection with drug or alcohol abuse preliminary to a determination as to eligibility to participate in a treatment or rehabilitation program.

(ma)

- 57 -

Page 3 - E. Kontz Bennett, Senior, Esq.

performance of any alcohol or drug abuse prevention function 5/ conducted, regulated, or directly or indirectly assisted by any department or agency of the United States. Direct and indirect forms of federal assistance which make a treatment provider's alcohol or drug abuse records subject to the regulations are set forth in § 2.12 "Applicability-Rules." Note that the alcohol and drug abuse patient records of many hospitals which perform alcohol or drug abuse prevention functions are subject to the regulations because of assistance by the Internal Revenue Service "through the allowance of income tax deductions for contributions ... or by way of a tax exempt status" (see (2.12(a)(4)) and (2.12-1(d)). Our conclusions about the effect of the federal confidentiality statutes and regulations on Memorial Mospital's obligation to make reports to local authorities are based upon the assumption that the hospital is subject to the confidentiality statutes and regulations because it performs alcohol and drug abuse prevention functions which are federally assisted within the meaning of § 2.12.

Disclosures of patient records by federally assisted hospitals for the purpose of initiating or substantiating any criminal charges against a patient or for the purpose of investigating a patient may only be made in accordance with subsections (b)(2)(C) and (c) of the federal confidentiality statutes (21 U.S.C. 1175 and 42 U.S.C. 4582) which provide:

"(Ъ) ...

(2) Whether or not the patient, with respect to whom any given record ... is maintained, gives his written consent,

An alcohol abuse or drug abuse prevention function is 5/ defined in § 2.11(k) of the regulations as:

> "any program or activity relating to alcohol abuse or drug abuse education, training, treatment, rehabilitation, or research, and includes any such function even when performed by an organization whose primary mission is in the field of law enforcement or is unrelated to alcohol or drugs."

> > - 58 -

7

(C) If authorized by an appropriate order of a court of competent jurisdiction granted after application showing good cause therefor. In assessing good cause the court shall weigh the public interest and the need for disclosure against the injury to the patient, to the physicianpatient relationship, and to the treatment services. Upon the granting of such order, the court, in determining the extent to which any disclosure of all or any part of any record is necessary, shall impose appropriate safeguards against unauthorized disclosure. (c) Except as authorized by a court

order granted under subsection (b) (2) (C) of this section, no record referred to in subsection (a) may be used to initiate or substantiate any criminal charges against a patient or to conduct any investigation of a patient." (Emphasis added.)

These statutory provisions are implemented by Subpart E (\$§ 2.61-2.67-1) of the confidentiality regulations. Under § 2.65 of that subpart a court may suthorize a disclosure of patient records for the purpose of conducting an investigation of, or a prosecution for, a crime of which the patient is suspected only if the court specifically finds that:

> "(1) The crime was extremely serious, such as one involving kidnapping, homicide, assault with a deadly weapon, armed robbery, rape, or other acts causing or directly threatening loss of life or serious bodily injury, or was believed to have been committed on the premises of the program or against personnel of the program.

(2) There is a reasonable likelihood that the records in question will disclose

Page 4 - E. Kontz Bennett, Senior, Esq.

the content of such record may be disclosed as follows:

- 59 -

Page 5 - E. Kontz Bennett, Senior, Esq.

material information or evidence of substantial value in connection with the. investigation or prosecution.

(3) There is no other practicable way of obtaining the information or evidence.

(4) The actual or potential injury to the physician-patient relationship in the program affected and in other programs similarly situated, and the actual or potential harm to the ability of such programs to attract and retain patients, is outweighed by the public interest in authorizing the disclosure sought."

Under § 2.63, an authorizing court order entered under § 2.65 "may not extend to communications by a patient to personnel of the program, but shall be limited to the facts or dates of enrollment, discharge, attendance, medication, and similar objective data" unless the patient "in litigation offers testimony or other evidence pertaining to the content of his communications with a program."

Thus, a hospital which performs alcohol or drug abuse prevention functions which are federally assisted may only report the names of alcohol or drug abuse patients and information relating to them to local law enforcement officials pursuant to an authorizing court order issued by a court of competent jurisdiction in accordance with subsections (b) (2) (C) and (c) of 21 U.S.C. 1175 and 42 U.S.C. 4532, and Subpart E of the regulations. The regulations do not, however, restrict reports to law enforcement officials which do not contain names or other patient identifying information. Section 2.11(p)(3) provides that the following type of communication is not a disclosure restricted by the regulations:

> "... (3) Communications of information which includes neither patient identifying information nor identifying numbers assigned by the program to patients."

Patient identifying information is defined in § 2.11(j) as:

"... the name, address, social security number, or similar information by which the identity of a patient can be determined with reasonable accuracy and speed

- 60 -

either directly or by reference to other publicly available information. The term does not include a patient identifying number assigned by a program."

The effect of these provisions upon the reporting of crimes committed on program premises or against personnel of the program is reflected in § 2.13(d) of the regulations. That section provides:

> "Where a patient commits or threatens to commit a crime on the premises of the program or against personnel of the program, nothing in this part shall be construed as prohibiting personnel of the program from seeking the assistance of, or reporting such crime to a law enforcement agency, but such report shall not identify the suspect as a patient. In any such situation, immediate consideration should be given to seeking an order under Subpart E of this part to permit the disclosure of such limited information about the patient as may be necessary under the circumstances."

The foregoing discussion of the requirements of the federal confidentiality statutes and regulations may be surmarized as follows: (1) the statutes and regulations do not restrict the reporting of crimes to law enforcement personnel so long as the reports do not contain the names of alcohol or drug abuse patients or other patient identifying information; (2) such patient identifying information may be reported to law enforcement personnel only if an authorizing court order is entered in accordance with Subpart E of the regulations; and (3) since Subpart E of the regulations limits the entry of such orders to disclosures for the purpose of investigating or prosecuting extremely serious crimes (acts causing or directly threatening; loss of life or serious bodily injury) and those believed to have been committed on the premises of the program or against personnel of the program, the reporting of crimes which do not come within these categories is absolutely prohibited.

The letter from Mr. Stubbs, the Executive Assistant Attorney General of Georgia, which you enclosed indicates that there are no state statutory reporting requirements in Georgia pertaining to driving under the influence, drug abuse, or

Page 6 - E. Kontz Bennett, Senior, Esq.

- 61 -

Page 7 - E. Kontz Bennett, Senior, Esq.

gunshot wounds, but that "suppression of direct evidence" of such criminal conduct "might be an offense under Georgia statutes (Georgia Code Annotated § 26.2503) unless that information is otherwise privileged by a specific statute as, for example, Georgia Code Annotated § 54-6318." Thus, it is not clear that Georgia law requires the type of reporting which is recommended by the Grand Jury and sought by the District Attorney.

However, to the extent state or local law is interpreted to require such reporting, it is superseded by the restrictions of the federal statutes and by the regulations promulgated thereunder. As provided in § 2.23 of the regulations "no State law ... may either authorize or compel any disclosure prohibited by this part." Since the use of patient records for the purpose of initiating or substantlating criminal charges against a patient or to conduct an investigation of a patient is specifically limited by 21 U.S.C. 1175(c) and 42 U.S.C. 4562(c), those statutory provisions and the regulations implementing them would supersede any conflicting state law under the Supremacy Clause (Article VI, Clause 2) of the United States Constitution. See New York v. Dublino, 413 U.S. 405, 424 n.29, 93 S. Ct. 2507, 2513 n.29, Ming V. Smith, 392 U.S. 309, 333 n.34, 88 S. Ct. 2128, 2141 n.34 (1968).

Sincerely yours,

Richard Beattle Deputy General Counsel

Enclosure

Prepared by: GH, LANMAN: GREENE: ack, 6/2/77, 443-3096

- 62 -

Mr. Jimmy B. Laster Chief of Fulice Rogers Police Department 212 West Lin Street Rogers, Arkansas 72756

Dear Mr. Luster:

Your letter to the Attorney General regarding the federal laws on the confidentiality of drug abuse patient records has been referred to this office for response. We regret the delay in answering your inquiry.

The federal statute governing the confidentiality of drug abuse patient records is 21 United States Code (USC) 1175. 1/ It is set forth in § 2.1 of the Confidentiality of Alcohol and Drug Abuse Patient Record regulations, 42 Code of Federal Regulations (CFR) Part 2, a copy of which is attached for your use. These regulations are authorized by subsection (g) of 21 U.S.C. 1175 and subsection (g) of 42 USC 4582, a comparable statute protecting alcohol abuse patient records. Thus, the regulations have the force and effect of federal law.

The basic purpose of the statute, 21 USC 1175, is to restrict the circumstances under which disclosures of information may be made from the records of a drug abuse patient. The statute, as implemented by the regulations, applies to patient "records" 2/ maintained in connection

- performing a function of a federal agency.

Section 2.11(o) of the regulations defines records broadly to include "any information, whether recorded or not, relating to a patient." Thus, the regulations restrict disclosures of any patient related information by programs to which the regulations are applicable. (See §§ 2.12 and 2.12-1 regarding applicability of the regulations.) For example, drug abuse patient records in a hospital or mental health clinic which receives federal grant funds to provide drug abuse treatment are subject to the restrictions of 21 USC 1175 and the implementing regulations. Prepared by: Gll, SNGreene: RBLanman: 7/20/77, 443-3096 D.F. 1253

July 20, 1977

We note that neither the Freedom of Information Act, 5 USC 552, nor the Privacy Act, 5 USC 552a, are pertiment. They apply only to records maintained by a federal agency or in the case of the Privacy Act, to records maintained by a federal contractor Page 2 - Mr. Jimmy B. Luster

5/

The set have been

with the performance of a "drug abuse prevention function" (including drug abuse treatment) that is directly or indirectly assisted by the federál government. 3/

The Congressional debates indicate that a fundamental objective of the Drug Abuse Office and Treatment Act of 1972, Pub. L. 92-255, of which the drug abuse patient confidentiality provision is a part, was to increase the availability of drug abuse treatment in order to counteract the rising erime rate attributable to drug addicts. 4/ The confidentiality provision was intended to facilitate this objective by encouraging addicts to seek such treatment. 5/ Toward this end, the statute clearly implements the Congressional intention to restrict access by law enforcement officials to drug abuse patient records but not to completely prohibit it. Note in particular the restrictions in subsection (c) of the statute which provides:

- See 21 USC 1175(a). An (alcohol or) "drug abuse prevention function" 3/ is defined in § 2.11(k) of the regulations. The types of direct or indirect federal assistance which will subject a recipient's alcohol. or drug abuse patient records to the regulations are set forth and discussed in § 2.12 and § 2.12-1.
- 4/ Congressional Record, Vol. 117, Part 34, 92d Cong., 1st Sess., pp. 44085-6, 44099; Scn. Rep. 92-509, 92d Cong., 1st Scss., pp. 2-4, 13.

"The conferces wish to stress their conviction that the strictest adherence to the provisions of this section is absolutely essential to the success of all drug abuse prevention programs. Every patient. and former patient must be assured that his right to privacy will be protected. Without that assurance, fear of public disclosure of drug abuse or of records that will attach for life will discourage thousands from seeking the treatment they must have if this tragic national problem is to be overcome.

Every person having control over or access to patient's records must understand that disclosure is permitted only under the circumstances and conditions set forth in this section. Records are not to be made available to investigators for the purpose of law enforcement or for any other private or public purpose or in any manner not specified in this section."

H.R. Report No. 92-920, 92d Cong., 2d Sess. 33 (1972): 2 U.S. Code Cong. & Ad. News 2045, 2071-72 (1972).

Page 3 - Mr. Jimay B. Luster

"Except as authorized by a court order under (b)(2)(C) of this section, no record referred to in subsection (a) may be used to initiate or substantiate any criminal charges against a patient or to conduct any investigation of a patient."

Thus, a court of competent jurisdiction may authorize the disclosure of confidential information pertaining to a drug abuse patient for purposes of investigation or prosecution but only in accordance with subsection (b)(2)(C) of the statute and Subpart E of the regulations which implements subsection (b)(2)(C). Both subsection (b)(2)(C) of the statute and Subpart E of the regulations set forth determinations which a court must wake to authorize a disclosure and require that restrictions be imposed on the disclosures which are authorized. In particular, § 2.65 in Subpart E, which applies to applications by an investigative. law enforcement or prosecutorial agency for an order authorizing a disclosure, provides in pertinent part:

> prosecution. sought.

- 64 -

"(c) ... A court may authorize disclosure of records pertaining to a patient for the purpose of conducting an investigation of or a prosecution for a crime of which the patient is suspected only if the court finds that all of the following criteria are met: " (1) The crime was extremely serious, such as one involving kidnapping, homicide, assault with a deadly weapon, armed robbery, rape, or other acts causing or directly threatening loss of life or serious bodily injury, or was believed to have been committed on the premises of the program or against personnel of the program.

" (2) There is a reasonable likelihood that the records in question will disclose material information or evidence of substantial value in connection with the investigation or

" (3) There is no other practicable way of obtaining the information or evidence. " (4) The actual or potential injury to the physician-patient relationship in the program affected and in other programs similarly situated, and the actual or potential harm to the ability of such programs to attract and retain patients, is outweighed by the public interest in authorizing the disclosure

- 65 -

Page 4 - Mr. Jirmy B. Luster

"(d) ... Both disclosure and dissemination of any information from the records in question shall be limited under the terms of the order to assure that no information will be unnecessarily disclosed and that dissemination will be no wider than necessary. Under no circumstances may an order under this section authorize a program to turn over patient records in general, pursuant to a subpoena or otherwise, to a grand jury or a law enforcement, investigative, or prosecutorial agency."

Therefore, to obtain an order authorizing a federally assisted program, such as the Ozark Guidance Center, to disclose information pertaining to a drug abuse patient, a law enforcement agency would be required to demonstrate to a court having jurisdiction over the center that (1) the crime being investigated is extremely serious, i.e., threatening loss of life or serious bodily injury, and (2) the other criteria set forth in § 2.65 and quoted above are met. Note that \$\$ 2.61 through 2.64 of Subpart E are applicable to this proceeding as well.

We trust this analysis clarifies the purpose, scope, and application of the federal taws pertaining to the confidentiality of drug * and alcohol abuse patient records.

- 66

Sincerely yours.

Robert B. Lanman Senior Attorney Public licalth Division

Enclosure

cc: Mr. Ed Gleiman, OAMB, FIPS

MEMORANDUM

10 : Ms. Sheila Gardner Staff Assistant for Confidentiality Compliance, DCA, NIDA

FROM : Attorney Advisor Public Health Division

SUBJECT: Cooperative Agreements Between Drug Treatment Programs and Local Police Departments--42 CFR Part 2--Communications Not Prohibited by the Regulations, Including Communications Under § 2.13(d) and Those Not Constituting Disclosures Under § 2.11(p)(3)--Disclosures of Patient Records With Patient Consent Under § 2.40 and With an Authorizing Court Order Under § 2.65--GII Ref. No. 78-2481 (D.F. #25B)

In response to your August 31, 1978, request, we are unable to provide legal clearance for the cooperative agreements between the drug treatment programs and their respective police departments in Charlotte, North Carolina, and Flint, Michigan, because, in our view, the agreements are not in compliance with 21 U.S.C. 1175 and 42 CFR Part 2, the confidentiality of drug abuse patient records statute and regulations.

Two memoranda attached to your request describe the essential nature of these agreements (for your convenience, copies of these memoranda are attached to this response). The memorandum, City of Charlotte, North Carolina (Miller) to Bureau Commanders, November 7, 1973, states in paragraph three:

"Open House has stated that it has no desire to be a sanctuary for arrest and will cooperate with the Police Department in serving warrants. All clients will sign a release which will allow the staff to state who is on the premises." (Emphasis added.)

DEPARTMENT OF HEALTH, EDUCATION, AND WELFARE OFFICE OF THE SECRETARY

OFFICE OF THE GENERAL COUNSEL

DATE: January 24, 1979

- 67

Page 2 - Mg. Sheila Gardner

The memorandum, City of Flint, Michigan, Police Division (Gilmore) to Durbin, Chief of Police, April 24, 1978, states in paragraphs one and five, respectively:

"On April 19, 1978, the undersigned officer held a meeting with the staff at the Rubicon-Odyssey House, 1125 Detroit Street. As suggested at a prior meeting, the staff had prepared a form whereby information relative to residents could be released to the Flint Police Department for official purposes. Copies of that form are attached." (Emphasis added.)

"Information will be provided to the officer relative to the location of the subject in guestion "

Apparently, these agreements envision the disclosure of patient records for purposes of serving arrest warrrants upon or conducting criminal investigations of the patients whose records are disclosed.

Because such disclosures may be made only as authorized under the regulations 1/ and the attached agreements either do not specify the appropriate authorization which must be obtained or incorrectly set forth the basis for such authorization, we are unable to give legal clearance to the agreements.

We have previously concluded that under 2.11(p)(3) of the regulations (which states that "communications of information which includes [sic] neither patient identifying information nor identifying numbers assigned by the program to patients" do not constitute disclosures of records)

§ 2.13(c) of the regulations states:

"The prohibition on unauthorized disclosure covers all information about patients, including their attendance or absence, physical whereabouts, or status as patients, whether or not recorded, in the possession of program personnel, except as provided in paragraph (d) of this section [which discusses crimes on program premises or against program personnel]."

Thus, disclosures of the physical whereabouts of patients or of their patient status are prohibited except as authorized under the regulations.

- 68 -

Page 3 - Ms. Sheila Gardner

programs may communicate the names or whereabouts of individuals who are patients so long as the individuals are not identified as patients. In an opinion letter, GH (Lanman) to Karten, Feburary 1, 1978 (D.F. #25B), we reached this conclusion with respect to general hospitals that treat a variety of medical conditions because we believe that under those circumstances communications of patient information may be made without identifying the patients as alcohol or drug abuse patients. However, we believe that it is unrealistic as a practical matter to expect that communications of the names or whereabouts of patients by drug abuse programs which do not treat a variety of medical conditions can be made without disclosing the patients' status as drug abuse patients. Thus, assuming that Open House and Rubicon-Odyssey House are not part of a general medical facility such as a community hospital, we conclude that the agreed to disclosures are not within the exception to the regulations provided by § 2.11(p)(3) and, thus, are prohibited except as authorized by the regulations.

The attached cooperative agreements apparently envision that written patient consent will be obtained under Subpart C of the regulations before a patient's identity or physical whereabouts is disclosed under the agreements. In our view, however, such consent would not authorize the agreed to disclosures because those disclosures would constitute use of a patient record to conduct an investigation or prosecution of the patient within the meaning of 21 U.S.C. 1175(c) and 42 CFR § 2.65.

21 U.S.C: 1175(c) provides:

"Except as authorized by a court order granted under subsection (b) (2) (C) of this section, no record referred to in subsection (a) may be used to initiate or substantiate any criminal charges against a patient or to conduct any investigation of a patient."

In our view, this section requires that disclosures of a patient record, including a patient's identity or physical whereabouts, to a police officer for the purposes of serving an arrest warrant or conducting an investigation of the patient or other patients must be authorized by a court

Page 4 - Ms. Sheila Gardner

order under 42 CFR § 2.65. 2/ In addition, such disclosures must be authorized by a court order even if the officer already has the information sought (see § 2.13(b)) or the disclosure is only implied (see § 2.13(e)).

We cannot clear the cooperative agreements which are the subject of your inquiry because they do not meet the requirements of 21 U.S.C. 1175(c) and 42 CFR § 2.65 for an authorizing court order. However, in situations other than those described in the attached agreements, drug treatment programs may agree to disclose information to local police departments for purposes of assisting in their investigative or prosecutorial functions without the necessity of obtaining an authorizing court order under § 2.65. These situations generally fall under two categories: (1) those in which such information may be disclosed without the necessity of obtaining any authorization under the regulations and (2) those in which written consent is needed under Subpart C or an authorizing court order is needed under 2.61-2.64 or 2.66 of Subpart E (these sections authorize court orders under circumstances different from those covered by § 2.65).

Communications which, under § 2.11(p)(3), include neither patient identifying information nor identifying numbers assigned by the program to patients or which are not restricted by the regulations (because no patient record is disclosed and, thus, the regulations do not apply) may be made without an authorizing court order or any other authorization under the regulations. These communications are discussed below.

As indicated at page 6 of the opinion letter, GH (Lanman) to Chief Vines, July 18, 1978 (D.F. #25B) (copy attached), it is our opinion that the regulations "do not restrict reports of crimes committed by program personnel" if no patient record is disclosed (because the regulations would not apply to such reports). 3/ Also, in the attached

Accord, letter, Comp. No. 76-23, NIDA (Besteman) to Brown, April 27, 1976 (D.F. #25B).

See, however, the restrictions on the use of "informants" 3/ and "undercover agents" in §§ 2.19 and 2.67 of the regulations.

- 70 -

Page 5 - Ms. Sheila Gårdner

4/

opinion letter, Comp. No. 77-29, GH (Lanman) to Bybee, December 8, 1977 (D.F. #25B), we have concluded that the confidentiality regulations do not restrict disclosures that a named individual is not and never has been an alcohol or drug abuse patient because, where an individual has never been a patient as defined by the regulations, there would be no "record" to be disclosed. Furthermore, § 2.13(d) provides that the regulations do not prohibit a program from reporting crimes committed by a patient on program premises or against program personnel or threats to do so (if the suspect is not identified as a patient). In this regard, we have previously advised in cases of "hot pursuit" that because the individual's flight is generally considered to be a crime, the individual's presence on the program premises may be construed as falling within the authorization for reporting crimes under § 2.13(d). 4/ In addition, if no patient records are disclosed, it is clear that the regulations do not restrict communications by program personnel or patients about crimes committed by nonpatients.

In summary, the following communications may be made by program personnel to law enforcement officials without obtaining any authorization under the regulations:

(1) communications which do not constitute disclosures of patient records restricted by the regulations (because no patient record is communicated) including,

(a) communications for the purpose of reporting crimes committed by program personnel or other nonpatients and,

(b) communications that a named individual is not and never has been a patient;

(2) communications made under § 2.13(d) for the purpose of reporting crimes committed by patients on program premises or against program personnel or threats to do so, including communications to assist police in "hot pursuit" (the suspect may not be identified as a patient unless an authorizing court order is obtained under § 2.65); and,

See the attached opinion letter, GH (Lanman) to Westergren, September 22, 1978 (D.F. #25B).

- 71 -

Page 6 - Ms. Sheila Gardner

(3) communications made under § 2.11(p)(3) which include neither patient identifying information nor identifying numbers assigned by the program to patients.

In addition, programs may agree to disclose patient identities or otherwise disclose patient records to local police departments for purposes of assisting in their investigative and prosecutorial efforts if those efforts are not directed at the patient whose record is disclosed or any other patient and if such disclosures are authorized under Subpart C or §§ 2.61-2.64 or 2.66 of Subpart E. Thus, with such authorization, programs may, for instance, disclose a patient's identity or physical whereabouts to the police in order to arrange a patient interview about a crime committed by a program employee or a nonpatient or for purposes other than criminal investigation or prosecution of any patient.

In our view, such disclosures may be authorized under § 2.40 of Subpart C as disclosures for the benefit of a patient if (1) the program finds that the disclosures promote a cooperative relationship with the local police and lessen the likelihood of disruption which might prove harmful to the program's treatment environment and (2) the program otherwise makes the determinations required by § 2.40. We have previously advised that the confidentiality regulations do not generally prohibit the provision of treatment conditioned upon the receipt of a written consent to certain disclosures. 5/ We conclude, therefore, that drug treatment programs may condition treatment upon receipt of written consent to disclosure of patient records, including patients' identities or physical whereabouts, for purposes of assisting police in investigating crimes committed by program employees or nonpatients or for purposes other than criminal investigation or prosecution of any patient (as stated above, disclosures for the purpose of a criminal investigation or prosecution of any patient may not be made unless an authorizing court order is obtained under § 2.65).

Letter, Comp. 77-12, GH (Lanman) to Clark, pp. 5-6 5/ and 9, June 6, 1977 (D.F. #25B).

- 72 -

Page 7 - Ms. Sheila Gardner

Accordingly, in our opinion, the confidentiality regulations do not prohibit drug treatment programs from entering into cooperative agreements with local police departments so long as the appropriate authorization is obtained for any agreed to disclosures of records, as described above. However, we wish to emphasize that the confidentiality regulations permit but do not mandate disclosures. This is illustrated by the discussion in § 2.61 which states that a subpoena or other compulsory process is necessary in addition to an authorizing court order to compel a disclosure under Subpart E. Thus, although a drug treatment program may enter into agreements to make certain disclosures if it obtains a written consent or an authorizing court order, such voluntary disclosures are not required by the regulations. We caution, therefore, that, in weighing the merits of a cooperative agreement with local police, drug treatment programs consider carefully what benefits they receive from the agreement and whether the agreement may have a chilling effect upon the voluntary participation in the program of current and prospective clients.

In summary, we conclude:

- other than patients;

(1) we cannot clear the attached cooperative agreements because, in our view, they provide for the disclosure of patient records for purposes of criminally investigating or prosecuting patients without specifying that an authorizing court order under § 2.65 is required;

treatment programs may enter into cooperative agreements with local police departments and may agree to (a) communicate (without obtaining authorization under the regulations) information which is not restricted by the regulations (because no patient record is disclosed) or which includes neither patient identifying information nor identifying numbers assigned by the program to patients and (b) disclose (if patient consent is obtained under Subpart C or an authorizing court order under \$\$-2.61-2.64 or 2.66 of Subpart E) patient records, including patients' identities or their physical whereabouts, for purposes of assisting police in the investigation and prosecution of persons

- 73 -

Page 8 - Ms. Sheila Gardner

(3) under the confidentiality regulations, programs may, in order to effectuate the agreed to disclosures described above, condition treatment upon receipt of written patient consent to make the disclosures.

If you need additional information or wish to discuss our advice, please let us know.

Chris B. Pascal

Attachments (5)

Cooperative Agreements Letter to Vines, dated July 18, 1978 Comp. #77-29 Letter to Westergren, dated September 22, 1978

- 74 -

cc: Fleetwood Roberts, NIAAA DCF Compilation

OGC/CPascal:jal:1/24/79



